SETTLEMENT AGREEMENT

dated as of

April 27, 2007

by and among

BLUE CROSS AND BLUE SHIELD ASSOCIATION BLUE CROSS AND BLUE SHIELD OF ALABAMA PREMERA BLUE CROSS ALSO DBA PREMERA BLUE CROSS BLUE SHIELD OF ALASKA CAREFIRST, INC.

GROUP HOSPITALIZATION AND MEDICAL SERVICES INC. CAREFIRST OF MARYLAND, INC.

BLUE CROSS AND BLUE SHIELD OF FLORIDA, INC. HAWAII MEDICAL SERVICE ASSOCIATION

> THE REGENCE GROUP REGENCE BLUESHIELD

REGENCE BLUESHIELD OF IDAHO, INC. REGENCE BLUECROSS BLUESHIELD OREGON REGENCE BLUECROSS BLUESHIELD OF UTAH

WELLMARK, INC. DBA WELLMARK BLUE CROSS AND BLUE SHIELD OF IOWA WELLMARK OF SOUTH DAKOTA, INC.

DBA WELLMARK BLUE CROSS AND BLUE SHIELD OF SOUTH DAKOTA LOUISIANA HEALTH SERVICE & INDEMNITY COMPANY. DBA BLUE CROSS AND BLUE SHIELD OF LOUISIANA HMO LOUISIANA, INC.

BLUE CROSS AND BLUE SHIELD OF MASSACHUSETTS, INC. BLUE CROSS AND BLUE SHIELD OF MASSACHUSETTS HMO BLUE, INC. BLUE CROSS BLUE SHIELD OF MICHIGAN, INC. BCBSM, INC., DBA BLUECROSS BLUESHIELD OF MINNESOTA HMO OF MISSISSIPPI, INC.

BLUE CROSS & BLUE SHIELD OF MISSISSIPPI BLUE CROSS AND BLUE SHIELD OF MONTANA, INC. HORIZON HEALTH CARE SERVICES, INC., DBA HORIZON BLUE CROSS BLUE SHIELD OF NEW JERSEY EMPIRE HEALTHCHOICE ASSURANCE, INC., DBA EMPIRE BLUE CROSS BLUE SHIELD

BLUE CROSS AND BLUE SHIELD OF NORTH CAROLINA HOSPITAL SERVICE ASSOCIATION OF NORTHEASTERN PENNSYLVANIA TRIPLE-S, INC.

> TRIPLE-S, INC. OF PUERTO RICO BLUE CROSS BLUE SHIELD OF RHODE ISLAND BLUE CROSS AND BLUE SHIELD OF SOUTH CAROLINA BLUECROSS BLUESHIELD OF TENNESSEE, INC. TENNESSEE HEALTH CARE NETWORK, INC. HEALTH CARE SERVICE CORPORATION

And other Subsidiaries listed on the Signature Pages

THE REPRESENTATIVE PLAINTIFFS,

THE SIGNATORY MEDICAL SOCIETIES,

AND CLASS COUNSEL

SETTLEMENT AGREEMENT

This Agreement is made and entered into as of the Execution Date by and among the Representative Plaintiffs in the Action (on behalf of themselves and each of the Class Members (as defined below)), by and through Class Counsel, Blue Cross Blue Shield Association, Blue Plans, and those medical societies identified on the signature pages hereto (such medical societies are herein collectively referred to as the "Signatory Medical Societies") (the Representative Plaintiffs, the Class Members, Blue Parties and the Signatory Medical Societies are herein collectively referred to as the "Parties"). The Parties intend this Agreement to resolve, discharge and settle the Released Claims, fully, finally and forever according to the terms and conditions set forth below.

WITNESSETH:

WHEREAS, certain Representative Plaintiffs in the Action filed the Complaint;

WHEREAS, Blue Parties deny the material factual allegations and legal claims asserted in the Complaint, including without limitation any and all charges of wrongdoing or liability arising out of any of the conduct, statements, acts or omissions alleged, or that could have been alleged, in the Complaint including without limitation the allegations that the Representative Plaintiffs and/or other Class Members have suffered damages; that Blue Parties improperly manipulated claim procedures or capitation payments or any other payments; that Blue Parties paid at incorrect rates or improperly applied reimbursement policies; that Blue Parties fraudulently misrepresented the criteria for insurance coverage determination, treatment decisions, claims payments and adequacy of capitation payments; that Blue Parties conspired with or aided and abetted wrongful conduct of any other Person; and that the Representative Plaintiffs and/or other Class Members were harmed by the conduct alleged in the Complaints;

WHEREAS, Blue Parties have asserted a number of defenses to the claims set forth in the Complaint that Blue Parties believe are meritorious; nonetheless, Blue Parties have a desire to make more transparent, simplify and otherwise improve the systems through which they conduct business with Class Members and have concluded that further conduct of the Action would be protracted and expensive and that it is desirable that the Action be fully and finally settled in the manner and upon the terms and conditions set forth in this Agreement;

WHEREAS, the Representative Plaintiffs believe that the claims asserted in the Complaint have merit; nonetheless Representative Plaintiffs and Class Counsel recognize and acknowledge the expense and length of continued proceedings that would be necessary to prosecute the Action against Blue Parties through trial and appeals;

WHEREAS, Representative Plaintiffs and Class Counsel also have taken into account the uncertain outcome and the risk of any class action, especially in complex actions such as the Action, as well as the difficulties and delays inherent in such actions, and counsel for the Representative Plaintiffs believe that the settlement set forth in this

Page 5 of 58

Agreement confers substantial benefits upon the Representative Plaintiffs and the other Class Members:

WHEREAS, based on their evaluation of all of these factors, and recognizing that Blue Parties' compliance with the terms of this Agreement is beneficial to Class Members and that such compliance does not and shall not violate any legal right of Class Members, the Representative Plaintiffs and Class Counsel have determined that this Agreement is in the best interests of themselves and the other Class Members; and

WHEREAS, the Signatory Medical Societies have determined that it is in their best interests to obtain the benefits afforded to such Signatory Medical Societies by the applicable provisions of this Agreement, and, in exchange therefor, to make the commitments and agreements contained herein, including without limitation those contained in § 13.

NOW, THEREFORE, IT IS HEREBY STIPULATED AND AGREED by and among the Parties that, subject to the approval of the Court, the Action and the Released Claims shall be finally and fully resolved, compromised, discharged and settled under the following terms and conditions:

1. Definitions.

As used in this Agreement and all exhibits to this Agreement, the following terms have the meaning specified.

- 1.1 "Action" means Love, et al. v. Blue Cross Blue Shield Association, et al., (formerly Thomas, et al. v. Blue Cross Blue Shield Association, et al.), Civil Action No. CV-03-21296-MORENO\SIMONTON in the United States District Court for the Southern District of Florida, Miami Division.
- 1.2 "Active Physician" means a Class Member who is a Physician and who is not a Retired Physician as of the Preliminary Approval Date.
- 1.3 "Active Physician Amount" shall have the meaning assigned to that term in § 8.3(b) of this Agreement.
- 1.4 "Adverse Determination" shall have the meaning assigned to that term in § 7.11(b)(i) of this Agreement.
- 1.5 "Affiliate" or "Affiliates" means with respect to any Person, any other Person controlling, controlled by or under common control with such first Person. The term "control" (including without limitation, with correlative meaning, the terms "controlled by" "under common control with"), as used with respect to any Person, means the possession, directly or indirectly, of the power to direct or cause the direction of the management and policies of such Person, whether through the ownership of voting securities or otherwise.
- 1.6 "Agreement" means this Settlement Agreement, inclusive of all exhibits hereto.

Case 1:03-cv-21296-FAM

- "AMA" means the American Medical Association. 1.7
- "Attorneys' Fees" means the funds for attorneys' fees and expenses that 1.8 may be awarded by the Court to Class Counsel.
- "Bar Order" means an order of the Court barring the assertion of claims against the Released Parties for contribution, indemnity or other similar claims by other Persons in the form included as part of the Final Order and Judgment.
- 1.10 "Base Amount" shall have the meaning assigned to that term in § 8.3(d) of this Agreement.
- 1.11 "BCBSA" means the Blue Cross and Blue Shield Association.
- "Billing Dispute" shall have the meaning assigned to that term in § 7.10(a) 1.12 of this Agreement.
- "Billing Dispute Reviewer" shall have the meaning assigned to that term 1.13 in § 7.10 of this Agreement. The Billing Dispute Reviewer shall have the requisite skill, experience and independence to carry out the functions set forth in § 7.10.
- "BlueCard Program" means the program governed by BCBSA that requires licensees of BCBSA to use certain policies, procedures and/or technology to service Blue Cross and/or Blue Shield members located outside of a licensee's service area.
- "Blue Compliance Dispute Facilitator" shall have the meaning assigned to that term in § 12.1(a) of this Agreement.
- "Blue Parties" means BCBSA and all of the Blue Plans, and "Blue Party" 1.16 means any one of the Blue Parties.
- "Blue Plan" means an entity identified as such on the signature pages hereto and its Subsidiaries.
- "Capitation" means the payment by a Blue Plan to Physicians, Physician Groups or Physician Organizations of a per member per month amount (including but not limited to percentage of premium) by which a Blue Plan transfers to the Physicians, Physician Groups or Physician Organizations the financial risk for those Covered Services as set forth in the contract between the Blue Plan and the Physicians, Physician Groups or Physician Organizations.
- "CCI" means CMS's published list of edits and adjustments that are made to health care providers' claims submitted for services or supplies provided to patients insured under the federal Medicare program and/or under other federal insurance programs.
- "Claim" shall have the meaning assigned to that term in § 13.1(a) of this Agreement.

- "Claim Form" shall have the meaning assigned to that term in § 8.3(h) of this Agreement.
- "Claim Form Instructions" shall have the meaning assigned to that term in § 8.3(h) of this Agreement.
- "Class" means any and all Physicians, Physician Groups and Physician Organizations who provided Covered Services to any Plan Member or services to any individual enrolled in or covered by a Plan offered or administered by any Person named as a defendant in the Complaint or by any other primary licensee of the BCBSA or by any of their respective current or former subsidiaries or Affiliates, in each case from May 22, 1999 through the Preliminary Approval Date. The Class shall exclude: (i) all Persons who, in accordance with the terms of this Agreement, execute a timely request for exclusion (Opt-Out) from the Class; and (ii) the Blue Parties, their Affiliates and any of their officers, directors and employees.
- "Class Compliance Dispute Facilitator" shall have the meaning assigned to that term in § 12.1(a) of this Agreement.
- "Class Counsel" means those persons identified in § 5 as Class Counsel. 1.25
- "Class Member" means any Person who is a member of the Class. 1.26
- "Clinical Information" means clinical, operative or other medical records 1.27 and reports kept in the ordinary course of a Physician's, Physician Group's or Physician Organization's business, and, where applicable, requested statements of Medical Necessity.
- "CMS" means the Centers for Medicare and Medicaid Services (formerly known as Health Care Financing Administration).
- "CMS-1500" means the health care provider claim form number 1500 created by CMS, as such form exists on the date of this Agreement and as it may be amended, modified or superseded thereafter during the term of this Agreement.
- "Complaint" means the initial complaint and any and all subsequent complaints filed in the Action.
- "Complete Claim" means a claim for Covered Services that (a) is timely received by the Blue Plan, (b) has a corresponding referral (whether in paper or electronic format), if required for the applicable claim, (c) meets all the requirements of § 7.17(b), (d) (i) when submitted via paper has all the elements of the CMS-1500 (or successor standard) forms or (ii) when submitted via an electronic transaction, uses only permitted standard code sets (e.g., CPT®-4, ICD-9, HCPCS) and has all the elements of the standard electronic formats, as required by applicable federal authority and state regulatory authority, (e) is a claim for which the Blue Plan is the primary payor or its responsibility as a secondary payor has been established, (f) contains no defect or error that would affect the

adjudication of the claim, (g) includes supporting documentation consistent with this Agreement sufficient for the Blue Plan to make a payment determination, and (h) is under a Plan for which all applicable premiums have been paid.

- "Compliance Dispute" means any claim that a Blue Party has failed to carry out any of its obligations under § 7 of this Agreement and Exhibit H to this Agreement; provided, however, that none of the following shall be deemed a Compliance Dispute: (a) a Released Claim, (b) a Retained Claim, (c) a Billing Dispute, or (d) a claim for which the review process for Adverse Determinations set forth in § 7.11 is available.
- "Compliance Dispute Claim Form" means a document in substantially the 1.33 same form as Exhibit C, attached hereto.
- "Compliance Dispute Facilitator" means the individuals selected to be the Class Compliance Dispute Facilitator, the Blue Compliance Dispute Facilitator or both, as the term is used herein.
- "Compliance Dispute Officer" shall have the meaning assigned to that 1.35 term in § 12 of this Agreement.
- "Court" means the United States District Court for the Southern District 1.36 of Florida.
- "Covered Services" means, with respect to a particular Blue Plan, a health care benefit that is within the coverage described in the Plan Documents applicable to an eligible Plan Member of such Blue Plan.
- "CPT®", "CPT® Codes" or "AMA CPT® book" means medical nomenclature in the publication entitled "CPT, Standard Edition", "CPT® Professional Edition", "CPT Assistant" and "Principals of CPT Coding" published by the AMA containing a systematic listing and coding of procedures and services provided to patients by physicians and certain non-physician health professionals. When used herein, "CPT®", "CPT® Codes" or "AMA CPT® book" refers to such medical nomenclature in the publication as it exists as of the date of this Agreement and as it may be amended, modified or superseded thereafter during the term of this Agreement.
- "CPT® Conventions" mean rules for the application of codes that go across all sections and subsections of the AMA CPT book, e.g. indented codes and add on codes.
- "CPT® Guidelines" mean those guidelines set out in the introduction, in 1.40 the beginning to each of the six major sections, in the subsections and in the code level parenthetic statements and cross references (excluding from this definition any reference to another publication that is not subject to the existing CPT Editorial Panel process, such as, but not limited to "CPT Assistant" or "Principles of CPT Coding") contained in the AMA publication "CPT, Professional Edition".

1.41 "Credentialing Committee" means any committee maintained by a Blue Plan that has decision-making authority regarding credentialing and recredentialing of individual Physicians as Participating Physicians with the Blue Plan.

Document 928-2

- "Delegated Entity" (1) as the term applies to arrangements in California 1.42 with respect to which a Blue Plan is operating as a Blue Cross and/or Blue Shield licensee, means (i) a risk-bearing organization, organized delivery system, limited or specialized licensed health plan or other risk-bearing entity as defined by California law, or (ii) a full service licensed health plan where it is reasonably necessary because a Blue Plan does not have reasonable capacity to provide or administer coverage in those geographic areas or specialty services; and (2) as the term applies otherwise, means, with respect to a particular Blue Plan, an entity that: (i) is not an Affiliate of such Blue Plan and is not another licensee of BCBSA. (ii) maintains its own contracts with Physicians separate from any contracts between such Blue Plan and Physicians, and (iii) by agreement with such Blue Plan, (A) agrees to provide Plan Members with access to such Physicians pursuant to terms of such agreements, and (B) performs some or all of the functions with respect to Plans which otherwise would be performed by such Blue Plan, including without limitation claims adjudication, utilization review, utilizations management and Physician credentialing.
- 1.43 "Downcoding" shall have the meaning assigned to that term in § 7.19 of this Agreement.
- 1.44 "Edit" means a practice or procedure pursuant to which one or more adjustments are made to CPT® Codes or HCPCS Level II Codes included in a claim that result in (a) payment being made based on some, but not all, of the CPT® Codes or HCPCS Level II Codes included in the claim, (b) payment being made based on different CPT® Codes or HCPCS Level II Codes than those included in the claim, (c) payment for one or more of the CPT® Codes or HCPCS Level II Codes included in the claim being reduced by application of Multiple Procedure Logic, (d) payment for one or more of the CPT® Codes or HCPCS Level II Codes being denied, or (e) any combination of the above.
- 1.45 "Effective Date" shall have the meaning assigned to that term in § 14.4 of this Agreement.
- 1.46 "Effective Period" shall have the meaning assigned to that term in the preamble to § 7 of this Agreement.
- 1.47 "Enrollment" or "Enrollment Date" shall mean the date upon which a Plan Member becomes eligible to receive Covered Services.
- 1.48 "EOB" means an Explanation of Benefit or any comparable form or statement communicating to a Plan Member the results of a Blue Plan's adjudication of claim(s) with respect to or on behalf of such Plan Member.

- "Execution Date" means the later of (i) the date on which the signatures of all Blue Parties have been delivered to Class Counsel, and (ii) the date on which the signatures of all Representative Plaintiffs, Signatory Medical Societies and Class Counsel have been delivered to all Blue Parties.
- "External Review" shall have the meaning assigned to that term in § 7.11(e)(i) of this Agreement.
- "FDA" means the U.S. Food and Drug Administration. 1.52
- "FEHBA" means the Federal Employees Heath Benefits Act and the rules and regulations and contracts promulgated or entered thereunder.
- 1.54 "Final Order and Judgment" means the order and form of judgment approving this Agreement and dismissing the Blue Parties, with prejudice, in the Action in the form attached hereto as Exhibit D.
- "Final Order Date" means the date on which the Court enters the Final 1.55 Order and Judgment.
- 1.56 "Foundation" shall have the meaning assigned to that term in § 8.3(g) of this Agreement.
- 1.57 "Force Majeure" shall have the meaning assigned to that term in § 7.32 of this Agreement.
- "Fully-Insured Plan" means a Plan as to which a Blue Plan assumes all or a majority of healthcare cost and/or utilization risk.
- "HCPCS Level II Codes" means alphanumeric codes used to identify those codes not included in CPT® and that are commonly referred to as Healthcare Common Procedure Coding System Level II Codes.
- "HIPAA" means the Health Insurance Portability and Accountability Act 1.60 of 1996.
- "Implementation Date" shall have the meaning assigned to that term in the preamble to § 7 of this Agreement.
- "Independent Review Organization" shall have the meaning assigned to that term in $\S 7.11(e)(i)$.
- "Individually Negotiated Contract" means a contract pursuant to which the parties to the contract, as a result of negotiation, agreed to one or more modifications to the terms of a Blue Plan's applicable standard form agreement that substantially modify the standard form agreement and that are made to individually suit, in whole or in part, the needs of a Participating Physician,

Participating Physician Group or Participating Physician Organization (including but not limited to higher or customized rates and/or other customized payment methodologies).

- 1.64 "Intellectual Property Claim" means any claim of patent, copyright, or trademark infringement or trade secret or other proprietary rights asserted or threatened against a Blue Party.
- 1.65 "Interest Rate" means a 4.75% annual rate of return without compounding.
- 1.66 "Joint Compliance Dispute Program" shall have the meaning assigned to that term in § 12.1(c) of this Agreement.
- 1.67 "Mailed Notice" means the form of the notice attached hereto as Exhibit E.
- 1.68 "Medical Necessity" and "Medically Necessary" shall have the meaning assigned to those terms in § 7.16(a) of this Agreement.
- 1.69 "Multiple Procedure Logic" means the practices or procedures used by a Blue Plan to reduce the allowable amount for one or more of the CPT® Codes or HCPCS Level II Codes included in a claim as a result of multiple surgical procedures or services having been performed on the same patient on the same date of service.
- 1.70 "NASCO" means National Account Service Company, LLC.
- 1.71 "Negotiated Purchaser" shall have the meaning assigned to that term in § 7.36 of this Agreement.
- 1.72 "Non-Participating" means, with respect to a Physician, Physician Group, or Physician Organization, a Physician, Physician Group, or Physician Organization that is not a Participating Physician, Participating Physician Group, or Participating Physician Organization.
- 1.73 "Notice Administrator" shall mean the Person appointed to administer the provisions relating to Notice to potential Class Members, under § 6.
- 1.74 "Notice Date" shall have the meaning assigned to that term in § 6.1 of this Agreement.
- 1.75 "Objection Date" shall have the meaning assigned to that term in § 6 of this Agreement.
- 1.76 "Opt-Out" shall have the meaning assigned to that term in § 6.1 of this Agreement.
- 1.77 "Opt-Out Deadline" shall have the meaning assigned to that term in § 6.1 of this Agreement.

Case 1:03-cv-21296-FAM

- 1.78 "Overpayment" means, with respect to a claim submitted by or on behalf of a Physician, Physician Group or Physician Organization, any erroneous or excess payment that a Blue Plan makes for any reason including, but not limited to, (i) payment at an incorrect rate, (ii) duplicate payments for the same Physician Service, (iii) payment with respect to an individual who was not a Plan Member on the date the Physician provided the Physician Service(s) that are the subject of such payment, and (iv) payment for any non-Covered Service.
- 1.79 "Participating Physician" means a Physician who has entered into a valid written contract with a Blue Plan (or who has agreed pursuant to an arrangement with a Physician Group, Physician Organization or other entity which has a valid written contract with a Blue Plan) to provide Covered Services to that Blue Plan's Plan Members and, where applicable, who meets the Blue Plan's credentialing requirements, during the effective period of such a contract. The fact that a Physician has entered into an agreement with a rental network does not make that Physician a Participating Physician.
- 1.80 "Participating Physician Group" means a Physician Group that has entered into a valid written contract with a Blue Plan to provide Covered Services to that Blue Plan's Plan Members.
- 1.81 "Participating Physician Organization" means a Physician Organization that has entered into a valid written contract with a Blue Plan to provide Covered Services to that Blue Plan's Plan Members.
- 1.82 "Participating Psychiatrist" means a Psychiatrist who is a Participating Physician.
- 1.83 "Parties", each a "Party" shall have meaning assigned to that term in the preamble of this Agreement.
- 1.84 "Person" or "Persons" means all persons and entities (including without limitation natural persons, firms, corporations, limited liability companies, joint ventures, joint stock companies, unincorporated organizations, agencies, bodies, governments, political subdivisions, governmental agencies and authorities, associations, partnerships, limited liability partnerships, trusts, and their predecessors, successors, administrators, executors, heirs and assigns).
- 1.85 "Physician" means an individual duly licensed by a state licensing board as a Medical Doctor or as a Doctor of Osteopathy and shall include both Participating Physicians and Non-Participating Physicians.
- 1.86 "Physician Advisory Committee" shall have the meaning assigned to that term in § 7.9(a) of this Agreement.
- 1.87 "Physician Group" means two or more Physicians, and those claiming by or through them, who practice under a single taxpayer identification number.

- "Physician Organization" means any association, partnership, corporation or other form of organization (including without limitation independent practice associations and physician hospital organizations), that arranges for care to be provided to Plan Members by Physicians organized under multiple taxpayer identification numbers.
- "Physician Services" means Covered Services that a Physician provides to a Plan Member, as specified in applicable agreements with a Blue Plan or otherwise.
- "Physician Specialty Society" means a United States medical specialty 1.90 society that represents diplomats certified by a board recognized by the American Board of Medical Specialties.
- "Plan" means a benefit plan through which a Plan Member obtains health care benefits set forth in pertinent Plan Documents.
- "Plan Documents" means the documents defining the health care benefits available to a Plan Member, including the Plan Member's summary plan description, certificate of coverage or other applicable coverage document, and the terms and conditions under which such benefits are available under the Plan.
- "Plan Member" means an individual enrolled in or covered by a Plan offered and administered by a Blue Plan.
- "Post-Service Appeal" shall have the meaning assigned to that term in § 1.94 7.11(c)(ii)(A) of this Agreement.
- 1.95 "Precertification" ("Precertify" or "Precertifies") means the prior approval by a Blue Plan that the service or supply is Medically Necessary and/or not experimental or investigational.
- "Preliminary Approval Date" means the date the Preliminary Approval 1.96 Order is entered by the Court.
- "Preliminary Approval Hearing" shall have the meaning assigned to that term in § 4 of this Agreement.
- "Preliminary Approval Order" means the preliminary approval order as attached hereto at Exhibit F.
- "Pre-Service Appeal" shall have the meaning assigned to that term in § 1.99 7.11(c)(i) of this Agreement.
- 1.100 "Product Network" means a network of Participating Physicians who, pursuant to contracts with a Blue Plan, provide Covered Services to Plan Members for one or more products or types of products offered by the Blue Plan (e.g., HMO, PPO, POS, Indemnity) in exchange for a specified type of compensation (e.g., fee for service, Capitation).

- 1.101 "Programs" shall have the meaning assigned to that term in § 7.29(k)(iii) of this Agreement.
- 1.102 "Provider Website" means the secure (password protected) online resources for Participating Physicians to obtain information about a Blue Plan, its products and policies and other information described in more detail in this Agreement.
- 1.103 "Psychiatrist" means a Physician who is duly licensed by a state licensing board to provide mental health services and shall include without limitation both Participating Physicians and Non-Participating Physicians.
- 1.104 "Public Website" means the online resources for the public to obtain information about a Blue Party, its products and policies and other information.
- 1.105 "Published Notice" means the form of notice attached hereto as Exhibit G.
- 1.106 "Qualified Reviewer" shall have the meaning assigned to that term in § 7.11(c)(ii)(A) of this Agreement.
- 1.107 "Released Parties" shall have the meaning assigned to that term in § 13.1(a) of this Agreement.
- 1.108 "Released Claims" shall have the meaning assigned to that term in § 13.1(c) of this Agreement.
- 1.109 "Releasing Parties" shall have the meaning assigned to that term in § 13.1(a) of this Agreement.
- 1.110 "Remittance Advice" means the form sent by a Blue Plan to health care providers explaining the Blue Plan's computation of benefits and payment amounts on a claim. The Remittance Advice is sometimes referred to as an "Explanation of Payment" form or "EOP."
- 1.111 "Representative Plaintiffs" means collectively Rick Love, M.D., Joe Frank Smith, M.D., Scott Elledge, M.D., and Andreas Melendez-Desos, M.D.
- 1.112 "Representative Plaintiffs in Other Actions" means collectively Maxwell Cooper, M.D., Michon Morita, M.D., Andrew Cheng, M.D., Charles Rothberg, M.D., Mitchell Garden, M.D., Miriam Levitt, M.D., Robert J. Hughes, M.D., Joseph K. Newsom, M.D., Joseph K. Newsom, Jr., M.D., Robert Waters, M.D., John R. Nobles, M.D., Zachary Rosenberg, M.D., Dewayne P. Darby, M.D., Donald S. Horner, M.D., Amity Obstetrics & Gynecology Associates, P.A., G. William Haggerson, M.D., Forsyth Surgical Associates, P.A., and Lakeland Anesthesia, Inc.
- 1.113 "Retained Claims" shall have the meaning assigned to that term in § 13.6 of this Agreement.

Page 15 of 58

- 1.114 "Retired Physician" means a Class Member who, subsequent to May 22, 1999, has become an inactive Physician, has retired from the practice of, or has otherwise ceased to practice, medicine or has died as of the Preliminary Approval
- 1.115 "Retired Physician Amount" shall have the meaning assigned to that term in § 8.3(a) of this Agreement.
- 1.116 "Reversion Amount" shall have the meaning assigned to that term in § 8.4 of this Agreement.
- 1.117 "Self-Insured Plan" means any Plan other than a Fully-Insured Plan.
- 1.118 "Settlement Administration Account" shall have the meaning assigned to that term in § 8.2(b) of this Agreement.
- 1.119 "Settlement Administrator" shall have the meaning assigned to that term in § 8.3 of this Agreement.
- 1.120 "Settlement Amount" shall have the meaning assigned to that term in § 8.2 of this Agreement.
- 1.121 "Settlement Fund" shall have the meaning assigned to that term in § 8.2 of this Agreement.
- 1.122 "Settlement Fund Payment" shall have the meaning assigned to that term in § 8.2(a) of this Agreement.
- 1.123 "Settlement Hearing" shall have the meaning assigned to that term in § 6.2 of this Agreement.
- 1.124 "Settlement Hearing Date" shall have the meaning assigned to that term in § 6.2 of this Agreement.
- 1.125 "Shane" shall mean Shane v. Humana, Inc., et al., Master File No. 00-1334 MDL-Moreno ("Shane I") and Shane v. Humana, Inc., et al., Case No. 04-21589-CIV-Moreno ("Shane II") in the United States District Court for the Southern District of Florida, Miami Division.
- 1.126 "Signatory Medical Societies" shall have the meaning assigned to that term in the preamble of this Agreement.
- 1.127 "Significant Edit" means an Edit that a Blue Plan reasonably believes, based on its experience with submitted claims, will cause, on the initial review of submitted claims, the denial of or reduction in payment for a particular CPT® Code or HCPCS Level II Code more than two-hundred and fifty (250) times per year in any state in which the Blue Plan operates.
- 1.128 "Subsidiary" or "Subsidiaries" shall mean the entity or entities identified as such on the signature page of this Agreement.

- 1.129 "Tag Along Actions" shall have the meaning set forth in § 15.1.
- 1.130 "Terminating Blue Plan" shall have the meaning set forth in § 14.2(h).
- 1.131 "Termination Date" shall have the meaning assigned to that term in § 14.6 of this Agreement.
- 1.132 "TriCare" shall have the meaning assigned to that term in \S 7.30(b) of this Agreement.
- 1.133 "Unopposed Amount" shall have the meaning assigned to that term in § 9.1 of this Agreement.

2. Intentionally Left Blank

3. Commitment to Support and Communications with Class Members

The Parties agree that it is in their best interests to consummate this Agreement and all the terms and conditions contained herein and to cooperate with each other and to take all actions reasonably necessary to obtain Court approval of this Agreement and entry of the orders of the Court that are required to implement its provisions. They also agree to support this Agreement in accordance with and subject to the provisions of this Agreement.

Class Counsel and Representative Plaintiffs and Representative Plaintiffs in Other Actions shall make every reasonable effort to encourage putative Class Members to participate and not to Opt-Out. In addition, Class Counsel shall make all reasonable efforts to enforce the Compliance Dispute resolution provisions of this Agreement set forth in § 12.

Representative Plaintiffs, Representative Plaintiffs in Other Actions, Class Counsel and Blue Parties agree that Blue Parties may communicate with putative Class Members regarding the provisions of this Agreement, so long as such communications are not inconsistent with the Mailed Notice or other agreed upon communications concerning the Agreement. Blue Parties will refer to the Settlement Administrator or to Class Counsel any inquiries from Class Members about claims to be filed under this Agreement.

4. Preliminary Approval of Settlement

Pursuant to Rule 23(e) of the Federal Rules of Civil Procedure, the Parties shall submit this Agreement, together with the exhibits attached hereto, to the Court as soon as practicable following the Execution Date at a hearing (the "Preliminary Approval Hearing") for, among other things, its conditional certification of a settlement class, preliminary approval of the Agreement, the Mailed Notice, the Published Notice and the Claim Form and shall apply to the Court for an order of preliminary approval and conditional class certification, substantially in the form of Exhibit F ("Preliminary Approval Order").

5. Notice to Putative Class Members; Notice to Parties Pursuant to This Agreement

After the Court has entered the Preliminary Approval Order and approved the Mailed Notice, the Published Notice and the Claim Form, notice to putative Class Members shall be disseminated in such form as the Court shall direct; provided that the forms of notice are substantially similar to the Mailed Notice and the Published Notice. A copy of the Claim Form shall be included with the copy of the Mailed Notice that is disseminated to putative Class Members.

Class Counsel and Blue Parties shall be jointly responsible for identifying names and addresses of putative Class Members and shall cooperate with each other and the Notice Administrator to make such identifications and determinations.

Blue Parties shall pay the reasonable cost of notice to putative Class Members, including without limitation first class mail costs for the mailing of the Mailed Notice, substantially in the same form as Exhibit E. Payment by Blue Parties of the cost of the Mailed Notice shall be non-refundable and shall be in addition to the other agreements made herein. Blue Parties shall pay for the cost to publish the Published Notice three times in the legal notices section in the national editions of THE WALL STREET JOURNAL and USA TODAY. If publication in one or more of said publications on the foregoing schedule is determined not to be practicable, then either Class Counsel or Blue Parties may apply to the Court for alternative notice by publication. Each Blue Plan shall also publish the Mailed Notice on its Public Website. The Blue Parties shall, to the extent feasible, also publish the Published Notice in a nationwide periodical addressing issues of concern to Physicians such as The Journal of the American Medical Association or The American Medical News. Each Blue Plan shall maintain the Public Website notices at such Blue Plan's cost through at least the Objection Date.

All notices to any Party (including without limitation any designations made by Class Counsel pursuant to this Agreement) required under this Agreement shall be sent by first class U.S. Mail, by hand delivery, or by facsimile, to the recipients designated in this Agreement. Timeliness of all submissions and notices shall be measured by the date of receipt, unless the addressee refuses or delays receipt. The Persons designated to receive notices under this Agreement are as follows, unless notification of any change to such designation is given to each other Party hereto pursuant to this § 5:

Representative Plaintiffs and Signatory Medical Societies: Notice to be given to Class Counsel on behalf of Representative Plaintiffs and Signatory Medical Societies.

Class Counsel:

Archie C. Lamb, Jr. Law Offices of Archie C. Lamb, LLC 2017 Second Avenue North Harley S. Tropin
Janet L. Humphreys
Adam M. Moskowitz
Kozyak Tropin & Throckmorton, PA

Birmingham, AL 35203

Aaron S. Podhurst Podhurst Orseck, PA 25 W. Flagler Street, Suite 800 Miami, FL 33130-1780

Nicholas B. Roth Eyster Key Tubb Weaver & Roth, LLP 402 East Moulton Street, SE Eyster Building Decatur, AL 35601

Mark Gray **GRAY WHITE & WEISS** 1200 PNC Plaza 500 West Jefferson Louisville, KY 40202

Robert Foote FOOTE & MEYERS 416 S. 2nd Street Geneva, IL 60134

James B. Tilghman STEWART TILGHMAN FOX & BIANCHI 1 SE 3rd Avenue, Ste 3000 Miami, FL 33131-1764

Dennis G. Pantazis WIGGINS CHILDS QUINN & **PANTAZIS** 1400 SouthTrust Tower 420 North 20th Street Birmingham, AL 35203

Kenneth Trujillo TRUJILLO RODRIGUEZ & RICHARDS, LLC 1717 Arch Street Suite 3838 Philadelphia, Pennsylvania 19103

2525 Ponce de Leon, 9th Floor Miami, FL 33131-2335

Edith M. Kallas Joseph P. Guglielmo Whatley Drake & Kallas, LLC 1540 Broadway New York, NY 10036

Joe R. Whatley, Jr. Charlene P. Ford Othni J. Lathram Whatley Drake & Kallas, LLC 2323 Second Avenue North Birmingham, AL 35203-3807

J. Mark White WHITE ANDREWS ARNOLD & DOWD 2025 3rd Avenue North, Ste 600 Birmingham, AL 35203

Guido Saveri R. Alexander Saveri Cadio Zirpoli SAVERI & SAVERI 111 Pine Street, Ste 1700 San Francisco, CA 94111-5619

Kenneth S. Canfield Ralph Knowles DOFFERMYRE SHIELDS CANFIELD KNOWLES & DEVINE 1355 Peachtree St., Ste 1600 Atlanta, GA 30309

James E. Hartley, Jr. DRUBNER HARTLEY & O'CONNOR 500 Chase Parkway, 4th Fl. Waterbury, CT 06708

STROM LAW FIRM LLC 2110 N. Beltline Boulevard Suite A Columbia, South Carolina 29204-3905 POMERANTZ HAUDEK BLOCK GROSSMAN & GROSS 100 Park Avenue New York, New York 10017-5516

David L. Steed **CORNELIUS & COLLINS** Suite 1500 Nashville City Center 511 Union Street, P.O. Box 190695 Nashville, Tennessee 37219-0695

MCCUTCHEN BLANTON JOHNSON & BARNETTE LLP 1414 Lady Street P.O. Drawer 11209 Columbia, South Carolina 29211

Blue Parties:

Blue Cross and Blue Shield Association	Chester T. Kamin JENNER & BLOCK LLP 330 North Wabash Chicago, IL 60611	Daniel A. Engel Blue Cross and Blue Shield Association 225 North Michigan Avenue Chicago, IL 60601
Blue Cross and Blue Shield of Alabama	Michael A. Pope MCDERMOTT WILL & EMERY LLP 227 W. Monroe Street Chicago, IL 60606	Grey Till, Esquire Blue Cross and Blue Shield of Alabama 450 River Chase Parkway East Birmingham, Alabama 35244
Premera Blue Cross, also dba Premera Blue Cross Blue Shield of Alaska	Gwendolyn C. Payton LANE, POWELL PC 1420 5th Avenue, Suite 4100 Seattle, WA 98101	
CareFirst, Inc.; Group Hospitalization and Medical Services Inc.; CareFirst of Maryland, Inc.	Daly D. E. Temchine EPSTEIN, BECKER & GREEN, P.C. 1227 25th Street, N.W., Suite 700 Washington, D.C. 20037	

Blue Cross and Blue Shield of Florida, Inc.	Michael A. Pope MCDERMOTT WILL & EMERY LLP 227 W. Monroe Street Chicago, IL 60606 Jess H. Griffiths	
Association	GODBEY GRIFFITHS REISS 1001 Bishop St., 2300 Pauahi Tower Honolulu, HI 96813	
The Regence Group Regence BlueShield; Regence BlueShield of Idaho, Inc.; Regence BlueCross BlueShield Oregon; Regence BlueCross BlueShield of Utah	Michael A. Pope MCDERMOTT WILL & EMERY LLP 227 W. Monroe Street Chicago, IL 60606	
Wellmark, Inc. dba Wellmark Blue Cross Blue Shield of Iowa; Wellmark of South Dakota, Inc. dba Wellmark Blue Cross and Blue Shield of South Dakota	Michael A. Pope MCDERMOTT WILL & EMERY LLP 227 W. Monroe Street Chicago, IL 60606	
Louisiana Health Service & Indemnity Company dba Blue Cross and Blue Shield of Louisiana; HMO Louisiana, Inc.	Michael A. Pope MCDERMOTT WILL & EMERY LLP 227 W. Monroe Street Chicago, IL 60606	
Blue Cross and Blue Shield of Massachusetts, Inc.; Blue Cross and Blue Shield of Massachusetts HMO Blue, Inc.	Daly D. E. Temchine EPSTEIN, BECKER & GREEN, P.C. 1227 25th Street, N.W., Suite 700 Washington, D.C. 20037	

Page 21 of 58

LLP

227 W. Monroe Street

Chicago, IL 60606

Northeastern

Pennsylvania

Triple-S, Inc.; Triple-S, Inc. of Puerto Rico	Cesar Alcover, Esq. Casellas Alcover & Burgos Popular Center Bldg Suite 1400 208 Ponce de León Avenue San Juan, Puerto Rico 00918 P.O. Box 364924 San Juan, Puerto Rico 00936-4924	
Blue Cross Blue Shield of Rhode Island	Michael A. Pope MCDERMOTT WILL & EMERY LLP 227 W. Monroe Street Chicago, IL 60606	
Blue Cross and Blue Shield of South Carolina	Michael A. Pope MCDERMOTT WILL & EMERY LLP 227 W. Monroe Street Chicago, IL 60606	
Health Care Service Corporation	Michael A. Pope MCDERMOTT WILL & EMERY LLP 227 W. Monroe Street Chicago, IL 60606	
BlueCross BlueShield of Tennessee, Inc.; Tennessee Health Care Network, Inc.	Michael A. Pope MCDERMOTT WILL & EMERY LLP 227 W. Monroe Street Chicago, IL 60606	

In the event that any Party receives a notice from any other Party (in accordance with the provisions of § 5 of this Agreement and as required by any other provision of this Agreement), for which there is a written acknowledgement of receipt, and such receiving Party does not respond to such notice within thirty (30) days of receipt thereof (unless a different time is set forth in this Agreement), such receiving Party shall be deemed to have accepted any proposal made by the notifying Party in such notice and shall be deemed to have waived any rights under this Agreement with respect to the matter that is the subject of such notice.

6. Procedure for Final Approval; Limited Waiver

Following the dissemination of the Mailed Notice and Published Notice as described in § 5, Representative Plaintiffs, Signatory Medical Societies, Class

Counsel and the Blue Parties shall seek the Court's final approval of this Agreement. Putative Class Members shall have until the Objection Date to file, in the manner specified in the Mailed Notice, any objection or other response to this Agreement. The Parties agree to urge the Court to set the Objection Date for a date that is at least seventy-five (75) days after the Notice Date (the "Objection Date").

6.1 **Opt-Out Timing and Rights**

The Parties will jointly request of the Court that the Mailed Notice and the Published Notice be disseminated no earlier than 30 days and no later than 60 days after the Preliminary Approval Date. The Notice Date (the "Notice Date") shall be the date on which the notice is first mailed or published.

The Mailed Notice and the Published Notice shall provide that putative Class Members may request exclusion from the Class by providing notice, in the manner specified in the Notice, on or before a date set by the Court as the Opt-Out Deadline. Representative Plaintiffs, Class Counsel and the Blue Parties agree to urge the Court to set the Opt-Out Deadline for the date that is 60 days after the Notice Date (the "Opt-Out Deadline").

Putative Class Members have the right to exclude themselves ("Opt-Out") from this Agreement and from the Class by timely submitting a request to Opt-Out and otherwise complying with the agreed upon Opt-Out procedure approved by the Court. Putative Class Members who are . Physician Groups or Physician Organizations shall be entitled to exclude themselves, as distinct legal entities, from the Class, but shall have no right to exercise the Opt-Out rights of individual Physicians practicing under their auspices, unless the Physician Group or Physician Organization has written authorization to act on behalf of the Physician which has been submitted to the Notice Administrator; otherwise, in order to Opt-Out, individual Physicians associated with a Physician Group or Physician Organization must submit a request to Opt-Out in the manner specified in the Notice by the Opt-Out Deadline. Putative Class Members who so timely request to Opt-Out shall be excluded from this Agreement and from the Class. Any putative Class Member who does not submit a request to Opt-Out in the manner specified in the Notice by the Opt-Out Deadline or who does not otherwise comply with the agreed upon Opt-Out procedure approved by the Court shall be a Class Member and shall be bound by the terms of this Agreement and the Final Order and Judgment. Any Class Member shall be deemed to have taken all actions necessary to withdraw and revoke the assignment to any Person of any claim against the Blue Parties.

Any putative Class Member who timely submits a request to Opt-Out shall have until the Settlement Hearing Date to deliver to the Notice Administrator a written revocation of such putative Class Member's

request to Opt-Out. The Notice Administrator shall timely apprise the Parties of such revocations as of that date. The Parties shall timely apprise the Court of all such revocations.

Within fifteen (15) days after the Opt-Out Deadline, the Notice Administrator shall furnish the Blue Parties with a complete list in machine-readable form of all Opt-Out requests submitted by the Opt-Out Deadline. Within ten (10) days after the Settlement Hearing Date, the Notice Administrator shall furnish the Blue Parties with a complete list, in machine readable form, of all Opt-Out requests submitted by the Opt-Out deadline and not withdrawn by the Settlement Hearing Date. The Blue Parties shall pay costs of obtaining a copy of the Opt-Out requests.

6.2 Setting the Settlement Hearing Date and Settlement Hearing Proceedings

The Representative Plaintiffs, the Signatory Medical Societies, Class Counsel and the Blue Parties agree to urge the Court to hold a hearing to consider final approval of this Agreement (the "Settlement Hearing") on a date that is at least 120 days and is as soon as practicable after the Notice Date (the "Settlement Hearing Date"), and to work together to identify and submit any evidence that may be required by the Court to satisfy the burden of proof for obtaining approval of this Agreement and the orders of the Court that are necessary to effectuate the provisions of this Agreement, including without limitation, the Final Order and Judgment and the orders contained therein. At the Settlement Hearing, the Representative Plaintiffs, the Signatory Medical Societies, Class Counsel and the Blue Parties shall present evidence necessary and appropriate to obtain the Court's approval of this Agreement, the Final Order and Judgment and the orders contained therein (including without limitation the Bar Order), and shall meet and confer prior to the Settlement Hearing to coordinate their presentation to the Court in support of Court approval thereof.

7. Settlement Consideration: Business Practice Initiatives

The settlement consideration to the Class Members includes, among other things, initiatives and other commitments with respect to the Blue Plans' business practices. The Parties agree that the business practice initiatives and other commitments set forth below, which absent this Agreement Blue Plans would be under no obligation to undertake, constitute substantial value, and will enhance and facilitate the delivery of Physician Services by Class Members. Many of the Blue Plans have investigated and begun to implement certain of the business practice initiatives described in or contemplated by this § 7 after the Action began and/or while the Parties were engaged in discussions to resolve the Action. Such initial and partial implementation, which shows the Parties' good faith desire to resolve the Action, were undertaken to form part of the consideration of the settlement. Blue Plans shall have the unilateral and unrestricted right to block

access to and/or not apply any or all of the business practice initiatives described in or contemplated by this § 7 to Physicians, if any, who Opt-Out of the Agreement. Without in any way qualifying or limiting the foregoing, Blue Plans (a) are informed that it is not uncommon for some members of a class action to opt out for a variety of reasons independent of, among other things, the substantive allegations in the complaint or the terms of a proposed settlement, and (b) state their present intention to exercise, in whole or in part, the rights referred to in the immediately preceding sentence to Persons eligible to become Class Members who nevertheless Opt-Out.

Each Blue Plan covenants and agrees that, during the period from and after the Execution Date and until the Preliminary Approval Date, they shall not effect any material changes in the business practices that are the subject of the Complaint and governed by the provisions of this Agreement, except changes to such business practices that are contemplated by or otherwise consistent with this Agreement.

Except as otherwise provided in Exhibit H, each Blue Party shall be obligated to implement those commitments applicable to it as provided in this § 7 and to commence such implementation as set forth in Exhibit I (the applicable "Implementation Date"). Each Blue Party shall continue implementing each such commitment until the Termination Date, except as and only to the extent that a term of this Agreement has been extended by a Blue Party as provided in § 14.6(b). With respect to each commitment of a Blue Party set forth in this § 7 (and not otherwise amended, modified, limited or excluded by Exhibit H), the "Effective Period" shall be the period of time beginning on the Implementation Date shown on Exhibit I and ending as specified in § 14.6(b). Notwithstanding anything to the contrary contained in this Agreement, with respect to each commitment set forth in this § 7 (and as otherwise amended, modified, limited or excluded by Exhibit H), from and after the Termination Date, a Blue Party shall be under no obligation whatsoever to continue to implement such commitment, except as provided in § 14.6(b).

7.1 Intentionally Left Blank

7.2 Intentionally Left Blank

7.3 Availability of Fee Schedules and Scheduled Payment Dates

Each Blue Plan shall develop and implement a plan not later than twelve (12) months after the Final Order Date reasonably designed to permit its Participating Physician, Participating Physician Group, or Participating Physician Organization that, in each case, has entered into a written contract directly with the Blue Plan, to the extent the Participating Physician, Participating Physician Group or Participating Physician Organization is compensated on a non-capitated basis, to view, by CD-ROM or electronically (at the Blue Plan's option), on a confidential basis, complete fee information showing the applicable fee schedule amounts for

Page 26 of 58

such Participating Physician, Participating Physician Group, or Participating Physician Organization pursuant to that Participating Physician's, Participating Physician Group's, or Participating Physician Organization's direct written agreement with the Blue Plan. A Participating Physician, Participating Physician Group or Participating Physician Organization may elect to receive a hard copy of the fee schedule in lieu of the foregoing. The fee schedule information will be provided by the fee for service dollar amount allowable for each CPT® Code for those CPT® Codes that a Participating Physician, Participating Physician Group, or Participating Physician Organization in the same specialty typically uses in providing Covered Services. A Participating Physician, Participating Physician Group or Participating Physician Organization may request and the Blue Plan will provide the fee for service dollar amount allowable for other CPT® Codes that its Participating Physician, Participating Physician Group or Participating Physician Organization actually bills the Blue Plan. A Blue Plan may base actual compensation on the Blue Plan's maximum allowable amount and/or other contract adjustments. Commencing with the Final Order Date and continuing until implementation of the initiative described above, each Blue Plan, upon written request from a Participating Physician, Participating Physician Group, or Participating Physician Organization that, in each case, has entered into a written contract directly with that Blue Plan, will provide, by hard copy, the fee schedule for up to one hundred (100) CPT® Codes customarily and routinely used by such Participating Physician, Participating Physician Group, or Participating Physician Organization, as specified by such Participating Physician, Participating Physician Group, or Participating Physician Organization. Each Blue Plan shall be obligated to honor only two such requests made annually by such Participating Physician, Participating Physician Group, or Participating Physician Organization. Each Blue Plan will attempt to include provisions in its agreements with Delegated Entities that require comparable disclosure. Each Blue Plan will not require its Participating Physicians, Participating Physician Groups, or Participating Physician Organizations to provide that Blue Plan with billing rates as a precondition to that Blue Plan providing fee information pursuant to this section.

7.4 Intentionally Left Blank

7.5 Reduced Precertification Requirements

A number of Blue Plans have reduced, and each Blue Plan will commence or continue to attempt to limit, the number of services and supplies requiring Precertification, and have standardized the services and supplies for which Precertification is required within each market, line of business (e.g., group, individual, etc.) or product for its Fully-Insured and Self-Insured Plans. Each Blue Plan will continue to review its Precertification requirements for further opportunities to reduce the number of services and supplies requiring Precertification. Each Blue Plan may continue to

require Precertification for services and supplies and may alter or amend the number of services and supplies requiring Precertification in response to changes in market conditions, medical technology, and utilization patterns. Each Blue Plan shall post to its Provider Website not later than three (3) months after the Final Order Date those services or supplies for which Precertification is routinely required for its products, and shall update such posting to the extent the services or supplies for which Precertification is routinely required changes. Notwithstanding the above, a Blue Plan's Self-Insured Plan customers may specify services or supplies for which Precertification is required that differ from or are in addition to the services or supplies for which that Blue Plan routinely requires Precertification for its Fully-Insured Plans, and such Self-Insured Plans may contract with a different entity to provide Precertification services. Each Blue Plan will propose to its Self-Insured Plan customers that they utilize the Blue Plan's standard list of services and supplies for which Precertification is required. With a Self-Insured Plan's approval, each such Blue Plan will post such Self-Insured Plan's customized list of Precertification requirements to the Blue Plan's Provider Website.

7.6 Greater Notice of Policy and Procedure Changes

Each Blue Plan shall, if it intends to make a material adverse change(s) in the terms of its contracts (including policies and procedures incorporated by reference therein) with its Participating Physicians, Participating Physician Groups, or Participating Physician Organizations, give at least ninety (90) days written notice to each Participating Physician, Participating Physician Group, or Participating Physician Organization affected thereby with whom the Blue Plan has directly contracted (except to the extent that a shorter notice period is required to comply with changes in applicable law), which notice shall reasonably apprise its Participating Physician, Participating Physician Group, or Participating Physician Organization of such change(s), and the change(s) shall not become effective before the conclusion of the notice period. If a Participating Physician, Participating Physician Group, or Participating Physician Organization objects to the change(s) that is subject to the notice, the Participating Physician, Participating Physician Group, or Participating Physician Organization must, within thirty (30) days of the date of the notice (which shall be the date the notice is sent by United States mail, by facsimile, or, if the Blue Plan offers it, electronically at the option of the Physician, Physician Group, or Physician Organization), give written notice to terminate his, her, or its contract with the Blue Plan, which termination shall be effective at the end of the notice period of the material adverse change unless, within sixty-five (65) days of the date of the original notice of change(s), the Blue Plan gives written notice to the objecting Participating Physician, Participating Physician Group, or Participating Physician Organization that it will not implement, as to the objecting Participating Physician, Participating Physician Group, or Participating Physician Organization, the material adverse change(s) to

which the Participating Physician, Participating Physician Group, or Participating Physician Organization objected. The continuation of care provisions in § 7.13(c) shall apply to any contract termination pursuant to this § 7.6.

Intentionally Left Blank 7.7

Disclosure of and Commitments Concerning Claims Payment 7.8 Practices

- Each Blue Plan recognizes the benefit of greater standardization in (a) its claims systems and, to that end, each Blue Plan expects to consolidate its claims systems in certain of its multi-state regions, where applicable, which will result in greater consistency with respect to its automated "bundling" and other claims payment rules within those regions.
- Each Blue Plan agrees that, except for Medicaid, State Childrens' (b) Health Insurance Programs, and other similar government programs for low-income persons and/or for members of stateestablished high risk pools, its automated "bundling" and other claims payment rules shall be consistent in all material respects, within each state in which such Blue Plan operates as a Blue Cross and/or Blue Shield licensee, for claims submitted by or on behalf of such Blue Plan's Plan Members.
- (c) Intentionally Left Blank
- Each Blue Plan agrees to disclose its Significant Edits on its (d) Provider Website by not later than six (6) months after the Final Order Date, or as soon thereafter as practicable. Each Blue Plan agrees to update its disclosure of Significant Edits once per calendar year to reflect changes in the Blue Plan's Significant Edits and the Blue Plan's experience with submitted claims; provided that the Blue Plan shall promptly disclose newly-adopted Significant Edits.
 - Not later than six (6) months after the Final Order Date or (i) as soon thereafter as practicable, each Blue Plan shall publish on its Provider Website, for each commercially available claims editing software product then in use by the Blue Plan, a list identifying each customized Edit added to the standard claims editing software product at the Blue Plan's request.
 - Not later than the Final Order Date, a Blue Plan shall not (ii) routinely require submission of Clinical Information, before or after payment of claims, in connection with that

Blue Plan's adjudication of a Physician's claims for payment, except as to claims for unlisted codes, claims to which a modifier 22 is appended, and other limited categories of claims as to which the Blue Plan determines that routine review of Clinical Information is appropriate; provided that the Blue Plan shall disclose any of its categories of such nature on its Public Website and its Provider Website. Notwithstanding the foregoing, a Blue Plan may require submission of Clinical Information in connection with a Blue Plan's adjudication of a Physician's claims for payment for the purpose of investigating fraudulent or abusive (whether intentional or unintentional) billing practices, but only so long as, and only during such times as, the Blue Plan has a reasonable basis for believing that such investigation is warranted. A Participating Physician may contest, in accordance with § 12, any requirement that the Participating Physician submit Clinical Information in connection with a Blue Plan's adjudication of the Participating Physician's claims for payment for the purpose of investigating fraudulent or abusive (whether intentional or unintentional) billing practices. Nothing in this Agreement is intended or shall be construed to limit a Blue Plan's right to require submission of Clinical Information when such requirement is not in connection with a Blue Plan's adjudication of a Physician's claims for payment or is otherwise permitted by this Agreement, including, but not limited to, the right to require submission of Clinical Information for Precertification purposes consistent with § 7.5.

- (iii) Not later than six (6) months after the Final Order Date, each Blue Plan shall publish on its Provider Website those limited code combinations as to which it has determined that particular services or procedures, relative to modifiers 25 and 59, are not appropriately reported together with those modifiers and the Blue Plan's application of the rule differs from CPT® Codes; provided that no such determination shall be inconsistent with the undertakings set forth in this Agreement.
- (e) Each Blue Plan shall promptly update the disclosures required by § 7.8(d)(i), (d)(ii) and (d)(iii) when changes are made to the policies, procedures, or determinations referenced therein.

7.9 Physician Advisory Committee

(a) Prior to the later to occur of (i) three (3) months after the Final Order Date and (ii) selection of the members of the Physician

Advisory Committee in accordance with § 7.9(b) of this Agreement, each Blue Plan shall take all actions reasonably necessary on its part to establish a Physician Advisory Committee ("Physician Advisory Committee") to discuss issues arising from or related to the relationships and interactions between and among Physicians, their patients, and the Blue Plan. These issues may include, but are not limited to: (a) improvement of health care and clinical quality; (b) improvement of communications, relations and cooperation between Physicians and the Blue Plan; and/or (c) matters of a clinical or administrative nature that impact the interaction between Physicians and the Blue Plan. The Physician Advisory Committee shall meet at least once every six months during the Effective Period. All communications to the Physician Advisory Committee by Participating Physicians and/or Non-Participating Physicians shall be accomplished through members of the Physician Advisory Committee who shall represent the interests of such Participating and/or Non-Participating Physicians and whose contact information shall be posted on the Provider Website.

(b) The Physician Advisory Committee shall include twelve (12) members, one of whom shall be the Blue Plan's Chief Medical Officer or his designee, who shall serve as chairperson of the Physician Advisory Committee.

Except as otherwise provided in this § 7.9(b), the remaining members of the Physician Advisory Committee shall be Participating Physicians.

The Blue Plan shall select three (3) members in addition to its Chief Medical Officer not later than sixty (60) days after the Preliminary Approval Date. The state medical society shall select four (4) members of a Blue Plan's Physician Advisory Committee with consultation, as such state society deems necessary, with appropriate organized medical and Physician professional organizations. The state medical society selected Physicians shall include at least one board-certified primary care Participating Physician, at least one board-certified specialist Participating Physician, and at least one Participating Physician who occupies a leadership position with a specialty medical society, state or local medical society, or large free-standing or hospital based group Physician practice. Those eight (8) members shall select the remaining four (4) members, one of whom may be a Non-Participating Physician. All members of the Physician Advisory Committee for a Blue Plan must practice or reside within the service area of such Blue Plan.

The Blue Plan and state medical society as provided above shall strive to select Physicians who are committed to the Physician

Advisory Committee functioning as a constructive and collaborative body. If any member discontinues serving on the Physician Advisory Committee, that member's position shall be filled in the same manner as the member was originally selected.

The names of the members of the Physician Advisory Committee and the dates of the Physician Advisory Committee meetings shall be posted on the Blue Plan's Provider Website.

(c) Any motion for the Physician Advisory Committee to consider an issue must be proposed by the chairperson or any other voting member of the Physician Advisory Committee. The issue shall be heard only if, at a meeting at which a quorum exists, a majority of the voting members of the Physician Advisory Committee present vote in favor of hearing the issue.

For purposes of this subparagraph (c), "quorum" shall mean seven (7) or more voting members of the Physician Advisory Committee of which at least two (2) members were selected by the state medical society as provided in § 7.9(b), two (2) members were selected by the Blue Plan, and two (2) members were selected by the members selected by the Blue Plan and the state medical society as provided in § 7.9(b).

Upon a majority vote of the voting members of the Physician Advisory Committee, the Physician Advisory Committee may make recommendations to the Blue Plan, provided that such recommendations are within the Physician Advisory Committee's purview as described in § 7.9(a).

The Blue Plan shall consider whether the implementation of any recommendation of the Physician Advisory Committee is: (a) reasonable considering the opportunities and constraints of the current health care financing/administration marketplace; (b) consistent with the best interests of the Blue Plan's Participating Physicians, Plan Members, customers, shareholders and other constituents; and (c) in furtherance of scientifically and clinically sound medical care. If the Blue Plan decides not to accept a recommendation of the Physician Advisory Committee, the Blue Plan shall communicate that decision in writing to the Committee with an explanation of the Blue Plan's reasons, and the Blue Plan shall also disclose the recommendation and response on its Provider Website. The Blue Plan agrees to post on its Provider Website a listing of all Physician Advisory Committee recommendations made to the Blue Plan and the Blue Plan's responses to such recommendations.

(d) Each member of the Physician Advisory Committee shall agree to maintain and treat as confidential any proprietary information

Page 32 of 58

Each Blue Plan shall develop and implement reasonable payment (e) provisions for the expenses of members of its Physician Advisory Committee, including without limitation a reasonable per diem to be set by the Blue Plan. Such payment provisions shall be consistent with the Blue Plan's typical payment provisions for Physicians serving on its existing committees of this type.

exclude Physicians who serve on credentialing or similar

New Dispute Resolution Process for Physician Billing Disputes 7.10

committees for other companies.

- Not later than four (4) months after the Final Order Date, each of (a) the Blue Plans shall take actions necessary to establish a Billing Dispute External Review Process. The Billing Dispute External Review Process shall provide for a Billing Dispute Reviewer, to resolve disputes with Physicians and Physician Groups arising from Covered Services provided to the Blue Plan's Plan Members by such Physicians and/or Physician Groups concerning (i) the Blue Plan's application of the Blue Plan's coding and payment rules and methodologies for fee for service claims (including without limitation any bundling, Downcoding, application of a CPT® modifier, and/or other reassignment of a code by the Blue Plan) to patient specific factual situations, including without limitation the appropriate payment when two or more CPT® Codes are billed together, or whether a payment enhancing modifier is appropriate, or (ii) any Retained Claims, so long as such Retained Claims are submitted by the Physician to the Billing Dispute Reviewer prior to the later to occur of (x) ninety (90) days after the Final Order Date, or (y) thirty (30) days after exhaustion of the Blue Plan's internal appeals process. Each such matter shall be a "Billing Dispute." The Billing Dispute Reviewer shall not have jurisdiction over any other disputes, including without limitation those disputes that fall within the scope of the External Review process set forth in § 7.11 of this Agreement, Compliance Disputes and disputes concerning the scope of Covered Services; nor shall any Billing Dispute Reviewer have jurisdiction or authority to revise or establish any reimbursement policy of the Blue Plan.
- Nothing contained in this § 7.10 is intended, or shall be construed, (b) to supersede, alter or limit the rights or remedies otherwise available to any Plan Member under § 502(a) of ERISA or to supersede in any respect the claims procedures for Plan Members

Page 33 of 58

of § 503 of ERISA, or required by applicable state or federal law or regulation. In the case of: (i) a state or federal law that requires the Blue Plan and Physician to use an external billing dispute review process, (ii) a state or federal law that requires the Blue Plan to establish and make available an external review process, and such process provides at least substantially the same procedural protections and rights as the process set forth in this § 7.10, and is a process that provides for disputes to be resolved at a cost to the Physicians that is not substantially greater than the cost set forth in this § 7.10 in time frames not materially longer than the time frames set forth in this § 7.10 and that any determinations are rendered by an independent, external person or entity, or (iii) an external review process that is established by a state or federal governmental body that the Blue Plan is required to make available and such process provides at least substantially the same procedural protection and rights as the process set forth in this § 7.10, only the program described in subclauses (i)-(iii) shall be utilized for Billing Disputes with respect to Plan Members covered by such process. Notwithstanding the foregoing, if there is any state or federal external review process that does not meet the requirements for use set forth herein, but is otherwise available to a Physician or Physician Group, the Physician or Physician Group shall be limited to bringing an appeal either under the process set forth in this § 7.10 or the available state or federal process, but not both.

- (c) Any individual Physician or Physician Group may submit a Billing Dispute to the Billing Dispute Reviewer after the Physician or Physician Group exhausts the Blue Plan's internal appeals process, and when the amount in dispute exceeds \$500. Billing Disputes may be submitted only by individual Physicians and Physician Groups. Each Blue Plan shall post a description of its Physician internal appeals process on its Provider Website.
 - (1) Notwithstanding the foregoing, an individual Physician or Physician Group may submit a Billing Dispute with an amount in dispute less than \$500 if such Physician or Physician Group notifies the Billing Dispute Reviewer that the Physician or Physician Group intends to submit additional Billing Disputes during the one (1) year period following the submission of the original Billing Dispute which involve issues that are similar to those of the original Billing Dispute. The Billing Dispute Reviewer will defer consideration of such Billing Dispute while the Physician or Physician Group accumulates such additional similar Billing Disputes. In the event that a Billing Dispute is deferred pursuant to the preceding sentence and, as of the Termination Date, the Physician or Physician Group has

not accumulated the requisite amount of Billing Disputes and the Blue Plan has chosen not to continue the Billing Dispute process following the Termination Date, then any rights the Physician or Physician Group had as to such Billing Disputes, including rights to arbitration, shall be tolled from the date the Billing Dispute was submitted to the Billing Dispute Reviewer through and including the Termination Date.

- In the event additional similar Billing Disputes (x) are not (2)submitted within one (1) year of the original Billing Dispute, or (y) do not involve disputes that in aggregate exceed \$500, the Billing Dispute Reviewer shall dismiss the original Billing Dispute and any such additional Billing Disputes.
- The Physician or Physician Group must exhaust the Blue (3) Plan's internal appeals process before submitting a Billing Dispute to the Billing Dispute Reviewer. A Physician or Physician Group shall be deemed to have exhausted the Blue Plan's internal appeals process if the Blue Plan does not communicate a decision on an internal appeal within thirty (30) days of the Blue Plan's receipt of all documentation reasonably needed to decide the internal appeal. In the event the Blue Plan and Physician or Physician Group disagree as to whether the requirements of the preceding sentence have been satisfied, such disagreement shall be resolved by the Billing Dispute Reviewer. Except as otherwise provided in this § 7.10(c), all Billing Disputes must be submitted to the Billing Dispute Reviewer no more than ninety (90) days after a Physician or Physician Group exhausts the Blue Plan's internal appeals process, and the Billing Dispute Reviewer shall not hear or decide any Billing Dispute submitted more than ninety (90) days after the Blue Plan's internal appeals process has been exhausted. The Blue Plan shall supply appropriate documentation to the Billing Dispute Reviewer not later than thirty (30) days after requested by the Billing Dispute Reviewer, which request shall not be made, if Billing Disputes are submitted pursuant to this § 7.10(c)(1), until Billing Disputes have been submitted with amounts in dispute that in aggregate exceed \$500.
- Except to the extent otherwise specified in this § 7.10(c), procedures for review by the Billing Dispute Reviewer, including without limitation the documentation to be supplied to the reviewer and the prohibition on ex parte communications between any party and the Billing

Document 928-2

Dispute Reviewer, shall be set by agreement between the Blue Plan and Class Counsel, or their designee, with input from the Billing Dispute Reviewer. Such procedures shall provide that (x) a Physician or Physician Group submitting a Billing Dispute to the Billing Dispute Reviewer shall state in the documents submitted the amount in dispute, and (y) that the Billing Dispute Reviewer shall not be permitted to issue an award that exceeds the greater of the amount in dispute stated by such Physician or Physician Group in the documents submitted to the Billing Dispute Reviewer or the amount payable under the terms of the applicable contract between the Blue Plan and the Physician or Physician Group.

- Each Blue Plan and Class Counsel shall select the person(s) or (d) organization(s) that shall serve as the Billing Dispute Reviewer (on a local, regional or national basis). If the Blue Plan and Class Counsel cannot agree on the Billing Dispute Reviewer within 120 days of the Preliminary Approval Date, the matter shall be deemed a Compliance Dispute and referred to the Compliance Dispute Officer. Billing Disputes shall be stayed and any time limitation shall be tolled pending resolution of such Compliance Dispute. Beginning one (1) year after any Billing Dispute Reviewer is selected by the Blue Plan and Class Counsel, either of Class Counsel or the Blue Plan may request the appointment of a new Billing Dispute Reviewer. The new person or organization to serve as the Billing Dispute Reviewer shall be selected within thirty (30) days of written notice of any such request. If the Blue Plan and Class Counsel cannot agree on the selection, the matter shall be deemed a Compliance Dispute and referred to the Compliance Dispute Review Officer. In deciding Billing Disputes, the Billing Dispute Reviewer shall be bound by the terms of the applicable Plan, any applicable agreement between the Physician or Physician Group and the Blue Plan, and the provisions of this Agreement. If the dispute cannot be resolved by reference to the foregoing documents, then the Billing Dispute Reviewer shall resolve Billing Disputes by determining, first, whether the billing was coded and submitted properly based on generally accepted medical coding standards, including but not limited to CPT® Coding and CCI/CMS guidelines, and second, whether the Blue Plan's reimbursement policies were properly applied, including those reimbursement policies required or permitted under this Agreement, including without limitation reimbursement policies posted by the Blue Plan pursuant to § 7.8(d).
- A Blue Plan's contract(s) with the person(s) or organization(s) (e) selected to serve as a Billing Dispute Reviewer shall require

- decisions to be rendered not later than thirty (30) days after receipt of the documents necessary for the review and to provide notice of such decision to the parties promptly thereafter.
- (f) In the event that the Billing Dispute Reviewer issues a decision requiring payment by the Blue Plan, that Blue Plan shall make such payment within fifteen (15) days after the Blue Plan receives notice of such decision.

Document 928-2

- Any decision under this § 7.10 shall be binding on the Blue Plan and the Physician or Physician Group. For Retained Claims, all Billing Disputes shall be directed not to the Court nor to any other state court, federal court, arbitration panel (except as hereinafter provided) or any other binding or non-binding dispute resolution mechanism but instead shall be submitted to final and binding resolution before the Billing Dispute Reviewer so long as such Billing Dispute arises after the selection of the Billing Dispute Reviewer pursuant to § 7.10(d). Retained Claims as defined in § 13.6 shall not be barred as untimely so long as they are submitted within thirty (30) days of the selection of the Billing Dispute Reviewer.
- (h) For any Billing Dispute that a Physician or Physician Group submits to the Billing Dispute Reviewer, the Physician or Physician Group submitting such Billing Dispute shall pay to the Blue Plan a filing fee calculated as follows: (i) if the amount in dispute is \$1,000 or less, the filing fee shall be equal to \$50; or (ii) if the amount in dispute exceeds \$1,000, the filing fee shall be equal to \$50 plus 5% of the amount by which the amount in dispute exceeds \$1,000, but in no event shall the fee be greater than 50% of the cost of the review. The Blue Plan shall refund the applicable filing fee paid by a Physician or Physician Group who submits a Billing Dispute to the Billing Dispute External Reviewer in the event the Physician or Physician Group is the prevailing party with respect to such Billing Dispute.
- (i) The determination made with respect to any Billing Dispute pursuant to this section shall not act as precedent as to any other Billing Dispute under this section or any other proceeding brought outside of this § 7.10, provided that evidence of the determination may be received in a Compliance Dispute involving a systemic remedy under § 12.6(f) of the Agreement.
- 7.11 Determinations Related to Medical Necessity or the Experimental or Investigational Nature of Any Proposed Health Care Service or Supply
 - (a) Initial Determinations

Case 1:03-cv-21296-FAM

A Physician designated by each Blue Plan shall be responsible for making the initial determination for such Blue Plan whether proposed health care services or supplies are Medically Necessary or experimental or investigational in nature. A nurse or other health care professional, acting for a medical director, may approve any proposed health care service or supply as being Medically Necessary, but only a Physician designated by the Blue Plan may deny any such service or supply as being not Medically Necessary or experimental or investigational in nature.

Plan Member Internal Appeal and External Review Process (b)

- Each Blue Plan shall maintain an internal appeal and (i) external review process permitting its Plan Members to seek internal and independent external review of any determination made by the Blue Plan that certain services provided to the Blue Plan's Plan Members by Physicians are not Covered Services because they are not Medically Necessary or are experimental or investigational in nature ("Adverse Determination") where the Blue Plan both makes the Adverse Determination and administers its Plan Member appeals and external review processes.
- As set forth in this § 7.11, each Blue Plan will establish and (ii) maintain an internal appeal and external review process for Physicians with respect to Adverse Determinations to the extent the Blue Plan both makes the Adverse Determination and administers its Plan Member appeals and/or external review processes.
- Except where any applicable law or regulation requires a (iii) different definition, each Blue Plan shall use the definition of Medical Necessity set forth in § 7.16 (a) of this Agreement in the internal appeal and external review processes set forth in this § 7.11; provided, however, that nothing in this Agreement shall: (a) limit or prevent the Blue Plan from denying coverage on the grounds that services are experimental or investigational; or (b) alter or restrict the Blue Plan's rights under its contracts with Participating Physicians to restrict or prohibit them from billing the Blue Plan's Plan Member for services determined to be not Medically Necessary or experimental or investigational. Each Blue Plan agrees that Physicians may bill its Plan Members for services determined to be not Medically Necessary or experimental or investigational when the Physician provides the Blue Plan's Plan Member with advance written notice that (a) identifies the proposed services, (b) informs that Plan Member that such services

may be deemed by the Blue Plan to be not Medically Necessary or experimental or investigational, and (c) provides an estimate of the cost to that Plan Member for such services and that Plan Member agrees in writing in advance of receiving such services to assume financial responsibility for such services.

(iv) In applying experimental and investigational exclusions in a Plan to either proposed health care services or as part of a Post-Service Appeal to the Blue Plan, each Blue Plan shall consider credible scientific evidence published in peer-reviewed medical literature generally recognized by the relevant medical community, Physician Specialty Society recommendations, the views of Physicians practicing in relevant clinical areas, the individual clinical circumstances of the particular Blue Plan's Plan Member, the views of the treating Physician and any other relevant factors.

(c) Physician Internal Appeals of Adverse Determinations.

(i) Pre-Service Appeals.

Physicians shall have the right to file an appeal of an Adverse Determination prior to rendering the service ("Pre-Service Appeals"), if they are appealing on behalf of the Blue Plan's Plan Member. For urgent Pre-Service Appeals, the Physician shall be automatically deemed the authorized representative of the Blue Plan's Plan Member. For all other Pre-Service Appeals, authorization must be obtained from the Blue Plan's Plan Member in writing. Pre-Service Appeals filed by Physicians on behalf of the Blue Plan's Plan Member will be handled by the Blue Plan under the appeal process available to its Plan Member based on the terms of that Plan Member's Plan and the applicable state and federal laws and regulations.

(ii) Post-Service Appeals

(A) With respect to an appeal of an Adverse
Determination made after the service has been
rendered ("Post-Service Appeals"), each Blue Plan
shall adopt a one level internal appeal process for
Physicians. That process shall ensure that only a
Physician in the same specialty, as defined in
subsection (B) below, as the Physician who treated
the condition (hereinafter "Qualified Reviewer"),
other than the Physician that made the initial
Adverse Determination, may deny the appeal of the
Physician who treated the condition. A nurse or

other health care professional employed by the Blue Plan may review the internal appeal and may grant but not deny the appeal. If the nurse or other healthcare professional does not grant the appeal, then a Qualified Reviewer, designated by the Blue Plan, other than the one that made the initial Adverse Determination, shall review and decide the internal appeal in accordance with the Blue Plan's health care clinical guidelines.

- (B) For purposes of this section, "same specialty" shall mean a Physician with similar credentials and licensure as those who typically treat the condition or health problem in question in the appeal or a Physician who has experience treating the same problems as those in question in the appeal, in addition to experience treating similar complications of those problems.
- (C) Prior to requesting an internal Post-Service Appeal, the Physician shall use best efforts to first seek written authorization to proceed as the representative of the Blue Plan's Plan Member. If the Physician obtains consent to proceed on behalf of the Blue Plan's Plan Member, then the Physician's appeal rights are those of the Blue Plan's Plan Member and the Physician is bound by the decision rendered in the Plan Member's appeal process.

(d) Timeframes for Physician Internal Appeals of Adverse Determinations

All internal Post-Service Appeals filed by Physicians shall be adjudicated within the time limits established under regulations issued by the Department of Labor regardless of whether ERISA applies.

(e) Adverse Determination External Review Process for Physicians

(i) If a Blue Plan upholds its initial Adverse Determination through the internal Post-Service Appeals process and the cost of the service at issue exceeds the threshold amount, if any, the Blue Plan's Plan Member would need to satisfy in order to seek external review under the terms of the applicable health benefit plan, the Blue Plan shall make available to its Physicians the option to seek external review of the Adverse Determination through an independent review organization ("Independent Review Organization") identified by the Blue Plan ("External")

- Review"). The Physician shall have the option to submit a written request for External Review within sixty (60) days from the date of the internal Post-Service Appeal denial decision by the Blue Plan. Election to pursue External Review under this § 7.11 is at the option of the Physician, who may instead choose any other remedy available as a matter of law or contract.
- (ii) External Review is not available for a Physician before that Physician has exhausted the Blue Plan's internal Post-Service Appeal process unless both the Blue Plan and the Physician agree to forego the internal Post-Service Appeal and proceed directly to External Review, or the Blue Plan cannot provide a Qualified Reviewer for internal appeal.
- (iii) Physicians seeking External Review shall pay the Blue Plan a filing fee of fifty dollars (\$50) if the amount in dispute is \$1,000 or less or \$250 if the amount in dispute exceeds \$1,000. Payment must be submitted along with the External Review request; provided, however, that the Physician shall be entitled to a refund of such payment in the event that the Physician prevails in the Blue Plan's External Review process.
- (iv) Any decision issued pursuant to an External Review process, regardless of whether such External Review process is initiated and pursued by the Blue Plan's Plan Member or the Physician, shall be binding upon both the Physician and the Blue Plan.
- The Blue Plan will contract with the Independent Review (v) Organization (on a local, regional, or national basis) to conduct a de novo review of the case. For coverage issues other than a determination of Medical Necessity, the Plan Documents of the Blue Plan's Plan Member will control. In the event an External Review process is initiated, the Blue Plan shall promptly, but in any event no later than ten (10) business days following receipt of the request, submit documentation pertaining to the appeal to the Independent Review Organization. The Blue Plan shall require that the Independent Review Organization provide a decision within thirty (30) days of the Blue Plan's submission of all necessary information. The external reviewer designated to conduct the review by the Independent Review Organization shall be of the same specialty (but not necessarily the same sub-specialty) as the appealing Physician.
- (vi) The Independent Review Organization's compensation shall not be tied to the outcome of the reviews performed. Likewise, the selection process among qualified

Page 41 of 58

- Independent Review Organizations will not create any incentives for Independent Review Organizations to make decisions in a biased manner.
- In the event that: (A)(i) a state or federal law requires the (vii) Blue Plan and Physician to use an external review process, (ii) a state or federal law requires the Blue Plan to establish and make available an external review process, and such process provides at least substantially the same procedural protections and rights as the process set forth in this § 7.11, and is a process that provides for disputes to be resolved at a cost to the Physicians that is not substantially greater than the cost set forth in this § 7.11(e) in time frames not materially longer than the time frames set forth in this § 7.11(e), and that any determinations are rendered by an independent, external person or entity, or (iii) an external review process is established by a state or federal governmental body that the Blue Plan is required to make available, and such process provides at least substantially the same procedural protections and rights as the process set forth in this § 7.11, as set forth in § 7.11 (e)(ii) above, and (B) any of these processes are: (i) made available to Physicians without the written authorization of the Blue Plan's Plan Member, or (ii) available to the Physician with the written authorization of the Blue Plan's Plan Member and the Physician has obtained such written authorization, then the Physician shall be required to use the state or federal external review process in lieu of the external review process established in this § 7.11. Notwithstanding the foregoing, if there is any state or federal external review process that does not meet the requirements for use set forth herein, but is otherwise available to a Physician or Physician Group, the Physician or Physician Group shall be limited to bringing an appeal either under the process set forth in this § 7.11 or the available state or federal process, but not both.
- Notwithstanding the preceding provisions of this § 7.11, (viii) and in addition to any requirements contained above, the Physician may not initiate an internal Post-Service Appeal or External Review of any denied service if:
 - The Blue Plan's Plan Member (or his or her (A) representative) or the Physician (either independently where the Blue Plan is required to accept an independent Physician appeal by state law or as the representative of the Blue Plan's Plan Member) filed a Pre-Service Appeal pertaining to the same denied service; or

Page 42 of 58

- Document 928-2
 - The Blue Plan's Plan Member (or his or her (B) representative) is currently seeking or has sought review related to the same denied service. In the event both Blue Plan's Plan Member (or his or her representative) and the Physician seek review of the same denied service, the Plan Member's review shall go forward and the Physician's request for review will be dismissed;
 - As to External Review only, the Blue Plan's Plan (C) Member is covered under a Self-Insured Plan and the Plan sponsor has not agreed by contract to participate in the Blue Plan's External Review program set forth in this § 7.11(e); or
 - (D) The Blue Plan's Plan Member (or his or her representative) has filed suit under § 502(a) of ERISA or other suit for the denial of health care services or supplies regarding an Adverse Determination. In that event, or if such a suit is subsequently initiated, the Plan Member's lawsuit shall go forward and the Physician's claims shall be dismissed and may not be brought by or on behalf of the Physician in any forum; provided that such dismissal shall be without prejudice to any Physician seeking to establish that the rights sought to be vindicated in such lawsuit belong to such Physician and not to such Blue Plan's Plan Member.
 - Nothing contained in this § 7.11 is intended, or shall (E) be construed, to supersede, alter or limit the rights or remedies otherwise available to any Person under § 502(a) of ERISA or to supersede in any respect the claims procedures under § 503 of ERISA.

Precedential Effect (f)

The determination made with respect to any Adverse Determination pursuant to any internal appeal and External Review process referenced in this § 7.11 shall not act as precedent as to any other Medical Necessity or experimental or investigational determination under this § 7.11 or any other proceeding brought outside of this § 7.11, provided that evidence of the determination may be received in a Compliance Dispute involving a systemic remedy under § 12.6(f) of the Agreement.

7.12 Intentionally Left Blank

Participating in Blue Plans' Networks 7.13

Case 1:03-cv-21296-FAM

(a) Credentialing of Physicians

Each Blue Plan will allow Physicians to submit credentialing applications (including, as relevant, licensure and hospital privileges or other required information) and will begin to process such applications prior to the time that the Physician formally changes or commences employment or changes location, provided that the Physician must represent that he or she has new employment or intends to move to a new location within the Blue Plan's service area. The Blue Plan shall complete primary source verification and notify the Physician as to whether he or she is credentialed within ninety (90) days of receiving a Physician's completed application to be a Participating Physician unless in spite of the Blue Plan's best efforts and because of a failure of a third party to provide necessary documentation, the Blue Plan cannot obtain the necessary information to make a decision within ninety (90) days. In such event, the Blue Plan shall make every effort to obtain such information as soon as possible. Each Blue Plan commits that the Credentialing Committee for its service area shall meet at least once every forty-five (45) days to consider credentialing applications for which primary source verification has been completed. If a Physician is already credentialed by the Blue Plan but changes employment or changes location, the Blue Plan will only require the submission of such additional information, if any, as is necessary to continue the Physician's credentials based upon the changed employment or location.

(b) All Products Clauses

Each Blue Plan agrees that it shall not require a Participating Physician to participate in a capitated fee arrangement in order to participate in Product Networks in which such Participating Physician is compensated on a fee for service basis. Each Blue Plan further agrees that it shall not require a Participating Physician to participate in its Medicare Advantage or Medicaid Product Networks in order to participate in its commercial Product Networks. Except where a Participating Physician (or Participating Physician Group comprised of Participating Physicians or Participating Physician Organization) has agreed in an Individually Negotiated Contract to participate in more than one Product Network for a specified period of time (in which case the terms of such Individually Negotiated Contract shall govern), if a Participating Physician (or Participating Physician Group comprised of Participating Physicians or Participating Physician Organization) either (a) chooses not to participate in all of the Blue Plan's Product Networks or (b) terminates participation in some of the Blue Plan's Product Networks, then the reimbursement levels

Page 44 of 58

(e.g., fee for service maximum allowable amount, Capitation rate or other reimbursement methodology) offered to or applied by the Blue Plan to such Participating Physician (or Participating Physician Group or Participating Physician Organization) for the Product Network(s) in which such Participating Physician (or Participating Physician Group or Participating Physician Organization) continues to participate shall not be lower than the Blue Plan's standard reimbursement levels (e.g., fee for service maximum allowable amount, Capitation rate or other reimbursement methodology) in that geographic market. Notwithstanding the foregoing, the Blue Plan may offer a higher reimbursement level (e.g., fee for service maximum allowable amount, Capitation rate or other reimbursement methodology) or other incentive to any Participating Physician (or Participating Physician Group or Participating Physician Organization) who elects to participate (or elects to continue participation) in more than one of the Blue Plan's Product Networks. Nothing in this paragraph shall obligate a Blue Plan to pay more than the lesser of the Participating Physician's billed charges or the Blue Plan's applicable fee for service amount.

(c) Termination Without Cause

Unless an Individually Negotiated Contract between a Blue Plan and a Participating Physician, Participating Physician Group, or Participating Physician Organization specifies a different period of notice, or specifies that the contract may not be terminated except for cause during a defined period of time, either party to a contract between a Blue Plan and a Participating Physician, Participating Physician Group, or Participating Physician Organization shall have the right to terminate the contract without cause upon prior written notice provided to the other party which notice shall be a definite period set forth in such agreement, which period shall be no less than sixty (60) or more than one hundred and twenty (120) calendar days.

In the event of a contract termination by either party, the following obligations shall apply with respect to the continuation of care for those patients of a Participating Physician, Participating Physician Group, or Participating Physician Organization who are entitled to continuation of care as reasonably defined under the Participating Physician's, Participating Physician Group's, or Participating Physician Organization's contract with the Blue Plan or under applicable law. In the case of a continuation of care situation as defined in the preceding sentence, the Participating Physician, Participating Physician Group, or Participating Physician Organization shall continue to render necessary care to the Blue Plan's Plan Member consistent with contractual or legal

Page 45 of 58

obligations; provided that, if, upon notice from the Participating Physician, Participating Physician Group, Participating Physician Organization, or the Blue Plan's Plan Member that a Plan Member is in a continuation of care situation, the Blue Plan does not use due diligence to make alternative care available to the Plan Member within ninety (90) days after receipt of such notice, then for continuation of care services provided after termination, the Blue Plan shall pay to the Participating Physician, Participating Physician Group, or Participating Physician Organization the standard rates paid to Non-Participating Physicians for that geographical area. Other than as specified in this § 7.13(c), the contractual provisions applicable to continuation of care shall apply.

Notwithstanding the foregoing obligations, a Blue Plan's obligations under this § 7.13(c) shall not apply to the extent that other Participating Physicians, Participating Physician Groups, or Participating Physician Organizations are not available to replace the terminating Physician, Physician Group, or Physician Organization due to: (i) geographic or travel-time barriers; or (ii) contractual provisions between the terminating Physician, Physician Group, or Physician Organization and a facility at which the Blue Plan's Plan Member receives care that limits or precludes other Participating Physicians, Participating Physician Groups, or Participating Physician Organizations from rendering replacement services to the Blue Plan's Plan Members (e.g., an exclusive services agreement between the terminating Physician, Physician Group, or Physician Organization and a facility where Plan Member receives services).

Fee Schedule Changes 7.14

Notices Regarding Fee Schedules (a)

Each Blue Plan agrees to establish and operate one or more standard fee schedules of fee for service payments to its Participating Physicians, Participating Physician Groups, and Participating Physician Organizations for each geographic market in which the Blue Plan maintains a Product Network. Each Blue Plan also agrees, effective January 1 of the year following the Effective Date, not to reduce the fees set forth in such fee schedules more than once per calendar year except as otherwise provided in this § 7.14(a). Each Blue Plan further agrees that it shall give notice of any such reductions in fees as a material adverse change subject to the provisions of § 7.6 hereof; provided, however, that to the extent a fee schedule is directly tied to the CMS fee schedules or state Medicaid fee schedules currently in effect, it shall adjust automatically to reflect applicable interim and annual revisions made by CMS or the state Medicaid agency without notice to the Physician. If an annual revision made by CMS or a state Medicaid agency results in a reduction in the fees in a fee schedule that is directly tied to the CMS fee schedules or state Medicaid fee schedules, a Participating Physician shall have the right to terminate his or her contract with the Blue Plan by giving the Blue Plan written notice of termination within thirty (30) days of the date on which CMS or the state Medicaid agency published notice of the annual revision, which termination shall be effective ninety (90) days after the date that such notice was published, subject to the continuation of care obligations described in § 7.13(c). Notwithstanding the foregoing, the Blue Plan may increase or reduce the fees set forth in such fee schedules by updating its fee schedules at any time (i) to reflect changes in market prices for vaccines, injectibles, pharmaceuticals, durable medical supplies, other goods, and non-Physician services, (ii) to add payment rates for newly-adopted CPT® Codes, (iii) to add payment rates for new technologies and new uses of established technologies that the Blue Plan concludes are eligible for payment, and (iv) to reflect applicable interim revisions made by CMS. Nothing contained in this § 7.14(a) shall prevent the Blue Plan from maintaining, altering or expanding the use of Capitation or other compensation methodologies. If an Individually Negotiated Contract has a term greater than one (1) year and has specific provisions for compensation that substantially differ from the standard fee schedule and that do not permit reduction in the compensation more frequently than once per year, then the provisions of § 7.14(a) shall not apply to such Individually Negotiated Contract.

(b) Payment Rules for Injectibles, DME, Administration of Vaccines, and Review of New Technologies

Each Blue Plan agrees to pay a fee for the administration of vaccines and injectibles by a Participating Physician. The Blue Plan also agrees to pay for the vaccines and injectibles themselves. The Blue Plan shall pay for newly recommended vaccines as of the effective date of a recommendation made by any of the following: the U.S. Preventive Services Task Force; the American Academy of Pediatrics; and the Advisory Committee on Immunization Practices. Other than as specified in the preceding sentence with respect to newly recommended vaccines, if a Physician Specialty Society recommends as an appropriate standard of care a new technology or treatment, or a new use for an established technology or treatment, the Blue Plan shall evaluate such recommendation and issue a coverage statement not later than 120 days after the Blue Plan learns of such Physician Specialty Society recommendation.

(c) Usual, Reasonable, and Customary Appeals

At least until the Termination Date, in the event the Plan Documents require that a Blue Plan utilize a usual, reasonable and customary standard for purposes of reimbursement to a Plan Member for services rendered by Non-Participating Physicians, and if a Non-Participating Physician initiates a dispute using a Blue Plan's internal dispute resolution procedures over how the Blue Plan has determined the usual, reasonable and customary amount for a given health care service or supply, and, consequently, over how the Blue Plan has computed the amount payable for that health care service or supply, the Blue Plan shall disclose to the Non-Participating Physician initiating the dispute the general methodology, including the percentile of included charge data on which the usual, reasonable and customary amount is based, and source of data used by the Blue Plan to determine the usual, reasonable and customary amount for that service or supply.

(d) Usual, Reasonable and Customary Determinations

Each Blue Plan agrees that, to the extent it uses Physician charge data to determine the usual, reasonable and customary amount to be paid for services performed by Non-Participating Physicians, it will not use any internal claims database that: (i) systematically under-reports the number of claims paid for procedures in the geographic area used by the Blue Plan to determine such amount; (ii) eliminates or excludes the highest charges for paid claims for any procedures in the geographic area used by the Blue Plan to determine such fees, provided, however, that such charges may be excluded if the Blue Plan excludes an equivalent number or percentage of the lowest charges for such procedures, or reasonably determines that any such charges are the result of erroneous data; (iii) includes charges for procedures performed in a geographic area other than the one used by the Blue Plan to determine such amount, provided, however, that such charges may be considered where the Blue Plan determines there is an insufficient number of charges in the relevant geographic area to reasonably determine a usual, reasonable and customary amount; (iv) calculates the usual, reasonable and customary amount based upon fees paid under a discounted fee schedule rather than billed charges; and (v) lacks quality controls sufficient to reasonably test the validity of the data included in the database.

7.15 Intentionally Left Blank.

7.16 Application of Clinical Judgment to Patient-Specific and Policy Issues

Page 48 of 58

Patient-specific Issues Involving Clinical Judgment (a)

Medical Necessity Definition

Document 928-2

Except where any applicable law or regulation requires a different definition, each Blue Plan shall apply as to its current agreements and include in its future agreements with Participating Physicians the following definition of "Medically Necessary" or comparable term in each such agreement: "Medically Necessary" or "Medical Necessity" shall mean health care services that a Physician, exercising prudent clinical judgment, would provide to a patient for the purpose of preventing, evaluating, diagnosing or treating an illness, injury, disease or its symptoms, and that are: (a) in accordance with generally accepted standards of medical practice; (b) clinically appropriate, in terms of type, frequency, extent, site and duration, and considered effective for the patient's illness, injury or disease; and (c) not primarily for the convenience of the patient, physician, or other health care provider, and not more costly than an alternative service or sequence of services at least as likely to produce equivalent therapeutic or diagnostic results as to the diagnosis or treatment of that patient's illness, injury or disease. For these purposes, "generally accepted standards of medical practice" means standards that are based on credible scientific evidence published in peer-reviewed medical literature generally recognized by the relevant medical community, Physician Specialty Society recommendations and the views of Physicians practicing in relevant clinical areas and any other relevant factors.

Adverse Determination Denial Rate

For the calendar year beginning after the Final Order Date, and thereafter during the Effective Period, each Blue Plan shall make an annual, aggregate disclosure of the number of Adverse Determinations sent to external review for final determination for the preceding calendar year and the percentage of such Adverse Determinations that are upheld or reversed. Each Blue Plan shall make this disclosure by means of its Provider Website or other comparable electronic medium.

Policy Issues Involving Clinical Judgment (b)

In formulating and adopting medical policies with respect to Covered Services, each Blue Plan shall rely on credible scientific evidence published in peer-reviewed medical literature generally recognized by the relevant medical community, and shall continue to make such policies readily available to its Plan Members and Participating Physicians via its Public Website or by other electronic means. In formulating and adopting such policies, each

Page 49 of 58

Blue Plan shall take into account national Physician Specialty Society recommendations and the views of prudent Physicians practicing in relevant clinical areas and any other clinically relevant factors. Promptly after adoption, each Blue Plan shall file a copy of each new policy or guideline with its Physicians' Advisory Committee.

(c) Future Consideration by Blue Plans of an Administrative Exemption Program

Each Blue Plan shall consider the feasibility and desirability of exempting certain Participating Physicians from certain administrative requirements based on criteria such as the Participating Physician's delivery of quality and cost effective medical care and accuracy and appropriateness of claims submissions. No Blue Plan shall be obliged to implement any such exemption process during the term hereof, and this § 7.16(c) is not intended and shall not be construed to limit a Blue Plan's ability to implement any such program on a pilot or experimental basis, base exemptions on any Blue Plan determined basis, or otherwise to implement one or more programs in only some markets.

7.17 Billing and Payment

(a) Time Period for Submission of Bills for Services Rendered

No Blue Plan shall contest the timeliness of bills for Covered Services provided under a Fully-Insured Plan if such bills are received by the Blue Plan within 180 days after: (i) the date of service when the Blue Plan is the primary payor; or (ii) the date of the Physician's receipt of the EOB from the primary payor when the Blue Plan is the secondary payor. Each Blue Plan shall propose to Self-Insured Plan sponsors that they adopt the 180 day time period referenced in the preceding sentence, in the event that a Self-Insured Plan has a more restrictive time period. Each Blue Plan shall extend the 180 day time period for a reasonable period, on a case by case basis, in the event that a Physician provides notice to the Blue Plan, along with appropriate evidence, of circumstances reasonably beyond the Physician's control that resulted in the delayed submission. The Blue Plan shall determine such circumstances and the reasonableness of the submission date. Nothing in this § 7.17(a) shall limit a Blue Plan's ability to provide incentives for prompt submission of bills.

(b) Claims Submission

Each Blue Plan agrees to accept from Participating Physicians and Non-Participating Physicians properly completed paper claims

Page 50 of 58

submitted on Form CMS-1500 or the equivalent. Each Blue Plan also agrees to accept electronic claims populated with similar information in HIPAA-compliant format using HIPAA-compliant code sets, subject to the Blue Plan's reasonable requirements pertaining to the exchange of electronic transactions. Each Blue Plan shall allow any of its Participating Physicians, at the Physician's election, to engage in any electronic transaction for which a standard transaction has been established by the HIPAA Standard Transactions and Code Sets Rule. If a Physician elects not to be compliant with the portions of HIPAA relating to the electronic submission of claims, no Blue Plan shall require such Physician to use electronic transactions or otherwise require such Physician to become compliant with HIPAA. Instead, the Blue Plan will maintain reasonable non-electronic systems to serve the information needs of such Physicians. Notwithstanding the above, a Blue Plan may continue to require submission of Clinical Information and other additional information in connection with its review of specific claims and as contemplated elsewhere in this Agreement, including without limitation §§ 7.5 and 7.8(d)(ii); provided, however, that nothing in this sentence is intended or shall be construed to alter or limit any restrictions set forth elsewhere in this Agreement concerning the Blue Plan's ability to make requests for Clinical Information in connection with adjudication of claims. Each Blue Plan shall disclose on its Provider Website and its Public Website its policies and procedures regarding (i) the appropriate format for claims submissions, and (ii) requests for Clinical Information.

Timelines for Processing and Payment of Complete Claims 7.18

Beginning not later than nine (9) months after the Final Order (a) Date, each Blue Plan shall direct the issuance of a check or an electronic funds transfer in payment for Complete Claims for Covered Services within thirty (30) calendar days following the later of the Blue Plan's receipt of such claim or the date on which the Blue Plan is in receipt of all information needed and in a format required for such claim to constitute a Complete Claim and is in receipt of all documentation which may be requested by a Blue Plan consistent with this Agreement and which is reasonably needed by the Blue Plan: (i) to determine that such claim does not contain any material defect or error; or (ii) to make a payment determination. Beginning one (1) year following the Effective Date, each Blue Plan shall direct the issuance of a check or an electronic funds transfer in payment for Complete Claims for Covered Services that are submitted electronically by Physicians within fifteen (15) business days following the later of the Blue Plan's receipt of such claim or the date on which the Blue Plan is in receipt of all information needed and in a format required for

Document 928-2

such claim to constitute a Complete Claim and is in receipt of all documentation which may be requested by a Blue Plan consistent with this Agreement and which is reasonably needed by the Blue Plan: (i) to determine that such claim does not contain any material defect or error; or (ii) to make a payment determination. If direction for the issuance of payment for Complete Claims for Covered Services is not made within the time periods specified in this § 7.18(a), the Blue Plan shall pay interest pursuant to § 7.18(b). For purposes of this § 7.18, where a state provides a definition of "clean claim" or "complete claim" which varies from the definition of Complete Claim provided herein, such definition shall be substituted for the definition of "Complete Claim" herein.

- For each Complete Claim with respect to which a Blue Plan has (b) directed the issuance of a check or the electronic funds transfer later than the applicable period specified in § 7.18(a), the Blue Plan shall pay interest at the rate of eight percent (8%) per annum on the balance due on each such claim computed from the sixteenth (16th) business day or the thirty-first (31st) calendar day (as appropriate based on the circumstances described in § 7.18(a)) up to the date on which the Blue Plan directs the issuance of the check or the electronic funds transfer for payment of such Complete Claim; provided that to the extent that payment is made later than the period specified by applicable law or regulation, the Blue Plan shall pay interest at any rate specified by such law or regulation in lieu of the interest payment otherwise contemplated by this sentence. Interest paid pursuant to this § 7.18(b) shall, at the Blue Plan's election, either be included in the claim payment check or wire transfer or be remitted periodically (but at least quarterly) in a separate check or wire transfer along with a report detailing the claims for which interest is being paid.
- No Blue Plan shall have an obligation to make any interest (c) payment pursuant to § 7.18(b) (i) with respect to any Complete Claim if, within thirty (30) days of the submission of an original claim, a duplicate claim is submitted while adjudication of the original claim is still in process; (ii) to any Participating Physician who balance bills a Plan Member in violation of such Participating Physician's agreement(s) with the Blue Plan; (iii) with respect to any time period during which a Force Majeure, as defined in § 7.32 of this Agreement, prevents adjudication of claims; or (iv) where payment is made to a Plan Member.
- Each Blue Plan shall affix to or on paper claims for Covered (d) Services, or otherwise maintain a system for determining, the date such claims are received by the Blue Plan. Each Blue Plan shall send an electronic acknowledgement of claims for Covered Services submitted electronically identifying the date such claims

Page 52 of 58

are received by the Blue Plan. If a Blue Plan determines that there is any defect or error in a claim that prevents the claim from entering the Blue Plan's adjudication system, it shall provide notice within ten (10) days of receipt of such claim. Nothing contained in this § 7.18 is intended or shall be construed to alter a Blue Plan's ability to request Clinical Information consistent with the provisions of § 7.8(d)(ii) or any other provision of this Agreement.

- Notwithstanding anything in the Agreement to the contrary, the (e) requirements of § 7.18 shall not apply to (i) claims for Covered Services that are processed under the BlueCard Program or any similar national account delivery program governed by the BCBSA (including but not limited to NASCO-to-NASCO arrangements) in which the Blue Plan participates but is not solely responsible for the processing and payment of the claim, and/or (ii) claims for Covered Services under a program offered or sponsored by any state or federal governmental entity other than in its capacity as an employer.
- A claim relating to § 7.18(a), where the claim and any interest (f) payment required under § 7.18(b) has been made, shall not be the basis for a Compliance Dispute.

No Automatic Downcoding of Evaluation and Management Claims 7.19

As of the Final Order Date, no Blue Plan shall automatically reassign or reduce the code level of evaluation and management codes billed for Covered Services ("Downcoding"), except that a Blue Plan may reassign a new patient visit code to an established patient visit code based solely on CPT® Codes, CPT® Guidelines, and CPT® Conventions. Notwithstanding the foregoing sentence, Blue Plans shall continue to have the right to deny, pend or adjust such claims for Covered Services on other bases and shall have the right to reassign or reduce the code level for selected claims for Covered Services (or claims for Covered Services submitted by selected Physicians or Physician Groups or Physician Organizations) based on a review of the information in the Clinical Information at the time the service was rendered for the particular claims or a review of information derived from a Blue Plan's fraud or abuse billing detection programs that create a reasonable belief of fraudulent or abusive (whether intentional or unintentional) billing practices; provided that the decision to reassign or reduce is based primarily on a review of Clinical Information.

Bundling and Other Computerized Claim Editing 7.20

Each Blue Plan agrees to take actions necessary on the Blue Plan's part to cause the claim-editing software program it uses to continue to produce

Page 53 of 58

Case 1:03-cv-21296-FAM

editing results consistent with the standards set forth in this § 7.20 and, if the Blue Plan has actual knowledge of non-conformity with such standards, to take reasonable actions necessary on its part to promptly modify such software to any extent necessary to conform to such standards; provided that nothing in this paragraph is intended or shall be construed to require a Blue Plan to pay for anything other than Covered Services for its Plan Members, to make payment at any particular rates, to limit a Blue Plan's right to deny, pend or adjust claims based on a reasonable belief of fraudulent or abusive (whether intentional or unintentional) billing practices (so long as the Physician has been given the opportunity to provide Clinical Information and the Blue Plan has reviewed any Clinical Information so provided before denying or adjusting the claims). For purposes of this § 7.20 only, if any change to CPT® affects a Blue Plan's obligations hereunder, the Blue Plan will promptly develop plans to cause its Physician payment practices to be consistent with the commitments set forth in this § 7.20. The obligations set forth below in this § 7.20 shall take effect on the dates set forth in Exhibit I.

For purposes of §§ 7.19 and 7.20, all references to the AMA CPT® book and to CPT® Codes in this Agreement refer to the AMA CPT® book and the CPT® Codes listed in the AMA CPT® book in effect at the time the services were provided; however, notwithstanding anything to the contrary contained in this Agreement, if there is any amendment, modification, or superseding change to the CPT® Codes, CPT® Conventions or CPT® Guidelines, which constitutes a departure from the procedures, criteria or scope of activities historically employed or undertaken by the AMA's CPT Editorial Panel and which materially expands the commitments of a Blue Plan in this Agreement or has a material adverse effect on the Blue Plan, then the Blue Plan may communicate to Physicians its decision not to recognize such modification by posting a notification on its website, or by other form of written or electronic communication to Physicians, and by notifying Class Counsel.

- (a) Each Blue Plan will process and separately reimburse those codes listed in the AMA CPT® book as modifier 51 exempt CPT® Codes without reducing payment under the Blue Plan's Multiple Procedure Logic, provided that the AMA CPT® book provides that such services are appropriately reported together.
- (b) Each Blue Plan will process and separately reimburse codes listed in the AMA CPT® book as add-on billing codes without reducing payment under the Blue Plan's Multiple Procedure Logic; provided that the AMA CPT® book provides that such add-on CPT® Codes are appropriately billed with proper primary procedure codes.
- (c) (i) No Blue Plan shall require a Physician to submit Clinical Information of their patient encounters solely because the

Physician seeks payment for both surgical procedures and CPT® evaluation and management services for the same patient on the same date of service, provided that the correct CPT® evaluation and management code, surgical code and modifier (e.g., CPT® modifiers 25 or 57) are included on the initial claim submission.

- (ii) If a bill contains a CPT® Code for an evaluation and management service, appended with a CPT® modifier 25 and a CPT® Code for performance of a non-evaluation and management service procedure code, both codes shall be recognized and separately eligible for payment, unless the Clinical Information indicates that use of the CPT® modifier 25 was inappropriate or the Blue Plan has disclosed pursuant to § 7.8(d)(iii) the limited number of finite code combinations that are not appropriately reported together. Payment shall only be made for one evaluation and management service for any single day unless payment for more than one is appropriate pursuant to the AMA CPT® book and is supported by appropriate diagnoses in the Clinical Information.
- (iii) Each Blue Plan will remove from its claim review and payment systems any Edits that generally deny payment for CPT® evaluation and management codes with a CPT® modifier 25 appended when submitted with surgical or other procedure codes for the same patient on the same date of service except for a limited number of exceptions, consistent with § 7.20(c)(ii) above, which will be disclosed on the Blue Plan's Provider Website.
- (iv) Nothing in this Agreement shall (A) prohibit a Blue Plan from requiring use of the appropriate CPT® Code modifiers for evaluation and management billing codes (e.g., CPT® modifiers 25 or 57) on their original claim forms, or (B) preclude a Blue Plan from requiring a Physician, Physician Group or Physician Organization to submit to an audit of claims submitted by such Physician, Physician Group or Physician Organization for payment directly to such Physician, Physician Group or Physician Organization (including, but not limited to, claims for surgical procedures and evaluation and management services on the same date of service submitted with the appropriate modifier), and to provide their Clinical Information in connection with such an audit.
- (d) A CPT® Code for supervision and interpretation or radiologic guidance (e.g., fluoroscopic, ultrasound or mammographic) shall be separately recognized and eligible for payment to the extent that

Page 55 of 58

the associated procedure code is recognized and eligible for payment; provided that, (i) the associated procedure code does not include supervision and interpretation or radiologic guidance according to the AMA CPT® book, and (ii) for each such procedure (e.g., review of x-ray or biopsy analysis or ultrasound guidance), no Blue Plan shall be required to pay for supervision or interpretation or radiologic guidance by more than one qualified health care professional.

- (e) With respect to indented codes, no Blue Plan shall reassign any CPT® Code into any other CPT® Code or deem a CPT® Code ineligible for payment based solely on the format of the published CPT® descriptions.
- (f) CPT® Codes submitted with a modifier 59 attached will be eligible for payment to the extent they follow the AMA CPT® book and they designate a distinct or independent procedure performed on the same day by the same Physician, but only to the extent that: (i) although such procedures or services are not normally reported together they are appropriately reported together under the particular presenting circumstances; and (ii) it would not be more appropriate to append any other CPT® recognized modifier to such CPT® Codes.
- (g) No global periods for surgical procedures shall be longer than the period then designated by CMS; provided that this limitation shall not restrict a Blue Plan from establishing a global period for surgical procedures (except where CMS has determined a global period is not appropriate or has identified a global period not associated with a specific number of days).
- (h) No Blue Plan shall automatically change a CPT® Code to one reflecting a reduced intensity of the service when such CPT® Code is one among or across a series that includes without limitation CPT® Codes that differentiate among simple, intermediate and complex, complete or limited, and/or size.
- (i) Not later than six (6) months after the Final Order Date, or as soon thereafter as is reasonably practicable, each Blue Plan shall update its claims editing software at least once each year to (A) cause its claim processing systems to recognize any new CPT® Codes or any reclassifications of existing CPT® Codes as modifier 51 exempt since the previous annual update, and (B) cause its claim processing personnel to recognize any additions to HCPCS Level II Codes promulgated by CMS since the prior annual update. As to both clauses (A) and (B) above, no Blue Plan shall be obligated to take any action prior to the effective date of the additions or reclassifications. Nothing in this subparagraph shall be interpreted to require a Blue Plan to recognize any such new or reclassified

Page 56 of 58

(j) Nothing contained in this § 7.20 shall be construed to limit a Blue Plan's recognition of CPT® modifiers to those CPT® modifiers specifically addressed in this § 7.20.

7.21 EOB and Remittance Advice Content

Case 1:03-cv-21296-FAM

Not later than six (6) months after the Final Order Date or as soon (a) thereafter as practicable, each Blue Plan's EOB forms for its Plan Members shall contain at least the following information: (i) the name of and a number identifying the Blue Plan's Plan Member; (ii) the date of service; (iii) the amount of payment for services provided; (iv) any adjustment to the invoice submitted; and (v) a generic explanation of any adjustment to the invoice submitted. Each such EOB form, or documents provided by the Blue Plan to its Plan Member along with each such EOB form, also shall specify an address and phone number for questions regarding the claim described on such EOB form. EOB contents must include the total amount originally billed by the Blue Plan's Participating Physician. Consistent with the desire that the Blue Plan's Plan Members receive accurate communications that do not disparage Non-Participating Physicians, each such EOB form shall indicate the amount, if any, for which the Physician may bill the Blue Plan's Plan Member and shall state the Physician's name, and a statement that the "Physician may bill you" such amount, if any, or contain substantially similar language, and shall not characterize disallowed amounts, if any, as unreasonable. In connection with a payment by a Blue Plan to a Plan Member for Covered Services rendered by a Non-Participating Physician, the EOB form or other Blue Plan document that accompanies the payment shall notify the Plan Member of his or her responsibility to apply such payment to the applicable claim received by the Plan Member from such Non-Participating Physician.

Not later than six (6) months after the Final Order Date or as soon thereafter as practicable, the Physician Remittance Advice or similar forms that each Blue Plan sends to its Participating Physicians communicating the results of claims adjudications shall contain at least: (i) the name of and a number identifying the Blue Plan's Plan Member; (ii) the date of service; (iii) the amount of payment per line item; (iv) the procedure code(s); (v) the amount of payment; (vi) any adjustment to the invoice submitted; (vii) a generic explanation of any adjustment of the invoice submitted that

Page 57 of 58

complies with HIPAA requirements; and (viii) any adjustment or change in any code on a line-by-line basis. Each such Physician Remittance Advice or similar form, or documents provided by the Blue Plan to its Participating Physician along with each such Physician Remittance Advice or similar form, also shall specify an address and phone number for questions by the Participating Physician regarding the claim described on such Physician Remittance Advice or similar form. This paragraph is not intended and shall not be construed to limit the Blue Plan's right to replace Physician Remittance Advice or similar forms with electronic Remittance Advices or the equivalent, to the extent such electronic Remittance Advices or the equivalent provide similar information so long as the Blue Plan complies with § 7.17(b).

This Agreement is not intended to alter or change rights of a Non-(b) Participating Physician to balance bill or bill a Blue Plan's Plan Member at rates and on terms that are agreed to between the Non-Participating Physician and that Plan Member.

Overpayment Recovery Procedures 7.22

As of the Final Order Date, each Blue Plan shall initiate or continue to take actions reasonably designed to reduce Overpayments. Such actions may include, without limitation, system enhancements to identify duplicate invoices prior to payment and construction and maintenance of one or more common Physician databases for use in connection with payment of Physician invoices. Each Blue Plan shall publish on its Public Website and its Provider Website an address and procedures for Physicians to return Overpayments. In addition, other than for recovery of duplicate payments or other similar adjustments including those relating to (i) claims where a Participating Physician has received payment for the same services from another payor whose obligation is primary, or (ii) timing or sequence of claims for the same Plan Member that are received by the Blue Plan out of chronological order in which the services were performed, each Blue Plan shall initiate Overpayment recovery efforts by providing Physicians with at least thirty (30) days written notice before engaging in additional Overpayment recovery efforts. Such notice shall include (i) the patient's name, (ii) the service date, (iii) the payment amount received by Physician, and (iv) a reasonably specific explanation of the proposed change (including, without limitation, procedure code where appropriate). No Blue Plan shall initiate Overpayment recovery efforts more than eighteen (18) months after the payment was received by Physician; provided, however, that no time limit shall apply to the initiation of Overpayment recovery efforts (a) based on a reasonable belief of fraud or other intentional misconduct, or (b) required by a Self-Insured Plan, or (c) required by a state or federal government program. Notwithstanding the above, in the event that a Physician asserts a claim of underpayment, a Blue Plan may defend or set off such claim based on

Page 58 of 58

Overpayments going back in time as far as the claimed underpayment. If a Physician requests an appeal within thirty (30) days of receipt of a request for repayment of an Overpayment, no Blue Plan shall require such Physician to repay the alleged Overpayment before such appeal is concluded. Nothing in this Agreement, including but not limited to the provisions of § 13, shall be deemed to limit a Blue Plan's right to pursue recovery of Overpayments that occurred prior to the Final Order Date where the Blue Plan has provided the Physician with notice of such recovery efforts prior to the Final Order Date.

7.23 Efforts to Improve Accuracy of Information about Eligibility of Plan Members

Commencing on the Final Order Date, each Blue Plan shall initiate or continue to take actions reasonably designed to reduce Overpayments and claim denials resulting from inaccurate information about the eligibility of its Plan Members. The Blue Plan may reduce, discontinue, or expand such activities commensurate with their demonstrated effectiveness.

7.24 Intentionally Left Blank

7.25 Effect of Blue Plan Confirmation of Patient Procedure/Medical Necessity

If the Blue Plan certifies or Precertifies (or approves or pre-approves) that a proposed service is medically necessary for one of its Plan Members, the Blue Plan shall not subsequently revoke that medical necessity determination absent evidence of fraud, evidence that the information submitted was materially erroneous or incomplete, or evidence of material change in that Plan Member's health condition between the date that the certification or Precertification was provided and the date of the service that makes the proposed service no longer medically necessary for such Plan Member. In the event that a Blue Plan certifies or Precertifies the medical necessity of a course of treatment limited by number, time period or otherwise, then a request for services beyond the certified course of treatment shall be deemed to be a new request and that Blue Plan's denial of such request shall not be deemed to be inconsistent with the preceding sentence.

7.26 Intentionally Left Blank

7.27 Information about Physicians Provided by Blue Parties

Information currently posted on each Blue Plan's Public Website about individual Physicians or contained in printed materials prepared by the Blue Plan is derived from data supplied by those Physicians and from applicable agreements between the Blue Plan and its Participating Physicians or their Participating Physician Groups or Participating

Page 1 of 45

Physician Organizations. Upon written notice of an inaccuracy sent to a Blue Plan (pursuant to the direction as to how to give such notice that will be posted on the Blue Plan's Provider Website), if the Blue Plan does not dispute that there is an inaccuracy the Blue Plan shall take steps reasonably necessary to ensure that the Public Website is updated within twenty (20) business days after receipt of such notice and that written materials are revised before the next edition of such materials is printed (to the extent there is sufficient time to make such revisions before the next printing) to reflect any corrections in the Physician information to make it accurate. Upon written notice that a Physician is incorrectly listed as a Participating Physician on a Blue Plan's Public Website or in printed materials prepared by a Blue Plan (pursuant to the direction as to how to give such notice that will be posted on the Blue Plan's Provider Website), if the Blue Plan does not dispute that there is an inaccuracy the Blue Plan shall take steps reasonably necessary to delete any such erroneous reference from the Public Website within twenty (20) business days after receipt of such notice and from any written materials before the next edition of such materials is printed (to the extent there is sufficient time to make such revisions before the next printing), and the Blue Plan shall make corresponding changes in systems affecting the level of payments and generation of EOBs within twenty (20) business days after receipt of such notice. If the Blue Plan disputes that there is an inaccuracy, it will so notify the Physician within the same time periods specified above, including the basis on which it disputes that there is an inaccuracy.

7.28 Intentionally Left Blank

7.29 Miscellaneous

Gag Clauses (a)

No Blue Plan shall include in its contracts with Participating Physicians, Participating Physician Groups, or Participating Physician Organizations any provision limiting the free, open and unrestricted exchange of information between its Physicians and its Plan Members regarding the nature of the Plan Member's medical conditions or treatment and provider options and the relative risks and benefits and costs to the Plan Member of such options, whether or not such treatment is covered under the Plan Member's Plan, and any right to appeal any adverse decision by the Blue Plan regarding coverage of treatment that has been recommended or rendered. Each Blue Plan agrees not to penalize or sanction Participating Physicians in any way for engaging in any free, open and unrestricted communication with a Plan Member with respect to the foregoing subjects or for advocating for any service on behalf of a Plan Member.

Intentionally Left Blank (b)

Page 2 of 45

Arbitration (c)

Document 928-3

- With respect to any arbitration proceeding between a Blue (i) Plan and its Participating Physician who practices individually or in a Participating Physician Group of less than six Physicians, the Blue Plan agrees that it shall refund any applicable filing fees and arbitrators' fees paid by such Physician in the event the Physician is the prevailing party with respect to such arbitration proceeding; provided, however, that this paragraph shall not apply with respect to any arbitration proceeding in which the Participating Physician purports to represent any Physician outside of his or her Physician Group.
- Each Blue Plan agrees not to include language in any (ii) agreement with a Physician, Physician Group, or Physician Organization (A) requiring that any arbitration panel have multiple members, (B) preventing the recovery of any statutory or otherwise legally available damages or other relief in an arbitration proceeding, (C) restricting the statutory or otherwise legally available scope or standard of review, (D) completely prohibiting discovery, (E) shortening any statute of limitations, or (F) requiring that any arbitration proceeding occur more than fifty (50) miles from the principal office of the Physician, Physician Group, or Physician Organization.

_(d) Impact of this Agreement on Standard Form Agreements and Individually Negotiated Contracts.

- (i) Each Blue Plan's future standard form agreements with Participating Physicians shall not be inconsistent with the commitments and undertakings the Blue Plan makes in this Agreement. To the extent that a Blue Plan's existing standard form agreements with Participating Physicians contain provisions inconsistent with the terms hereof, the Blue Plan shall administer such agreements consistent with the terms set forth in this Agreement.
- Except for those terms relating to higher or customized (ii) rates, length of term of the contract, and/or other customized payment methodologies or as otherwise permitted under §§ 7.13(b), 7.13(c), 7.14(a) and 7.29(r), this Agreement shall be deemed to modify or nullify any inconsistent terms of an Individually Negotiated Contract.

Case 1:03-cv-21296-FAM

With respect to Individually Negotiated Contracts executed (iii) after the Preliminary Approval Date, a Blue Plan may agree with individual Participating Physicians, Participating Physician Groups or Participating Physician Organizations on terms that deviate from the terms of this Agreement relating to higher or customized rates, length of term of the contract, and/or other customized payment methodologies or as otherwise permitted under §§ 7.13(b), 7.13(c), 7.14(a) and 7.29(r). In addition, a Blue Plan may agree with individual Participating Physicians, Participating Physician Groups or Participating Physician Organizations on terms that deviate from any other terms of this Agreement upon request of such individual Participating Physicians, Participating Physician Groups or Participating Physician Organizations.

Impact of this Agreement on Covered Services (e)

Nothing contained in this Agreement shall: (i) require a Blue Plan to pay a Participating Physician, Participating Physician Group or Participating Physician Organization at any particular amount, or (ii) prohibit a Blue Plan from utilizing a particular reimbursement methodology. Said payments or reimbursement methodologies shall be implemented consistent with the applicable terms of this Agreement. Notwithstanding anything to the contrary contained in this Agreement, nothing contained in this Agreement shall supersede or otherwise alter the scope of Covered Services of any Plan or require a Blue Plan or any Plan to pay for services that are not Covered Services. In determining whether services provided to one of its Plan Members are Covered Services under a Self-Insured Plan, each Blue Plan shall apply the definition of "Medically Necessary" (or any comparable term) contained in § 7.16(a) except with respect to the limited number of large Self-Insured Plans that require a different definition of "Medically Necessary" (or any comparable term) be applied. With respect to such Self-Insured Plans, each Blue Plan shall recommend that the definition of "Medically Necessary" (or any comparable term) contained in § 7.16(a) apply.

- **(f)** Intentionally Left Blank
- Intentionally Left Blank (g)
- Intentionally Left Blank (h)
- (i) **Pharmacy Provisions**

Document 928-

When a Blue Plan provides pharmacy coverage, the Blue Plan shall make formulary information available to its Plan Members. Each Blue Plan shall maintain the process, as reasonably amended, for covering medications not included in the formulary when Medically Necessary that is in place on the Execution Date. When a Blue Plan provides pharmacy coverage, each Blue Plan will continue to provide coverage for off-label uses of pharmaceuticals that have been approved by the FDA (but not approved for the prescribed use) provided that the drug is not contraindicated by the FDA for the off-label use prescribed, and that the drug has been proven safe, effective and accepted for the treatment of the specific medical condition for which the drug has been prescribed, as evidenced by the results of good quality controlled clinical studies published in at least two or more peer reviewed full length articles in respected national professional medical journals. In reaching a conclusion whether to provide coverage for a drug under the preceding sentence, the Blue Plan will consider supporting documentation in either the American Hospital Formulary Service Drug Information or the United States Pharmacopeia Drug Information. Blue Plans shall retain the right to Precertify coverage of specific medications for non-approved use. A Blue Plan's disclosure concerning Precertification and potential restrictions on non-approved use of prescription medications shall be similar in substance to disclosure concerning formularies, as described above.

(j) Restrictive Endorsements

Where a Blue Plan's reimbursement of a Physician for services performed by that Physician is a partial payment of allowable charges, a Physician may negotiate a check with a "Payment in Full" or other restrictive endorsement without waiving the right to pursue a remedy available under this Agreement.

(k) Scope of Blue Parties' Responsibilities

(i) The obligations undertaken by each Blue Party under § 7 of this Agreement shall be applicable only to those functions or activities performed directly by the Blue Party, its employees, and third parties (other than Delegated Entities) performing functions or activities on the Blue Party's behalf. Each Blue Plan shall make a good faith effort to include in contracts entered into with Delegated Entities subsequent to the Final Order Date terms that are substantially equivalent to the terms of this Agreement; provided that no Blue Party shall be liable under this Agreement in the event any Delegated Entity acts in a manner inconsistent with the terms of this Agreement.

- (ii) The provisions of § 7 of this Agreement shall apply to a Blue Plan's activities in connection with the BlueCard Program or any similar national account delivery program governed by the BCBSA (including but not limited to NASCO-to-NASCO arrangements) only to the extent that the Blue Plan is solely responsible for the activity, with respect to a claim under such a program, that is subject to § 7 of this Agreement.
- The Blue Parties recognize the desirability of improving (iii) Blue Plans' business relationships with Physicians who provide Covered Services to Blue Plans' Plan Members in connection with the BlueCard Program and other similar national account delivery programs governed by BCBSA (including but not limited to NASCO-to-NASCO arrangements) (the "Programs"). To that end, the Blue Parties have made investments and undertaken initiatives reasonably designed to streamline and improve the overall efficiency of the claim adjudication process for the Programs. These investments and initiatives seek to improve Physician satisfaction with the Programs by, among other things, working towards the goal of processing and paying claims under these Programs within the times set forth in § 7.18(a). The Blue Parties will continue such initiatives and improvements during the Effective Period. Since 2000 and through the Effective Period, BCBSA's investments in furtherance of these initiatives will exceed \$150,000,000. These investments are in addition to the investments referred to in § 7.31. Examples of those initiatives include, but are not limited to, those set forth in subparagraphs (A) through (C) below.
 - technology A multi-year information initiative aiming to improve the timeliness of claims implementing and payment by processing enhancements to transaction processing and renovating and upgrading the technologies used by the Blue Parties to improve their information web-enabled services. management and Improvements include the continued upgrading of BlueExchange® and the development and implementation of Blue²®. BlueExchange is an electronic, inter-Blue Plan system supporting Physicians' ability to make and receive responses to eligibility benefit inquiries, claim status requests, and referral authorizations and remittance advice. Blue²® is a web-based application initiated to facilitate real-time transaction transmission,

inventory management related to inter-Blue Plan claims, and inter-Blue Plan claims exception management, including the capability to enable electronic exchanges of claims attachments such as clinical records among Blue Plans.

- Measures to better facilitate resolution of (B) Physician issues related to inter-Blue Plan claims through standardized claims inquiry escalation among Blue Plans, information processes technology and management systems to develop, improve and accelerate dispute resolution, such as with respect to Blue Plans' activities pertaining to claims payment and billing issues, requests for "Medical Necessity" Clinical Information, determinations, and related disclosures, and the promotion of "best practices".
- (C) Measures to promote overall Physician satisfaction with the handling of inter-Blue Plan claims, including enhanced performance measurements and monitoring capabilities.

The Blue Parties may modify any specific initiatives described in subsections (A)-(C) of this § 7.29(k)(iii), or institute new initiatives to accomplish the same objectives.

requirements, rules, or regulations, or modifications thereof, that would cause any Blue Plan to be in violation of any of the requirements set forth in this Agreement or would cause any representation made herein to become untrue. Each of the Blue Plans agrees that it will not encourage or support any change in BCBSA requirements, rules or regulations that would conflict with the Blue Plan's obligations under this Agreement.

(I) Copies of Contracts

Each Blue Plan shall provide a copy of its contract with a particular Participating Physician (including without limitation a contract with a Physician Organization or a Physician Group in which such Participating Physician participates) to such Participating Physician, upon receipt by the Blue Plan of a written request by such Participating Physician to provide such copy, except in circumstances where the Blue Plan is restricted from providing a Participating Physician with a copy of the Blue Plan's contract with a Physician Organization or Physician Group

specifically because of terms contained in that contract. No Blue Plan shall require that a restriction as described in the previous sentence be included in its contracts with Physician Organizations or Physician Groups.

State and Federal Laws and Regulations (m)

Nothing contained in § 7 of this Agreement is intended to or shall, in any way, reduce, eliminate, or supersede any Party's obligation to comply with applicable provisions of relevant state and federal law and regulations. To the extent state or federal law or regulation imposes, with respect to a specific obligation created by § 7, a greater obligation than that specifically set forth in § 7, the Blue Parties shall comply with said law or regulation. The Compliance Dispute resolution procedures contained in § 12 shall apply with respect to any alleged breach of an obligation created by the preceding sentence. Nothing in this § 7.29(m) is intended to give rise to or should be construed as giving rise to any private right of action for any violation of any federal or state law or regulation (whether under a breach of contract theory, including a claim of breach of this Agreement, or any other theory) where federal or state law or regulation does not allow a Physician a right of action for such violation. The Compliance Dispute Officer shall not take any action inconsistent with any ruling, determination or directive by any court or regulatory agency. Any action taken by the Compliance Dispute Officer that is inconsistent with any subsequent ruling, determination or directive by any court or regulatory agency shall not be binding on a Blue Party as of the effective date of such subsequent ruling, determination or directive. The Compliance Dispute Officer shall not award liquidated, punitive or multiples of damages that might be available under state or federal law or regulations. Further, a Physician shall be required to elect whether to pursue a remedy under this Agreement through the process set forth in § 12, or to pursue any remedy available under such state or federal law or applicable regulations.

Intentionally Left Blank (n)

Limitations on Obligations of Non-Participating Physicians **(0)**

No affirmative obligation that § 7 imposes on Physicians shall apply to any Non-Participating Physician unless and until, and then only to the extent that, such Non-Participating Physician pursues with a Blue Plan a claim for payment on the Non-Participating Physician's own behalf or such Non-Participating Physician pursues benefits under this Agreement, in which case any affirmative obligations that § 7 imposes on Physicians shall apply

Page 8 of 45

to such Non-Participating Physician with respect to such claim or such benefits.

(p) Intentionally Left Blank

(q) Effect of Assignment of Benefits

The existence, submission, and/or acceptance of an assignment of benefits authorization in favor of a Non-Participating Physician shall not preclude the Non-Participating Physician from collecting from a Blue Plan's Plan Member the difference between the Non-Participating Physician's full fee and the payment (if any) received by the Non-Participating Physician from the Blue Plan. Notwithstanding anything to the contrary contained in this Agreement, nothing contained in this Agreement shall otherwise require the Blue Plan to make payment directly to any Non-Participating Physician.

(r) Most Favored Nations Clauses.

The Blue Plans will not include any "most favored nations" clauses in its contracts with Participating Physicians, Participating Physician Groups and Participating Physician Organizations, except for Individually Negotiated Contracts.

7.30 Compliance with Applicable Law and Requirements of Government Contracts

The obligations undertaken in § 7 herein shall be fulfilled by each (a) Blue Plan to the extent permissible under applicable laws and regulations, the terms and conditions of current and future government contracts, and applicable governmental directives. If, and during such time as, a Blue Plan is unable to fulfill an obligation under § 7 to the extent contemplated by this Agreement because to do so would require governmental approval or action, the Blue Plan shall perform such obligation to the extent permissible under applicable laws and regulations, the terms and conditions of current and future government contracts, and applicable governmental directives, and the Blue Plan shall continue to fulfill its other obligations under § 7 to the extent permitted under applicable laws and regulations, the terms and conditions of current and future government contracts, and applicable governmental directives. To the extent that any governmental approval is required for any Party to fulfill an obligation under § 7, such Party shall make reasonable efforts to obtain any necessary approvals from the appropriate governmental entities. For any obligation under § 7 that cannot be undertaken without governmental approval, the Effective Period as to that

- obligation shall be delayed until such approval is granted. Nothing in this § 7 shall apply to a Blue Party's role as a carrier providing administrative services for Medicare Part B or to FEHBA.
- Each Blue Party that contracts directly with the U.S. Department of (b) Defense under the TriCare Program will present the requirements of § 7 of this Agreement to the Tricare Management Activity ("TriCare") for consideration of incorporating such requirements into such Blue Party's TriCare Contract pursuant to the changes clause of such contract. Unless and until such time that a Blue Party and TriCare so amend the Contract, any Blue Party's TriCare Program is exempted from the requirements of this Agreement. The TriCare Program is regulated by the United States Department of Defense. The Department of Defense provides rules, regulations and contract provisions applicable to that Program, including certain mechanisms to enforce those rules, regulations and contract. Where a Blue Party's TriCare Contract has been amended as provided in this subsection, such Blue Party will comply with the rules and regulations related to such amendments, including the mechanisms to enforce those rules and regulations.

7.31 Estimated Value of § 7 Initiatives

Since the inception of this Action and through the Termination Date, the Blue Parties will have spent over \$250,000,000 in order to implement and carry out the initiatives described in § 7 of this Agreement and otherwise to improve relations with Physicians. Such initiatives have included automated adjudication of claims, increased internet and clearinghouse functionality, and functionality to improve responses to Physician inquiries. Taking into account these expenditures, the Parties estimate that the approximate aggregate value of the initiatives and other commitments regarding the Blue Parties' business practices set forth in § 7 of this Agreement is in excess of the amount stated above.

7.32 Force Majeure

No Party hereto shall be required to meet an obligation under this Agreement where the inability to meet such obligation is the result of any act of God, governmental act, act of terrorism, war, fire, flood, earthquake or other natural disaster, explosion or civil commotion ("Force Majeure"). The performance of a Party's obligations under this Agreement, to the extent affected by the delay, shall be suspended for the period during which the cause, or the Party's substantial inability to perform arising from the cause, persists. If the performance of any obligation under this Agreement is excused or delayed by Force Majeure and that obligation is a condition precedent for the performance of an obligation by another Party, performance of the obligation by the second

Party shall be excused or delayed to the same extent as the performance of the obligation by the first Party.

Managed Care Issues Relating to Mental Health and Substance Abuse 7.33

- Except where any applicable law or regulation requires a different (a) definition, each Blue Plan shall apply to its current agreements and include in its future agreements with Participating Physicians the definition of Medical Necessity in § 7.16(a) with respect to mental health services, including treatment for psychiatric illness and substance abuse, subject to the terms and conditions of the Plan of the Blue Plan's Plan Member; provided that in determining the clinical appropriateness of care, the following minimum standards relevant to mental health care must be met:
 - There is a diagnosis as defined by standard diagnostic (i) nomenclatures (DSM IV or its equivalent in ICD-9-CM, or replacement thereof) and an individualized treatment plan appropriate for the Plan Member's illness or condition; and
 - There is a reasonable expectation that the Plan Member's (ii) illness, condition, or level of functioning will be stabilized, improved, or maintained through ambulatory care, through treatment known to be effective for the Plan Member's illness; and
 - The mental health services are not primarily for the (iii) avoidance of incarceration of the Plan Member.
- Each Blue Plan agrees that its Participating Psychiatrists will be (b) listed in the Blue Plan's provider directory, via the hard-copy directory, via the Blue Plan's Public Website, via a "hot link" on the Blue Plan's Public Website, or otherwise. If a primary care Physician referral is required, each Blue Plan will allow its primary care Participating Physicians to make referrals to the Blue Plan's Participating Psychiatrists, provided that any such referral is subject to the same referral requirements that apply to referrals to other Participating Physicians. Nothing in the preceding sentence shall be construed to remove or change any applicable Plan Member or Physician Precertification requirements.
- Each Blue Plan agrees that, where a Psychiatrist has not entered (c) into a different agreement with the Blue Plan or the hospital or other mental health care facility where the services are rendered, and where the Blue Plan has not entered into a different agreement with such hospital or mental health care facility, the Blue Plan will separately consider and pay for Medically Necessary Covered Services provided to one of the Blue Plan's Plan Members by the

Psychiatrist, in accordance with the terms and conditions of the Plan Member's Plan.

- Where applicable, each Blue Plan adheres to state "prudent (d) layperson" laws which require payment of benefits for mental health services in the event of an emergency under prudent layperson standards. An emergency department Physician can make a decision regarding admission or physical or chemical restraints. Each Blue Plan agrees that, where a Physician has not entered into a different agreement with the Blue Plan or the hospital or other mental health care facility where the services are rendered, and where the Blue Plan has not entered into a different agreement with such hospital or mental health care facility, in the event of an emergency, the Blue Plan will pay for Medically Necessary emergency care mental health Covered Services provided by such Physician in accordance with applicable prudent layperson standards, the definition of Medical Necessity in § 7.16(a), and the terms and conditions of the Plan Member's Plan, and the Blue Plan will pay for Medically Necessary mental health Covered Services provided by Physicians resulting from the admission in accordance with the definition of Medical Necessity in § 7.16(a) and the terms and conditions of the Plan Member's Plan.
- (e) Each Blue Plan will post on its Provider Website an authorization form that Physicians providing mental health services to the Blue Plan's Plan Members may print or download to obtain Plan Member consent for release of Clinical Information to the Blue Plan.

7.34 Intentionally Left Blank

7.35 Use of Third Party Intellectual Property

If and to the extent that any activity of a Blue Party in compliance with one or more provisions of this Agreement has subjected or, in a Blue Party's reasonable judgment, may subject it to an Intellectual Property Claim, including but not limited to a claim by the AMA, the Blue Party's obligation to comply with that provision or those provisions is suspended. Within ten (10) days of determining that compliance with one or more provisions may subject it to an Intellectual Property Claim, the Blue Party will notify the Class Compliance Dispute Facilitator and Class Counsel and offer to meet and confer to resolve the issue including developing one or more alternative business practices that achieve comparable results to the affected provisions at no significant increase in cost to the Blue Plan. If the parties do not resolve the issue, and Class Counsel and the Class Compliance Dispute Facilitator believe there is an alternative business practice that achieves comparable results, without a significant increase in

cost and without creating an Intellectual Property Claim, Class Counsel and the Class Compliance Dispute Facilitator may bring a Compliance Dispute under § 12.

Competitive Flexibility 7.36

"Negotiated Purchaser" (a)

- A "Negotiated Purchaser" is an entity that meets all of the (i) following requirements: (x) the entity is an employer (including any governmental unit acting in the capacity of an employer, other than the federal government under the FEHBP), union, trust, association or other entity that sponsors a Self-Insured Plan or Fully-Insured Plan covering a group of active and/or retired employees; (y) the entity has negotiated, intends to negotiate (or otherwise has issued a request for proposals), or is negotiating a service agreement or insurance contract with the Blue Plan; and (z) as a result of the entity's size, covered lives, or strategic significance, changes made by the Blue Plan to its standard operating policies and procedures are both required by such entity and reasonably necessary in order to obtain or maintain an agreement with such entity. There will be a limited number of such entities for each Blue Plan during the Effective Period of this Agreement.
- Each Blue Plan will make reasonable good faith efforts to (ii) satisfy Negotiated Purchasers without any terms or conditions that differ from those contained in § 7. A Blue Plan's agreement with a Negotiated Purchaser may have terms or conditions that differ from those contained in § 7 if reasonably necessary in order for the Blue Plan to obtain or maintain a service agreement or insurance contract with the entity. All other terms and conditions of this Agreement shall remain in full force and effect. The Blue Plan will promptly notify the Class Compliance Dispute Facilitator of any agreement containing any such term or condition and promptly thereafter post on its Provider Website any material differences from the provisions of § 7 that relate to the terms and conditions of payment to Participating Physicians, Participating Physician Groups or Participating Physician Organizations. The Class Compliance Dispute Facilitator, sua sponte or upon receiving a complaint from a Class Member, may challenge any such term or condition through the Compliance Dispute process described in § 12.10. Such challenge shall be brought within five (5) business days following the date on which the Blue Plan provides notice to the Class

Compliance Dispute Facilitator of any agreement containing any such term or condition. If the Class Compliance Dispute Facilitator fails within the period set forth herein to challenge the proposed term or condition of any such agreement, such failure shall be deemed a waiver to challenge any such term or condition in such agreement. A final decision of such a challenge shall preclude further challenges to the same term or condition.

(b) Ability of Blue Parties to Substitute Business Practices

- (i) Each Blue Party may alter the method or means by which it makes any disclosure or otherwise transmits information as described in, and required by, this Agreement, so long as the Blue Party reasonably believes, expects and intends that the newly-adopted means or method of disclosure or transmission is as effective as or more effective than the means or method set forth in this Agreement.
- (ii) Each Blue Party may substitute for any of the business practices required of it by § 7 of this Agreement an alternative business practice so long as the alternative business practice is reasonably designed to achieve comparable results consistent with this Agreement. Prior to implementing any such alternative business practice, a Blue Party shall give notice to the Class Compliance Dispute Facilitator of such alternative business practice, and meet and confer with the Class Compliance Dispute Facilitator, if requested, within fifteen (15) business days following the date of the notice with respect thereto. If the matter is not resolved, and the Blue Party proceeds with implementation, then the Class Compliance Dispute Facilitator may bring a compliance dispute with respect to the alternative business practice within ten (10) business days following notice from the Blue Party of its implementation of such alternative business practice. Alternatively, the Blue Party may defer implementation and seek a review of the matter by the Compliance Dispute Officer. The Class Compliance Dispute Facilitator may confer with and seek the assistance of Class Counsel in connection with any such dispute. The Blue Party shall cease implementing any alternative business practice, to the extent it has done so, within sixty (60) days following the written notice to the Blue Party from the Compliance Dispute Officer that such alternative business practice is not reasonably designed to achieve comparable results consistent with this Agreement. If the Class Compliance Dispute Facilitator fails within the period set forth herein to request to meet and confer with

the Blue Party, such failure shall be deemed a waiver to challenge any such alternative business practice in the future. The Compliance Dispute Officer shall hold in confidence all information supplied by the Blue Party regarding such alternative business practice.

Modification for Specific Circumstances (c)

Notwithstanding the provisions of § 7.36(a), a Blue Plan may modify any undertaking if either (i) the modification is reasonably necessary to compete in a particular geographic marketplace, or (ii) performance of one or more provisions of this Agreement would be impractical. For this purpose, performance would be "impractical" if it would place the Blue Plan at a demonstrable operational disadvantage, would be unduly burdensome, or would, on account of new technology, be inefficient or less cost-effective relative to use of the new technology. Each Blue Plan shall have an obligation to use good faith efforts to address any issue that arises under this subsection by taking action consistent with the terms of § 7 if it is reasonably practical to do so. In the event a Blue Plan intends to implement a modification under this § 7.36(c), the Blue Plan shall give notice to Class Compliance Dispute Facilitator of the terms of such modification, which notice shall be in advance of implementing the modification unless competitive circumstances prevent advance notice in which case the notice shall be as soon as practicable. Class Compliance Dispute Facilitator shall have fourteen (14) days in which to provide written notice to the Blue Plan of any objections to such modification. Failure to provide such notice shall constitute a waiver of the right to object or assert a Compliance Dispute with regard to such modification, as otherwise permitted under subsection (c) of this § 7.36. Within ten (10) business days after the Blue Plan's receipt of any notice of objections, the Blue Plan, if requested in the notice of objections, shall meet and confer with Class Compliance Dispute Facilitator with respect to the objections. If the Blue Plan and Class Compliance Dispute Facilitator are unable to resolve the matter, the Blue Plan shall promptly notify the Class Compliance Dispute Facilitator whether it intends to implement or continue to implement the modification. If the Blue Plan gives notice that it intends to implement or continue to implement the modification, then within ten (10) business days of the notice, the Class Compliance Dispute Facilitator may initiate a Compliance Dispute. If the Compliance Dispute Officer rules in favor of the Blue Plan, such final decision of such a challenge shall preclude further challenges to the

modification. If the Compliance Dispute Officer rules against the Blue Plan, any remedy shall be limited to directing the Blue Plan to submit and implement an adequate corrective action plan, and imposing on the Blue Plan the obligation to pay actual damages including interest suffered by a Class Member as a result of the implementation of any modification. Any activity conducted by a Blue Plan under a corrective action plan shall be subject to the provisions of § 13.3(a)(iii). The Class Compliance Dispute Facilitator may confer with and seek the assistance of Class Counsel in connection with any such dispute.

8. Other Settlement Consideration

In addition to the business initiatives set forth in § 7 of this Agreement, the settlement consideration shall include a settlement fund (the "Settlement Fund") for payment of claims to Class Members, which will be established and operated in accordance with the provisions of §§ 8.2 through 8.4.

8.1 Intentionally omitted

8.2 Settlement Fund

- Each Blue Plan shall pay the amount specified for it in Exhibit K (a) to an account (the "Settlement Administration Account") (each such payment constituting a "Settlement Fund Payment"), as hereinafter provided in this § 8.2 and § 8.5. Subject to the provisions of § 8.2(e), the sum of all Settlement Fund Payments shall equal \$128,315,907 (the "Settlement Fund Amount") plus interest as provided in § 8.5. Each Settlement Fund Payment shall be treated as a payment to a Qualified or Designated Settlement Fund under I.R.C. § 468B and the regulations or proposed regulations promulgated thereunder (including without limitation Treasury Reg. § 1.468B-1-5 or any successor regulation).
- By no later than ten (10) days following the Preliminary Approval (b) Date, the Blue Parties shall cause to be established the Settlement Administration Account. The Blue Parties shall select and retain a settlement administrator (the "Settlement Administrator"). Under the joint supervision of the Blue Parties and Class Counsel or their designees, and subject to the supervision, direction and approval of the Court, the Settlement Administrator shall be responsible for the administration of the Settlement Administration Account.
- Each Blue Plan shall pay its Settlement Fund Payment by wire (c) transfer of immediately available funds, not later than ten (10) days after the Effective Date
- (d) The Settlement Fund shall consist of the total of all Settlement Fund Payments and all interest, dividends, and other income

Page 16 of 45

- earned on funds in the Settlement Administration Account, less any reserve for taxes.
- (e) The settlement consideration identified in §§ 7.31, 8.2(a), and 9.1 of this Agreement is premised on the participation in this Agreement by all Blue Plans. If any Blue Plan terminates its participation in this Agreement pursuant to § 14.2, (i) the Settlement Fund Amount defined in this § 8.2 shall be reduced by the amount specified in Exhibit K for the Terminating Blue Plan, as defined in § 14.2, and (ii) the Estimated Value defined in § 7.31, and the Unopposed Amount, defined in § 9.1, shall be reduced by the percentage specified in Exhibit K for the Terminating Blue Plan.

8.3 Responsibilities of the Settlement Administrator

In addition to the responsibilities set forth in § 8.2(b), the responsibilities of the Settlement Administrator shall expressly include without limitation: (i) the determination of the eligibility of any Class Member to receive payment from the Settlement Fund and the amount of payment to be made to each Class Member from the Settlement Administration Account, in accordance with the provisions of § 8.3 of this Agreement; (ii) the determination as to whether the election of any Class Member to contribute a settlement payment has been authorized by such Class Member, in accordance with the provisions of § 8.3(g) of this Agreement; (iii) the administration of an appropriate procedure for the adjudication of disputes that may arise with respect to the eligibility of a Class Member to receive a payment from the Settlement Fund or the amount of the payment authorized to be made to any Class Member under the provisions of this Agreement; (iv) the filing of any tax returns necessary to report any income earned by the Settlement Fund and the payment from the Settlement Administration Account, as and when legally required, of any tax payments (including interest and penalties) due on income earned by the Settlement Fund and to request refunds, when and if appropriate, with any such tax refunds that are issued to become part of the Settlement Fund; (v) the compliance with any other applicable law; and (vii) any other duties the Blue Parties and Class Counsel agree in writing to assign to the Settlement Administrator. The fees and expenses of the Settlement Administrator shall be paid by the Blue Parties; provided that neither the Blue Parties nor Class Counsel shall be responsible for any other costs, expenses or liabilities of the Settlement Fund.

(a) The portion of the Settlement Fund that will be available in the aggregate to satisfy claims by Retired Physicians (the "Retired Physician Amount") shall be equal to the Settlement Fund multiplied by two times the quotient derived by dividing the number of Retired Physicians who file valid Claim Forms within ninety (90) days of the Notice Date by the total number of Class Members, as determined by the Settlement Administrator. Each

Page 17 of 45

Retired Physician who files a valid Claim Form shall be entitled to receive a payment equal to the Retired Physician Amount divided by the total number of Retired Physician valid Claim Forms. A Retired Physician who retired after January 1, 2003 shall have the option of filing either as a Retired Physician or as an Active Physician pursuant to the provisions of §8.3(c). If the claim of a Physician is unclear as to whether it is brought by a Retired Physician or as an Active Physician, the Settlement Administrator will treat the claim in the manner that will result in the largest payment to the Physician.

- (b) The amount remaining in the Settlement Fund after subtracting the Retired Physician Amount will be available to satisfy claims by Class Members other than Retired Physicians (the "Active Physician Amount").
- (c) Each Active Physician who files a valid Claim Form within ninety (90) days of the Notice Date shall be entitled to receive payment from the Settlement Fund in an amount to be determined according to whether the Active Physician's gross receipts for providing Covered Services to Plan Members during the three calendar year period of 2004, 2005 and 2006 were (x) less than \$5,000, (y) at least \$5,000 but less than \$50,000, or (z) \$50,000 or greater. For purposes of this determination, gross receipts include amounts paid by Blue Plans or by Delegated Entities for providing Covered Services to Plan Members. The Settlement Administrator shall determine the category for each Active Physician based upon the certification in the Claim Form and/or such independent verification, if any, that the Settlement Administrator may undertake in its sole discretion.
- The Settlement Administrator shall determine the total number of (d) Active Physicians who fall within each of the three categories set forth in § 8.3(c) and determine the total number of distribution shares (each a "Base Amount") necessary to make distributions according to the following formula: The Active Physician Amount shall be allocated among all Active Physicians who file valid Claim Forms such that each Active Physician who falls within § 8.3(c)(x) shall be entitled to receive a single Base Amount; each Active Physician whose Claim Form establishes that he or she falls within § 8.3(c)(y) shall be entitled to receive five times the Base Amount; and each such Active Physician whose Claim Form establishes that he or she falls within § 8.3(c)(z) shall be entitled to receive ten times the Base Amount. A Class Member who files an otherwise valid Claim Form but does not specify a category of gross receipts for Covered Services to Plan Members, shall be deemed to be entitled to a single Base Amount, and the Settlement

Page 18 of 45

- Administrator has no obligation to pursue additional information about the Class Member's status or amount of receipts.
- (e) The Settlement Administrator shall establish procedures to permit an Active Physician to establish, through the submission of billing records or similar information, that he or she should fall into a category entitled to a higher payment from the Settlement Fund based on aggregate payments received for providing Covered Services to Plan Members over any other consecutive three-year period from January 1, 1997 through December 31, 2006.
- (f) If a Class Member submits a Claim Form requesting compensation under the wrong compensation category (e.g., a request under the Retired Physician Amount which should have been submitted as a request under the Active Physician Amount), the Settlement Administrator may at its sole discretion review the Claim Form under the provisions set forth herein for the correct settlement category unless the documentation submitted with said Claim Form is insufficient under those provisions.
- (g) Each Class Member who files a valid Claim Form may elect either to receive the payment from the Settlement Fund or to direct that such amount be contributed on his, her or its behalf to the Physicians' Foundation for Health Systems Innovations, Inc. ("Foundation") or to a charitable organization set forth on Exhibit L. The Foundation and the charitable organizations set forth on Exhibit L shall be identified on a list attached to the Claim Form.
- An eligible Class Member must submit a timely claim form (the (h) "Claim Form") to the Settlement Administrator using the Claim Form attached as Exhibit A hereto and in accordance with the Claim Form Instructions attached as Exhibit B and included in the Notice and in the Claim Form in order for such Class Member to have a valid right to receive payment from the Settlement Fund. Promptly after receipt of all timely Claim Forms, the Settlement Administrator shall calculate the amount that is payable to, or on behalf of, each Class Member pursuant to the provisions of § 8.3. Reasonably promptly upon completion by the Settlement Administrator of the calculations of the amounts that are payable and after all Settlement Fund Payments have been made, reserved for tax liabilities or as otherwise agreed by the Parties, the Settlement Administrator shall cause the Settlement Administration Account to disburse payment to Class Members in each category who or which submitted valid Claim Forms in accordance with § 8.3 or to the Foundation or to a charitable organization set forth on Exhibit L as directed by such Class Members. Any Class Member submitting a Claim Form shall, through the act of submitting that Claim Form, be subject to the jurisdiction of the Court for any related proceedings. Physician

Page 19 of 45

Groups and Physician Organizations shall be allowed to file Claim Forms on behalf of Physicians employed by or otherwise working with them at the time that the claims are made, without the necessity of individual signatures from the individual Physicians, but only if the Physician does not submit an individual claim on his/her own behalf. Notwithstanding the provisions in this § 8.3, an Active Physician Class Member against whom a Blue Plan has obtained a final finding of "fraud and/or abuse" from a judicial, arbitral, or governmental administrative proceeding and against whom that Blue Plan has obtained a corresponding final judgment for damages arising from a claim (or claims) for payment, during the same period for which claims may be asserted under § 8, shall not be entitled to receive payment from the Settlement Fund. For purposes of this § 8.3(h) only, the phrase "fraud and/or abuse" shall mean that the Active Physician Class Member engaged in any activities which are prohibited under 42 U.S.C. §§ 1320a-7, 1320a-7a, 1320a-7b, 1395nn, 1396b, 31 U.S.C. §§ 3729-3733, the federal CHAMPUS/TRICARE statute or other federal, state or local statutes or regulations related to the filing of false or fraudulent claims for payment for Covered Services or the making of false statements regarding the provision or payment of Covered Services. The prohibitions contained in these statutes or regulations include, but are not limited to, the following: (a) knowingly and willfully making or causing to be made a false statement or representation of a fact in any application for any benefit or payment; (b) knowingly and willfully making or causing to be made any false statement or representation of a fact for use in determining rights to any benefit or payment; (c) failing to disclose knowledge by a claimant of the occurrence of any event affecting the initial or continued right to any benefit or payment on its own behalf or on behalf of another, with intent to fraudulently secure such benefit or payment; and (d) knowingly and willfully soliciting or receiving any remuneration (including any kickback, bribe or rebate), directly or indirectly, overtly or covertly, in cash or in kind or offering to pay or receive such remuneration, including, but not limited to holding any financial interest in a provider of medical services that is prohibited under the federal law commonly referred to as the "Stark Law" (i) in return for referring an individual to a person for the furnishing or arranging for the furnishing of any item or service for which payment may be made in whole or in part by Medicare or Medicaid, or (ii) in return for purchasing, leasing, or ordering or arranging for or recommending purchasing, leasing, or ordering any good, facility, service or item for which payment may be made in whole or in part by Medicare or Medicaid. Each Blue Plan shall provide to the Settlement Administrator within ninety (90) days of the Notice Date the (i) name and tax identification number for each Class Member against whom the Blue Plan has obtained a finding of fraud and/or abuse from a

Page 20 of 45

judicial or administrative proceeding and (ii) the amount of the corresponding judgment for damages. A Blue Plan's failure to provide such information to the Settlement Administrator may be deemed a waiver of its right to object to the payment from the Settlement Fund for those Class Member(s) for whom the Blue Plan failed to timely provide the requisite information. The foregoing limitation shall not act to reduce the aggregate amount of the Settlement Fund.

8.4 Reversion to Charitable Organizations of Unclaimed Amounts

At a reasonable time determined by the Settlement Administrator, but not less than 120 days after all payments have been disbursed to Class Members or contributed at the direction of Class Members, in each case pursuant to § 8.3 of this Agreement, the Settlement Administrator shall determine, with respect to Class Members within each state, the amount of unclaimed funds remaining in the Settlement Fund including un-cashed checks and interest earned on such funds, but excluding taxes owed on such earnings (the "Reversion Amounts"). The Settlement Administrator shall provide written notice of the Reversion Amounts to the Blue Parties and Class Counsel and, no later than twenty (20) business days after providing such written notice, the Settlement Administrator shall cause the Settlement Fund to remit the Reversion Amounts to one or more of the charitable organizations listed on Exhibit L within the applicable state or if there is none to the Foundation. Following the Settlement Administrator's determination of the Reversion Amounts, stop payment orders may be placed on all unclaimed funds, and no Class Member shall have any claim on the Settlement Fund.

8.5 Payments of Interest

Each Settlement Fund Payment shall include interest at the Interest Rate accruing from the thirtieth (30th) day after the Preliminary Approval Date to the Effective Date, with no compounding.

8.6 Other Settlement Administration Provisions

- (a) The Parties and their respective counsel shall not have any responsibility for, interest in, or liability whatsoever with respect to: (i) the investment of or distribution of the Settlement Fund, (ii) the determination, administration, calculation, or payment of Claim Forms from the Settlement Fund, or (iii) any losses incurred in connection therewith.
- (b) No Person shall have any cause of action against the
 Representative Plaintiffs, Class Counsel, the Settlement
 Administrator, the Blue Parties, the Released Parties, or the Blue
 Parties' counsel, based on the administration or implementation of
 the Agreement or orders of the Court or based on the distribution

of monies under the Agreement, and the Person's sole remedy (other than those provided pursuant to the terms of the Agreement), shall be by an application to the Court for enforcement of the Agreement pursuant to § 12.

- (c) The Settlement Administrator shall make appropriate reports on Internal Revenue Code Form 1099 with respect to all payments it makes to Class Members under this Agreement. The Settlement Administrator shall file any tax returns necessary with respect to any income earned by the Settlement Fund and shall pay, as and when legally required to do so, any tax payments (including interest and penalties) due on income earned by such Fund, and shall request refunds, when and if appropriate, and shall apply any such refunds that are issued to the Settlement Fund to become a part thereof (or, if refunds are received after distribution, to the Foundation).
- (d) If the Final Order and Judgment is set aside or reversed, in whole or in part, for any reason, then at such time as the time for any appeal from the final order of set aside or reversal has elapsed (including without limitation any extension of time for the filing of any appeal that may result by operation of law or order of the Court) with no notice of appeal having been filed, all funds in the Settlement Administration Account, including interest accrued thereon, shall be released forthwith to the Blue Plans in proportion to their respective payments.
- 9. Attorneys' Fees, Representative Plaintiffs' Fees and Fees for Representative Plaintiffs in Other Actions
 - 9.1 The Blue Plans Shall Pay Attorneys' Fees

Class Counsel intend to apply to the Court for an award of Attorneys' Fees in an amount not to exceed \$49,061,965, (the "Unopposed Amount") which application the Blue Plans agree not to oppose. Each Blue Plan shall pay a portion of the amount of the Attorneys' Fees awarded by the Court ("Attorneys' Fees") in proportion to the percentage assigned to the Blue Plan in Exhibit K. The Blue Plans' total payments of Attorneys' Fees shall not exceed the Unopposed Amount and shall be made in accordance with § 9.3 of this Agreement. If any Blue Plan terminates its participation in this Agreement pursuant to §14.2, the Unopposed Amount shall be reduced by the percentage assigned to the Blue Plan in Exhibit K. If the Court awards Attorneys' Fees in excess of the Unopposed Amount, Class Counsel, on behalf of themselves and the Class, hereby covenant and agree to waive, release and forever discharge the amount of any such excess award and to make no effort of any kind or description ever to collect same. The Attorneys' Fees agreed to be paid pursuant to this provision are in addition to and separate from all other consideration and remedies paid to and available to the Class Members. The Blue Plans

Page 22 of 45

shall not be obligated to pay any attorneys' fees or expenses incurred by or on behalf of any Releasing Party in connection with the Action, other than the payment of Attorneys' Fees in accordance with this § 9.1.

9.2 The Blue Parties Shall Pay Representative Plaintiffs' Fees

In addition to Attorney's Fees, Class Counsel intends to apply to the Court for an award of fees for each Representative Plaintiff in the amount of \$7,500, which application the Blue Parties agree not to oppose. The Blue Parties shall pay such fees, not later than ten (10) days after the Effective Date to Representative Plaintiffs in the amount awarded by the Court up to but not exceeding such unopposed amount. If the Court awards fees to Representative Plaintiffs in excess of \$7,500 each, Representative Plaintiffs, on behalf of themselves and the Class, hereby covenant and agree to waive, release and forever discharge the amount of any such excess award and to make no effort of any kind or description ever to collect same. The fees to Representative Plaintiffs agreed to be paid pursuant to this § 9.2 are in addition to the other consideration afforded the Class Members. The Blue Parties shall support the award of fees to Representative Plaintiffs up to \$7,500 as reasonable and appropriate and shall not object to such request nor appeal an award up to that amount. The fees set forth in §§ 9.1 and 9.2 are the only consideration that Released Parties shall be obligated to give Class Counsel and the Representative Plaintiffs as a result of prosecuting and settling this Action, other than the additional express agreements made herein.

9.3 Timing of Fee Payments

Each Blue Plan shall wire transfer its portion of Attorneys' Fees to the trust account of Archie Lamb, PC, not later than ten (10) days after the Effective Date, with interest at the Interest Rate accruing from the thirtieth (30th) day following the Final Order Date to the Effective Date, with no compounding.

9.4 Fees to Representative Plaintiffs in Other Actions.

In addition to Attorney's Fees, Class Counsel intends to apply to the Court for an award of fees for each of the Representative Plaintiffs in Other Actions in the amount of \$7,500, which application the Blue Parties agree not to oppose. Those Blue Parties listed on Exhibit M shall pay such fees to the Representative Plaintiffs in Other Actions as designated on Exhibit M, not later than ten (10) days after the Effective Date in the amount awarded by the Court up to but not exceeding such unopposed amount. If the Court awards fees to Representative Plaintiffs in Other Actions in excess of \$7,500 each, Representative Plaintiffs in Other Actions, on behalf of themselves and the Class, shall covenant and agree to waive, release and forever discharge the amount of any such excess award and to make no effort of any kind or description ever to collect same. The fees to Representative Plaintiffs in Other Actions agreed to be paid pursuant to

this § 9.4 are in addition to the other consideration afforded the Class Members. The Blue Parties shall support the award of fees to Representative Plaintiffs in Other Actions up to \$7,500 as reasonable and appropriate and shall not object to such request nor appeal an award up to that amount. Such fees are the only consideration that Released Parties shall be obligated to give Representative Plaintiffs in Other Actions, other than the additional express agreements made herein.

10. Application to Fully-Insured Plans and Self-Insured Plans

Section 7 of this Agreement applies to each Blue Plan's conduct with respect to both Fully-Insured Plans and Self-Insured Plans, except: (i) as otherwise specified in this Agreement, or (ii) as provided by applicable law, including, without limitation, fulfillment of a Blue Plan's obligations as a plan fiduciary in connection with any Plan subject to the requirements of ERISA.

Limited Liability 11.

Case 1:03-cv-21296-FAM

11.1 **Settlement Implementation**

The Billing Dispute External Reviewer, the Independent Review Organization or Organizations (and members and agents, if any), the Class Compliance Dispute Facilitator (and agents, if any), the Blue Compliance Dispute Facilitators (and agents, if any), and the Compliance Dispute Officer (and agents, if any) (including, in the case of individual Compliance Dispute Facilitators and/or Compliance Dispute Officers, selected in accordance with § 12.7) do not owe a fiduciary duty to the Parties, and the Parties shall ask the Court to grant such persons limited immunity from liability for their actions under this Agreement, except for any willful misconduct or gross negligence.

11.2 **Blue Parties**

The liability of each Blue Party under this Agreement shall be several. No breach of a several obligation by a Blue Party shall result in any liability to any other Blue Party or provide any basis for any compliance action or other remedy against any other Blue Party.

Compliance Disputes Arising Under This Agreement 12.

Jurisdiction, Coordination and Costs 12.1

Compliance Dispute Facilitators (a)

Class Counsel shall appoint a Class Compliance Dispute Facilitator (the "Class Compliance Dispute Facilitator") to facilitate any disputes initiated by a Class Member within sixty (60) days after the Final Order Date. The Blue Parties shall appoint a Blue

Page 24 of 45

Compliance Dispute Facilitator (the "Blue Compliance Dispute Facilitator") to facilitate the resolution of any disputes within sixty (60) days after the Final Order Date; provided, however, that the Parties agree to hold any Compliance Dispute in abeyance until such time as each of the Parties' respective Compliance Dispute Facilitator and/or Compliance Dispute Officer has/have been appointed. All compliance dispute resolutions without full review under § 12.5 below shall be addressed by the Class Compliance Dispute Facilitator and the Blue Compliance Dispute Facilitator.

Compliance Dispute Officer. (b)

Class Counsel and the Blue Parties shall agree upon a Compliance Dispute Officer (the "Compliance Dispute Officer") within sixty (60) days after the Final Order Date.

All Compliance Dispute resolutions with full review under § 12.6 below shall be directed to the Compliance Dispute Officer. Compliance Disputes shall not be directed to the Court nor to any other state court, federal court, arbitration panel or any other binding or non-binding dispute resolution mechanism, except as provided in this Agreement.

One year after any Compliance Dispute Officer is appointed, either of Class Counsel or the Blue Parties may request the appointment of a new Compliance Dispute Officer. Class Counsel and the Blue Parties will agree upon a new Compliance Dispute Officer within thirty (30) days of any such request.

(c) **Program Title**

The compliance dispute program set forth in § 12.1 through § 12.6 shall be referred to as the "Joint Compliance Dispute Program".

Fees and Costs (d)

The Blue Parties shall pay the reasonable fees and expenses of the Compliance Dispute Officer, the Class Compliance Dispute Facilitator and the Blue Compliance Dispute Facilitator.

Who May Petition the Compliance Dispute Officer 12.2

Class Compliance Dispute Facilitator may petition the Compliance Dispute Officer on behalf of any Class Member or Signatory Medical Society.

Procedure for Initiating Compliance Disputes 12.3

To initiate a Compliance Dispute, a Class Member or Signatory Medical Society shall submit a completed Compliance Dispute Claim Form, attached hereto as Exhibit C, to the Class Compliance Dispute Facilitator. The Class Compliance Dispute Facilitator shall provide a copy of the completed Compliance Dispute Claim Form to the Blue Party's Compliance Dispute Facilitator.

12.4 Intentionally Left Blank

12.5 Dispute Resolution Without Full Review

- (a) If the Class Compliance Dispute Facilitator, in its sole judgment, determines that a dispute initiated by a Class Member or Signatory Medical Society is frivolous or the actual harm suffered is de minimis, it shall notify the Class Member or Signatory Medical Society and notify the affected Blue Compliance Dispute Facilitator that there will be no further proceedings on such dispute.
- (b) If the Class Compliance Dispute Facilitator, in its sole judgment, determines a dispute initiated by a Class Member or Signatory Medical Society requires full review (without any other attempt at resolution), it shall immediately refer the dispute to the Compliance Dispute Officer for full review and shall notify the Class Member or Signatory Medical Society and the Blue Compliance Dispute Facilitator of this referral.
- If the Class Compliance Dispute Facilitator, in its sole judgment, (c) determines a dispute initiated by a Class Member or Signatory Medical Society may be resolved without full review, the Class Compliance Dispute Facilitator and the Blue Compliance Dispute Facilitator shall assist in an attempt to achieve a resolution of the dispute agreed to by the Class Member or Signatory Medical Society and the affected Blue Party. All parties to such dispute agree to assist the Class Compliance Dispute Facilitator and the Blue Compliance Dispute Facilitator in such efforts. If such efforts do not achieve resolution of the dispute within sixty (60) days after submission of the Compliance Dispute Claim Form, the Class Compliance Dispute Facilitator shall refer the dispute to the Compliance Dispute Officer for full review and shall notify the Class Member or Signatory Medical Society and the Blue Compliance Dispute Facilitator of this referral.

12.6 Dispute Resolution With Full Review

(a) Requirements for a Compliance Dispute With Full Review.

Page 26 of 45

- (i) A Class Member or Signatory Medical Society submitted a properly completed Compliance Dispute Claim Form to the applicable Compliance Dispute Facilitator not later than ninety (90) days after such Compliance Dispute arose or reasonably could be known, whichever is later; and
- (ii) The Compliance Dispute Officer, in its sole judgment, determines that the Compliance Dispute:
 - (1) alleges a failure by a Blue Party to comply with an obligation under Section 7 of this Agreement; and
 - (2) is not properly the subject of a proceeding under §§ 7.10 or 7.11 of this Agreement;

(b) Memoranda to Compliance Dispute Officer.

Case 1:03-cv-21296-FAM

- (i) The Compliance Dispute Officer shall in writing request memoranda from the parties to the Compliance Dispute regarding the merits of the dispute and appropriate remedies. The complaining Class Member or Signatory Medical Society shall have fifteen (15) days from the date of such request to submit its memorandum, and the other party shall submit its memorandum in response within fifteen (15) days after receipt of the complaining Class Member's or Signatory Medical Society's memorandum.
- (ii) In the event that the Compliance Dispute relates to the Blue Plan's compliance with § 7.8(d)(ii) of this Agreement, the Blue Plan may present directly to the Compliance Dispute Officer ex parte and in camera the Blue Plan's arguments in support of the Clinical Information submission request, including, where applicable, arguments that the Blue Plan is invoking its right to obtain Clinical Information pursuant to § 7.8(d)(ii) because the Blue Plan has a reasonable basis for believing that an investigation of fraudulent or abusive (whether intentional or unintentional) billing practices is warranted. The Compliance Dispute Officer shall not disclose any information designated as "Confidential" by the Blue Plan.

(c) Oral Argument of Compliance Dispute.

The Compliance Dispute Officer, at its sole option, may request the parties to the Compliance Dispute to present oral argument of

Page 27 of 45

the Compliance Dispute at a time and place agreed to by the parties to the Compliance Dispute and the Compliance Dispute Officer.

(d) Decisions by the Compliance Dispute Officer.

- (i) In resolving a Compliance Dispute, the Compliance Dispute Officer shall provide a written decision, made in its sole judgment and based only on the Compliance Dispute Claim Form with supporting documents and testimony by affidavit, memoranda, and any oral argument, determining whether the affected Blue Party has failed to comply with its obligations under § 7 of this Agreement, and if so, directing what actions are to be taken by the affected Blue Party to fulfill its obligations under § 7 of this Agreement.
- In the event that the Compliance Dispute relates to the Blue (ii) Plan's compliance with § 7.8(d)(ii), the Compliance Dispute Officer's written decision to the parties to the Compliance Dispute shall state only that (a) the Blue Plan has complied with § 7.8(d)(ii) and the matter is thus closed, or (b) that the Blue Plan has not complied with § 7.8(d)(ii) and the requirement for submission of Clinical Information is to cease, as to that dispute. This limitation on the content of the Compliance Dispute's written opinion applies to all Compliance Disputes relating to the Blue Plan's compliance with § 7.8(d)(ii), regardless of whether the Compliance Dispute involves a Clinical Information submission request in connection with a Blue Plan's investigation of fraudulent or abusive billing practices. In the event that the Blue Plan has invoked its right to obtain Clinical Information, pursuant to § 7.8(d)(ii), because the Blue Plan has a reasonable basis for believing that an investigation of fraudulent or abusive (whether intentional or unintentional) billing practices is warranted, the Compliance Dispute Officer's sole responsibility in these circumstances shall be to make a determination as to whether the Blue Plan has a reasonable basis for its belief.

Intentionally left blank (e)

(f) Systemic Remedy

If the Compliance Dispute Officer determines, after adequate notice and a reasonable opportunity to respond consistent with this § 12, that the affected Blue Party has not fulfilled its obligations under § 7 of this Agreement with respect to other Class Members beyond the complaining Class Member or Signatory Medical Society in the Compliance Dispute, the Compliance Dispute

Page 28 of 45

Officer may order appropriate remedies only as necessary and designed to obtain compliance with the terms of this Agreement.

Finality of the Compliance Dispute Officer's Decision. (g)

The Parties agree that the decision of the Compliance Dispute Officer shall be final unless appealed to the Court. The Parties further agree that the scope of review by the Court shall be limited to whether the Compliance Dispute Officer's final decision was "arbitrary, capricious, an abuse of discretion, or otherwise not in accordance with law," as defined by 5 U.S.C. § 706(2)(A). If the Court finds such final decision was "arbitrary and capricious, an abuse of discretion, or otherwise not in accordance with law", the Court may remand the Compliance Dispute to the Compliance Dispute Officer for further proceedings.

Enforcement by the Court. (h)

If the Compliance Dispute Officer determines that the affected Blue Party has not complied with its final decision, it shall provide written notice of such noncompliance to the affected Blue Party. If the affected Blue Party does not comply within thirty (30) days from the date of such notice, the Compliance Dispute Officer shall provide written notice to the Class Member or Signatory Medical Society in the Compliance Dispute, and the Class Member or Signatory Medical Society may petition the Court for enforcement.

Individual Plan Compliance Dispute Program 12.7

The Blue Parties listed on Exhibit P will use an individual plan compliance dispute program that follows the requirements set out in §§12.1 through 12.6 above, except each listed Blue Party and Class Counsel shall agree upon an individual Compliance Dispute Officer and each will select an individual Compliance Dispute Facilitator in accordance with the requirements in section 12.1. The Blue Parties that use an individual plan compliance dispute program may not use the Joint Compliance Dispute Program, unless agreed to in writing by Class Counsel.

12.8 Compliance Reporting.

During the Effective Period, each Blue Party shall provide to a (a) designated Blue Compliance Reporting Officer at BCBSA information regarding its compliance with the obligations set forth in § 7 of this Agreement. Each Blue Plan will take steps reasonably necessary to monitor and certify its compliance with this Agreement and will designate an appropriate representative to be responsible for this activity. Within thirty (30) days after the end

Page 29 of 45

of each calendar year beginning in 2008, during the applicable Effective Period, the Blue Compliance Reporting Officer shall provide a written summary report of such information to the Compliance Dispute Officer. A copy of this report shall be provided by the Compliance Dispute Officer to any Party upon request. The summary report will contain the following information from each Blue Plan:

7.5	Blue Plan's standard Precertification lists		
7.6	Blue Plan's list of material adverse changes		
7.8(b)	Blue Plan's list of efforts to cause its claims payment systems to be consistent within each state in which the Blue Plan operates		
7.8(d)	Blue Plan's list of Significant Edits		
7.8(d)(i)	Blue Plan's list of customized Edits added to any standard claims editing software product at Blue Plan's request		
7.8(d)(ii)	Blue Plan's list of categories of claims for which there is routine review of Clinical Information		
7.8(d)(iii)	Blue Plan's list of any circumstances where particular services or procedures, relative to modifiers 25 and 59, are not appropriately reported together with those modifiers		
7.9(b)	Blue Plan's list of the dates of meetings of the Physician Advisory Committee and the members of the Physician Advisory Committee		
7.10(f)	Blue Plan's list of decisions issued by the Billing Dispute External Reviewer which required payment by the Blue Plan		
7.16(a)	Blue Plan's list of the number of Adverse Determinations sent to External Review for final determination for the preceding calendar year and the percentage of such Adverse Determinations that are reversed.		
7.23	Blue Plan's list of actions to improve accuracy of information about eligibility of Plan Members		

Page 30 of 45

	Blue Parties' approximate aggregate amount spent as provided in § 7.31 to be filed no later than thirty (30) days
1 1-	prior to the Termination Date.

(b) A Blue Party may elect not to participate in the joint compliance reporting program set out in § 12.8(a) above, and instead establish its own individual plan compliance reporting program, except for the reporting of information relating to compliance with § 7.31, which shall be included in the report filed under § 12.8(a). All Blue Parties electing to have an individual plan compliance reporting program are on the list attached as Exhibit Q. Each Blue Party electing to use an individual plan compliance reporting program will designate a local compliance reporting officer who will follow the requirements and procedures as set out for the joint compliance reporting program in § 12.8(a) above.

12.9 Compliance Monitoring

For purposes of monitoring compliance with this Agreement, each Blue Plan shall allow the Compliance Dispute Officer and the Class Compliance Dispute Facilitator access to its Provider Website; provided, however, that such access shall not include access to Physician fee schedules and protected health information as defined by HIPAA.

12.10 Compliance Dispute/Negotiated Purchaser Designation

In the event that a Compliance Dispute is brought by the Class (a) Compliance Dispute Facilitator, regarding a term or condition subject to § 7.36(a), the matter shall be submitted directly to the Compliance Dispute Officer. The Blue Plan may present directly to the Compliance Dispute Officer with a copy to the Class Compliance Dispute Facilitator, on a confidential basis, the Blue Plan's reasons for its action, including that the purchaser qualifies as a Negotiated Purchaser as defined in § 7.36(a)(i) and that the term or condition was reasonably necessary to obtain or maintain the service agreement or insurance contract. The Class Compliance Dispute Facilitator may respond to the submission of the Blue Plan within five (5) business days of receipt of a copy of the submission. The Compliance Dispute Officer's decision shall be limited to whether the purchaser is a Negotiated Purchaser, whether the term or condition was reasonably necessary, and, as a result, whether the term or condition is permitted under § 7.36. The Compliance Dispute Officer and the Class Compliance Dispute Facilitator shall not disclose any information that the Blue Plan has submitted as confidential. The Class Compliance Dispute Facilitator may confer with one or more of Class Counsel in connection with the dispute if the Class Compliance Dispute

Page 31 of 45

Facilitator does not disclose any confidential information, or if the Class Counsel involved agrees to keep the information confidential.

(b) The Compliance Dispute Officer shall render a final decision within ten (10) days following submission of information required by the Blue Plan under § 7.36(a). If the Compliance Dispute Officer rules against the Blue Plan, any remedy shall be limited to directing the Blue Plan to submit and implement an adequate corrective action plan, and imposing on the Blue Plan the obligation to pay actual damages including interest suffered by a Class Member as a result of the implementation of any modification. Any activity conducted by a Blue Plan under a corrective action plan shall be subject to the provisions of §13.3(a)(iii). The Class Compliance Dispute Facilitator may confer with and seek the assistance of Class Counsel in connection with any such dispute.

13. Release, Covenant Not to Sue, and Bar Order

13.1 Discharge of All Released Claims

(a)

Upon the Effective Date, the "Released Parties," which shall include the Blue Parties and each of their present and former parents, divisions and Affiliates and each of their respective current or former officers, directors, employees, agents, insurers and attorneys (and the predecessors, heirs, executors, administrators, legal representatives, successors and assigns of each of the foregoing), and all Persons who provided claims processing services, software, proprietary guidelines or technology to the Blue Parties, their subsidiaries or Affiliates, and those contracted agents processing claims on their behalf (including, without limitation, NASCO), together with each such Person's predecessors or successors (but only to the extent of such Person's services and work was done pursuant to contract with the Blue Parties or their subsidiaries or Affiliates), but excluding all Delegated Entities, shall be released and forever discharged by the Signatory Medical Societies and all Class Members, and by their respective current and former officers, directors, employees, attorneys, heirs, executors, administrators, agents, legal representatives, professional corporations, partnerships, assigns and successors, but only to the extent claims are derived by contract or operation of law from the claims of Class Members, (collectively, the "Releasing Parties") from any and all causes of action, judgments, liens, indebtedness, costs, damages, obligations, attorneys' fees, losses, claims, liabilities and demands of whatever kind, source or character whether arising under any federal or state law, which (consistent with the Parties' understanding of the settlements in Shane) includes, but is not limited to, the Racketeer

Page 32 of 45

Influenced and Corrupt Organizations Act, antitrust and other statutory and common law claims, intentional or non-intentional, (each a "Claim"), arising on or before the Effective Date, that are, were or could have been asserted against any of the Released Parties by reason of, arising out of, or in any way related to any of the facts, acts, events, transactions, occurrences, courses of conduct, business practices, representations, omissions, circumstances or other matters referenced in the Action, or addressed in this Agreement, whether any such Claim was or could have been asserted by any Releasing Party on its own behalf or on behalf of other Persons, or to the business practices that are the subject of § 7. This includes, without limitation and as to Released Parties only, any aspect of any fee for service claim submitted by any Class Member to a Blue Plan, and any claims of any Class Member related to or based upon any Capitation agreement between a Blue Plan and any Class Member or other Person or entity, or the delay, nonpayment or amount of any Capitation payments by a Blue Plan, and any allegation that any Blue Party has conspired with, aided and abetted, or otherwise acted in concert with other managed care organizations, other health insurance companies, Delegated Entities and/or other third parties with regard to any of the facts, acts, events, transactions, occurrences, courses of conduct, business practices, representations, omissions, circumstances or other matters related to the Action, or with regard to any Blue Party's liability for any other demands for payment submitted by any Class Member to such other managed care organizations, health insurance companies, Delegated Entities and/or other third parties.

Except for Claims for damages against defendants in the Action (b) that are not Parties relating to providing Covered Services to Blue Plans' Plan Members in connection with the BlueCard Program and other similar national account delivery programs governed by BCBSA (including but not limited to NASCO-to-NASCO arrangements), the Releasing Parties further agree to forever abandon and discharge any and all Claims that exist now or that might arise in the future against any other Persons which Claims arise from, or are based on, conduct by any of the Released Parties that occurred on or before the Effective Date and are, or could have been, alleged in the Complaint, whether any such Claim was or could have been asserted by any Releasing Party on its own behalf or on behalf of other Persons. Nothing in this Agreement is intended to relieve any Person or entity that is not a Released Party from responsibility for its own conduct or conduct of other Persons who are not Released Parties, or to preclude any Representative Plaintiff from introducing any competent and admissible evidence to the extent consistent with §§ 14 and 16.

Page 33 of 45

(c) Any Claims released or discharged pursuant to §§ 13.1(a) or 13.1(b), subject to the exception regarding Retained Claims contained in § 13.6, shall be referred to as "Released Claims."

13.2 Covenant Not to Sue

Case 1:03-cv-21296-FAM

- (a) The Releasing Parties and each of them agree and covenant not to sue or prosecute, institute or cooperate in the institution, commencement, filing, or prosecution of any suit or proceeding in any forum based upon or related to any Released Claims against any Released Party.
- (b) Notwithstanding any other provision of this Agreement (including, without limitation, this § 13.2), nothing in this Agreement shall be deemed to in any way impair, limit, or preclude the Releasing Parties' rights to enforce any provision of this Agreement, or any court order implementing this Agreement, in a manner consistent with the terms of the Agreement.

13.3 Bar Order

It is an essential element of the Agreement that the Released Parties obtain the fullest possible release from further liability to anyone relating to the Released Claims, and it is the intention of the Parties to this Agreement that the Agreement eliminate all further risk and liability of the Released Parties relating to the Released Claims. Accordingly, the Parties agree that the Court shall include in the Final Approval Order a Bar Order provision that meets all of the following requirements:

The Releasing Parties are permanently enjoined from: (i) filing, (a) commencing, prosecuting, intervening in, participating in (as class members or otherwise) or receiving any benefits from any lawsuit, arbitration, administrative or regulatory proceeding or order in any jurisdiction based on any or all Released Claims against one or more Released Parties; (ii) instituting, organizing class members in, joining with class members in, amending a pleading in or soliciting the participation of class members in, any action or arbitration, including but not limited to a purported class action, in any jurisdiction against one or more Released Parties based on, involving, or incorporating, directly or indirectly, any or all Released Claims; and (iii) filing, commencing, prosecuting, intervening in, participating in (as class members or otherwise) or receiving any benefits from any lawsuit, arbitration, administrative or regulatory proceeding or order in any jurisdiction based on an allegation that an action of the Blue Parties, which is in compliance with the provisions of the Settlement Agreement, violates any legal right of any Class Member.

(b)

All Persons who are, have been, could be, or could have been alleged to be joint tortfeasors, co-tortfeasors, co-conspirators, or co-obligors with the Released Parties or any of them respecting the Released Claims or any of them, are hereby, to the maximum extent permitted by law, barred and permanently enjoined from making, instituting, commencing, prosecuting, participating in or continuing any Claim, claim-over, cross-claim, action, or proceeding, however denominated, regardless of the allegations, facts, law, theories or principles on which they are based, in this Court or in any other court or tribunal, against the Released Parties or any of them with respect to the Released Claims, including without limitation equitable, partial, comparative, or complete contribution, set-off, indemnity, or otherwise, whether by contract, common law or statute, arising out of or relating in any way to the Released Claims. All such claims are hereby fully and finally barred, released, extinguished, discharged, satisfied and madeunenforceable to the maximum extent permitted by law, and no such claim may be commenced, maintained, or prosecuted against any Released Party. Any judgment or award obtained by a Class Member against any such Person shall be reduced by the amount or percentage, if any, necessary under applicable law to relieve any Released Party of all liability to such Person on such barred claims. Such judgment reduction, partial or complete release, settlement credit, relief, or setoff, if any, shall be in an amount or percentage sufficient under applicable law as determined by the Court to compensate such Person for the loss of any such barred claims against any Released Party. Where the claims of a Person who is, has been, could be, or could have been alleged to be a joint tortfeasor, co-tortfeasor, co-conspirator or co-obligor with a Released Party respecting the Released Claims have been barred and permanently enjoined by this § 13.3, the claims of Released Parties against that Person respecting those Released Claims are similarly fully and finally barred, released, extinguished, discharged, satisfied and made unenforceable to the maximum extent permitted by law.

Dismissal With Prejudice 13.4

The Releasing Parties shall dismiss the Action with prejudice as to Released Parties. It is the Parties' intention that such dismissal shall constitute a final judgment on the merits to which the principles of res judicata shall apply to the fullest extent of the law as to the Released Parties.

Waiver of California Civil Code § 1542 and Similar Provisions 13.5

Page 35 of 45

- (a) the provisions, rights and benefits conferred by § 1542 of the California Civil Code, which provides:
 - "A GENERAL RELEASE DOES NOT EXTEND TO CLAIMS WHICH THE CREDITOR DOES NOT KNOW OR SUSPECT TO EXIST IN HIS OR HER FAVOR AT THE TIME OF EXECUTING THE RELEASE, WHICH IF KNOWN BY HIM OR HER MUST HAVE MATERIALLY AFFECTED HIS OR HER SETTLEMENT WITH THE DEBTOR", and
- any law of any state or territory of the United States, federal law (b) or principle of common law, or of international or foreign law, which is similar, comparable or equivalent to § 1542 of the California Civil Code. Each Class Member and each Signatory Medical Society may hereafter discover facts other than or different from those which he, she or it knows or believes to be true with respect to the Claims which are the subject matter of the provisions of § 13, but each such Class Member and each Signatory Medical Society hereby expressly and unconditionally waives and fully, finally and forever settles and releases, upon the entry of Final Order and Judgment, any known or unknown, suspected or unsuspected, contingent or non contingent Claim, with respect to the subject matter of the provisions of § 13, whether or not concealed or hidden, without regard to the discovery or existence of such different or additional facts.

Retained Claims 13.6

Notwithstanding the foregoing, the Releasing Parties are not releasing claims for payment (each a "Retained Claim" and, collectively, the "Retained Claims") for Covered Services provided to Plan Members prior to or on the Effective Date as to which, as of the Effective Date: (i) no claim with respect to such Covered Services has been submitted to a Blue Plan, provided that the applicable period for filing such claim has not elapsed; or (ii) a claim with respect to such Covered Services has been filed with a Blue Plan but such claim has not been finally adjudicated by the Blue Plan. For purposes of clause (ii), above, final adjudication shall mean completion of the Blue Plan's internal appeals process. In the event that a claim referred to in clause (ii) is finally adjudicated less than thirty (30) days prior to the Effective Date, such claim shall constitute a Retained Claim if a Physician seeks relief under § 7.10 not later than ninety (90) days after notice of such final adjudication, but otherwise such claim shall constitute a Released Claim. Disputes relating to Retained

Page 36 of 45

Claims shall be resolved exclusively pursuant to the appropriate remedial provisions of this Agreement.

13.7 Covenant Not to Sue on Retained Claims and Compliance Disputes

Upon the Effective Date and through the Termination Date, the Releasing Parties and each of them agree and covenant not to sue or prosecute, institute or cooperate in the institution, commencement, filing, or prosecution of any suit or proceeding against any Released Party, in any forum (i) any Retained Claim or (ii) any Compliance Dispute (whether or not asserted), which respectively shall be asserted and pursued only pursuant to Sections 12 and 13.6 of this Agreement. This § 13.7 shall not apply to: (i) any claims that arise within twenty (20) days before the Termination Date that could not reasonably be presented or resolved pursuant to the procedures set forth in this Agreement; provided that any such claim shall be prosecuted on an individual basis only and not otherwise; or (ii) any remedy under a state or federal law or applicable regulation elected by a Physician pursuant to § 7.29(m).

Non-Released Persons and Non-Released Claims 13.8

- Nothing in this Agreement is intended to relieve any Person that is (a) not a Released Party from responsibility for its own conduct or conduct of other Persons who are not Released Parties for claims that are not Released Claims. Nothing in this Agreement is intended to preclude any Representative Plaintiffs from introducing any competent and admissible evidence to the extent consistent with §§ 14 and 16.
- Except as provided in § 13.1, nothing in this Agreement prevents (b) the Representative Plaintiffs and the Class from pursuing claims to hold any Person that is not a Released Party liable for damages caused by that Person's conduct in a conspiracy involving any Released Party.

Irreparable Harm 13.9

The Parties agree that the Blue Parties shall suffer irreparable harm if a Releasing Party takes action inconsistent with §§ 13.1, 13.2, 13.3, 13.4 and/or 13.7, and that in that event the Blue Parties may seek an injunction from the Court as to such action without further showing of irreparable harm.

13.10 Legislative Changes

Nothing contained in this Agreement is intended, or shall be construed, to preclude any Party from seeking legislative or regulatory changes as to matters addressed herein or from seeking to enforce any such changes using any available legal remedy.

14. Stay of Discovery, Termination, and Effective Date of Agreement

14.1 Suspension of Discovery and Other Proceedings

- Except as set forth in Exhibit O with respect to BCBSA, upon (a) execution of this Agreement, the Releasing Parties and Class Counsel shall discontinue all discovery activity or related proceedings in the Action against the Released Parties, provided that if this Agreement is terminated pursuant to § 14.2 as to a Blue Party, the Releasing Parties and Class Counsel may pursue discovery or related proceedings solely against the Blue Party as to which the Agreement is terminated and any defendant in this Action that has not settled. Where a Blue Plan provides or has provided claims processing services for a non-settling defendant or a Terminating Blue Plan, and such Blue Plan is in the possession of that non-settling defendant's or Terminating Blue Plan's claims data pursuant to the provision of those services, the foregoing sentence shall not prevent the Releasing Parties or Class Counsel from obtaining discovery of those claims data pursuant to a discovery request issued to the non-settling defendant or Terminating Blue Plan. Notwithstanding the foregoing, nothing herein shall require a Blue Plan to provide any data concerning any of its claims, including, but not limited to, claims under the BlueCard Program or any similar national account delivery program governed by BCBSA (including but not limited to NASCO-to-NASCO arrangements).
- (b) Upon entry of the Preliminary Approval Order, the Releasing Parties and Class Counsel agree to suspend and stay all proceedings in the Action against the Released Parties, other than proceedings as may be necessary to carry out the terms and conditions of the Agreement. The Preliminary Approval Order shall also bar and enjoin all Class Members from commencing or prosecuting any action asserting any Released Claims, and stay any other actions or proceedings brought by any Class Members asserting any Released Claims. If, and only to the extent that, this Agreement terminates pursuant to § 14.2 of this Agreement, the Parties shall not be deemed to have waived any rights as a result of such suspension or stay of proceedings.
- (c) Upon the Effective Date, and notwithstanding any of the other provisions in this Agreement, the Released Parties shall have no obligation to preserve documents and evidence with respect to Released Claims, including any obligations under the Agreed Order for Preservation of Records, and the Releasing Parties and Class Counsel shall not pursue any spoliation claims or other

Page 38 of 45

actions or sanctions against a Released Person with respect to documents or evidence related to the Released Claims.

14.2 Right to Terminate this Agreement

Case 1:03-cv-21296-FAM

- (a) If, at the Preliminary Approval Hearing or within thirty (30) days thereafter, the Court does not enter the Preliminary Approval Order and approve the Claim Form, Claim Form Instructions, Mailed Notice, and the Published Notice submitted to the Court pursuant to § 4 of this Agreement, in each case in substantially the same form as Exhibits A, B, E, and G, any Party shall have the right, in the sole and absolute discretion of such Party, to terminate its participation in this Agreement by delivering a notice of termination to the other Parties within forty-five (45) days following the Preliminary Approval Hearing.
- (b) If the Court does not grant the stay and injunctions as provided in § 14.1(b) and the interim injunction with respect to the Tag Along Actions, as provided in § 15.1, each in the form contained in the Preliminary Approval Order (Exhibit F), each Blue Party may in its sole and absolute discretion terminate its participation in this Agreement by delivering a notice of termination to the other Parties within forty-five (45) days following the Preliminary Approval Hearing.
- (c) If the number of putative Class Members submitting Opt-Out requests exceeds five percent (5%) of the total number of putative Class Members in the United States, each Blue Party may in its sole and absolute discretion terminate its participation in this Agreement by delivering a notice of termination to the other Parties within thirty (30) days of receipt of the complete list of Opt-Out requests from the Notice Administrator referred to in § 6.
- (d) If the number of putative Class Members in any Blue Plan's service area within a state submitting Opt-Out requests exceeds five percent (5%) of the total number of putative Class Members in that Blue Plan's service area within that state, except as provided in Exhibit H, the affected Blue Plan may in its sole and absolute discretion terminate its participation in this Agreement by delivering a notice of termination to the other Parties within thirty (30) days of receipt of the complete list of Opt-Out requests from the Notice Administrator referred to in § 6.
- (e) If the putative Class Members in any Blue Plan's service area submitting Opt-Out requests received, in the aggregate, at least five percent (5%) of the total dollar payments that such Blue Plan made to putative Class Members in calendar year 2006, the affected Blue Plan may in its sole and absolute discretion terminate

Page 39 of 45

its participation in this Agreement by delivering a notice of termination to the other Parties within thirty (30) days of receipt of the complete list of Opt-Out requests from the Notice Administrator referred to in § 6.

- (f) If the Court has not entered the Final Order and Judgment substantially in the forms attached hereto as Exhibit D by the date that is 180 days after the Preliminary Approval Date, any Party may, in its sole and absolute discretion, terminate its participation in this Agreement by delivering a notice of termination to the other Parties within 200 days after the Preliminary Approval Date.
- (g) If more than twenty-five percent (25%) of the total number of Blue Parties validly terminate their participation in this Agreement pursuant to this § 14.2, Class Counsel and each of the remaining Blue Parties may in its sole and absolute discretion terminate its participation in this Agreement by delivering a notice of termination to the other Parties, within thirty (30) days following the last date on which a termination notice may be provided under §§ 14.2(a)-(f) of this Agreement.
- (h) In the event of any termination of the Agreement in its entirety by Class Counsel or all Blue Parties collectively pursuant to the terms of this § 14.2, the Parties shall be restored to their original positions, except as expressly provided herein. In the event of any termination of individual participation in the Agreement by any Blue Plan (a "Terminating Blue Plan"), the Terminating Blue Plan, the Releasing Parties and Class Counsel shall be restored to their original positions, except as expressly provided herein.

14.3 Intentionally Left Blank

14.4 Effective Date

If the Final Order and Judgment is entered by the Court and the time for appeal from all of such orders and judgment has elapsed (including without limitation any extension of time for the filing of any appeal that may result by operation of law or order of the Court) with no notice of appeal having been filed, the "Effective Date" shall be the next business day after the last date on which notice of appeal could have been timely filed. If the Final Order and Judgment is entered and an appeal is filed as to any of them, the "Effective Date" shall be the next business day after the Final Order and Judgment is affirmed, all appeals are dismissed, and no further appeal to, or discretionary review in, any court remains. Notwithstanding the foregoing, at any time after the Final Order and Judgment is entered by the Court but before the Effective Date would otherwise occur, any Blue Party may in its sole and absolute discretion elect to declare that the Effective Date for its participation in this

Agreement has occurred by providing notice to the Parties, and thereafter that electing Blue Party, Class Counsel and the Releasing Parties shall be bound by this Agreement and shall perform their respective obligations hereunder as if the Final Order and Judgment had been affirmed in its entirety.

14.5 Intentionally Left Blank

14.6 Termination Date of Agreement

- This Agreement shall terminate (the "Termination Date") upon (a) the earliest to occur of (i) as to a Party, that Party's termination of this Agreement pursuant to § 14.2, above, (ii) the date on which the appellate court rejects, in whole or in part, the Final Order and Judgment, upon any appeal or discretionary review thereof, and no further appeal to, or discretionary review in, any court remains, or (iii) the four year anniversary of the Preliminary Approval Date. As of the Termination Date, the provisions of this Agreement shall immediately become void and of no further force and effect and there shall be no liability under this Agreement on the part of any of the Parties, except for willful or knowing breaches of this Agreement prior to the Termination Date; provided that in the event of a termination of this Agreement under clause (iii) of this § 14.6(a), (x) the provisions of §§ 13.1, 13.2, 13.3, 13.4, 13.5, 13.7, 13.8, 14.1, 15 and 16, shall survive such termination indefinitely, (y) the provisions of § 7.10 and § 7.11 shall survive such termination only with respect to, and only for so long as is necessary to resolve, any Billing Disputes that are in the process of being resolved as of the Termination Date and any disputes described in § 7.11 as of the Termination Date, and (z) the provisions of §§ 12.1 through 12.6 shall survive such termination only with respect to, and only for so long as is necessary to resolve, any Compliance Disputes involving a Blue Plan that are in the process of being resolved by the Compliance Dispute Officer as of the Termination Date.
- (b) On the Termination Date, all of a Blue Party's obligations under this Agreement shall be deemed satisfied except as and only to the extent that a term of § 7 of this Agreement has been extended by the Blue Party as provided in this § 14.6(b) as to that Blue Party only. No decision or ruling of the Compliance Dispute Officer shall have any force on the Parties after the Termination Date, and the Blue Parties shall be under no obligation to continue performance of any kind under this Agreement. A Blue Party may, in its sole and absolute discretion, elect to continue after the Termination Date, the implementation of various business practices described in this Agreement. A Blue Party also may,

Page 41 of 45

where it has a good faith basis, and notwithstanding any Implementation Date in § 7 of this Agreement or in Exhibit I hereto, delay the implementation, in whole or in part, of any provision of this Agreement upon notice to Class Counsel. Whether or not the foregoing notice is provided, in the event that implementation is delayed other than as provided in § 7, the term of the Agreement shall be extended solely with respect to the delayed provision for a period of time equal to the length of the delay.

15. Related Actions

15.1 Ordered Stays and Dismissals in Tag-Along Actions

As to any action brought by or on behalf of Class Members that asserts any claim that as of the Effective Date would constitute a Released Claim against any Blue Parties, other than the Action, that has been, or will in the future, be designated as a tag-along action or consolidated with the Provider Track actions under MDL Docket No. 1334 (the "Tag-Along Actions" (a list of such pending Tag-Along Actions is attached hereto as Exhibit N to this Agreement)), Class Counsel and relevant Parties shall cooperate to obtain an order of the Court, to be included in the Preliminary Approval Order, providing for the interim stay of all proceedings as to any Blue Parties in each such action pending entry of the Final Order and Judgment. In addition, no later than ten (10) business days after the Effective Date, Class Counsel and relevant Parties will submit an order of dismissal with prejudice as to any Blue Parties for each such pending Tag-Along Action on Exhibit N; provided that no such dismissal order shall be sought with respect to any Tag-Along Action with respect to any named plaintiff that has submitted a valid and timely Opt-Out request.

15.2 Certain Related State Court Actions

As to any action that is now pending in, or hereafter may be filed in or remanded to, any state court that asserts any Released Claim against any Blue Party on behalf of any Releasing Party, Class Counsel and the relevant Parties agree that they will cooperate with the Blue Party and file all documents necessary (a) to obtain an interim stay of all proceedings against the Blue Party and (b) on or promptly after the Effective Date, to obtain the dismissal with prejudice of any such action, other than with respect to any named plaintiff in such action that has submitted a valid and timely Opt-Out request.

15.3 Other Related Actions

As to any action not referred to in §§ 15.1 or 15.2 that is now pending or hereafter may be filed in any court that asserts any of the Released Claims against any Blue Parties on behalf of any Releasing Party, Class Counsel and the relevant Parties agree that they will cooperate with the Blue Party,

to the extent reasonably practicable, in the Blue Party's efforts to seek relief from the Court or the forum court to obtain the interim stay and dismissal with prejudice of such action as to any Blue Parties to the extent necessary to effectuate the other provisions of this Agreement.

Not Evidence; No Admission of Liability 16.

In no event shall this Agreement, in whole or in part, whether effective, terminated, or otherwise, or any of its provisions or any negotiations, statements, or proceedings relating to it in any way be construed as, offered as, received as, used as or deemed to be evidence of any kind in the Action, in any other action, or in any judicial, administrative, regulatory or other proceeding, except the Agreement may be used in a proceeding to enforce this Agreement. Without limiting the foregoing, neither this Agreement nor any related negotiations, statements or proceedings shall be construed as, offered as, received as, used as or deemed to be evidence, or an admission or concession of liability or wrongdoing whatsoever or breach of any duty on the part of the Blue Parties, the Representative Plaintiffs, or the Signatory Medical Societies, or as a waiver by the Blue Parties, the Representative Plaintiffs or the Signatory Medical Societies of any applicable defense, including without limitation any applicable statute of limitations. None of the Parties waives or intends to waive any applicable attorney-client privilege or work product protection or other privilege for any negotiations, statements or proceedings relating to this Agreement. This provision shall indefinitely survive the termination of this Agreement.

Entire Agreement; Amendment 17.

Entire Agreement 17.1

This Agreement, including its Exhibits, contains an entire, complete, and integrated statement of each and every term and provision agreed to by and among the Parties; it is not subject to any condition not provided for herein. This Agreement supersedes any prior agreements or understandings, whether written or oral, between and among Representative Plaintiffs, Class Members, Class Counsel, Blue Parties and the Signatory Medical Societies regarding the subject matter of the Action or this Agreement. This Agreement shall not be modified in any respect except by a writing executed by Class Counsel and each of the Blue Parties, except as provided in § 17.2.

17.2 **Modification Generally**

This Agreement may be amended or modified with respect to a particular Blue Party only as provided in a written instrument signed by or on behalf of that Blue Party and Class Counsel (or their successors in interest), or as set forth in § 7.36.

17.3 Intentionally Left Blank

18. No Presumption Against Drafter

None of the Parties shall be considered to be the drafter of this Agreement or any provision hereof for the purpose of any statute, case law, or rule of interpretation or construction that would or might cause any provision to be construed against the drafter hereof. This Agreement was drafted with substantial input by all Parties and their counsel, and no reliance was placed on any representations other than those contained herein. The Parties agree that this fully integrated Agreement shall be construed by its own terms and not by referring to, or considering, the terms of any other settlement agreement between plaintiffs and another defendant in the Action or Shane.

19. Captions and Headings

The use of captions and headings in this Agreement is solely for convenience and shall have no legal effect in construing the provisions of this Agreement.

20. Continuing Jurisdiction and Exclusive Venue

20.1 Continuing Jurisdiction

Except as otherwise provided in this Agreement, it is expressly agreed and stipulated that the United States District Court for the Southern District of Florida shall have exclusive jurisdiction and authority to consider, rule upon, and issue a final order with respect to proceedings, whether judicial, administrative or otherwise, which may be instituted by any Person, individually or derivatively, with respect to this Agreement. This reservation of jurisdiction does not limit any other reservation of jurisdiction in this Agreement nor do any other such reservations limit the reservation in this subsection. Except as otherwise provided in this Agreement, each of the Blue Parties, each Signatory Medical Society and each Class Member hereby irrevocably submits to the exclusive jurisdiction and venue of the United States District Court for the Southern District of Florida for any suit, action, proceeding, case, controversy, or dispute relating to this Agreement and/or Exhibits hereto and negotiation, performance or breach of same.

20.2 Parties Shall Not Contest Jurisdiction

In the event of any case, controversy, or dispute arising out of the negotiation of, approval of, performance of, or breach of this Agreement, and solely for purposes of such case, controversy, or dispute, to the fullest extent that they may effectively do so under applicable law, the Parties irrevocably agree that they are and shall be subject to the jurisdiction of the Court, and that the Court is a proper venue and convenient forum.

Furthermore, the parties shall jointly urge the Court to include the provisions of this § 20 in its Final Order and Judgment approving this Agreement.

21. Cooperation

Representative Plaintiffs, the Signatory Medical Societies, Class Counsel and the Blue Parties agree to move that the Court enter an order to the effect that should any Person desire any discovery incident to (or which the Person contends is necessary to) the approval of this Agreement, the Person must first obtain an order from the Court that permits such discovery.

22. Counterparts

This Agreement may be executed in counterparts, each of which shall constitute an original. Facsimile signatures shall be considered valid signatures as of the date hereof, although the original signature pages shall thereafter be appended to this Agreement.

23. Additional Signatory Medical Societies

The Parties agree that, from and after the date of this Agreement, additional medical societies may elect to execute a signature page to this Agreement and thereby agree to be bound by the provisions of this Agreement that are applicable to Signatory Medical Societies. Upon such execution of a signature page, each such additional medical society shall be deemed to be a Signatory Medical Society for all purposes of this Agreement and shall be bound by all of the provisions of this Agreement that are applicable to Signatory Medical Societies.

24. Successors and Assigns

The provisions of this Agreement shall be binding upon and inure to the benefit of each Blue Party and its respective successors and assigns; provided that no Blue Party may assign, delegate or otherwise transfer any of its rights or obligations under this Agreement to a third party that is not a successor or affiliate without the consent of Class Counsel. The provisions of this Agreement shall not apply with respect to any corporation, business, or other entity acquired by a Blue Party after the Preliminary Approval Date, and a Blue Party shall have no obligations under this Agreement with respect to such corporation, business, or entity or the business operations of such corporation, business or entity after the Preliminary Approval Date. The provisions of this Agreement shall not apply with respect to any activities related to FEHBA.

25. Governing Law

This Agreement and all agreements, exhibits, and documents relating to this Agreement shall be construed under the laws of the State of Florida, excluding its choice of law rules.

Contacts for Communications Between Class Counsel and The Blue Parties 26.

At all times during the term of this Agreement, Class Counsel, on the one hand, and the Blue Parties, on the other, shall each designate a representative to facilitate communications between or among the Parties, and provide written notification to the other of the person so designated.

27. Severable Agreement

The provisions of this Agreement are intended to be severable. Should any of the provisions be found illegal, invalid or unenforceable by any court of competent jurisdiction for any reason, it shall be severable from the remainder of this Agreement, and the remainder of this Agreement shall be unchanged and shall be read as if it did not contain the illegal or invalid provision.

Effect of Other Settlements 28.

The Representative Plaintiffs and the Signatory Medical Societies will not reach a settlement with any defendant in this Action which is not a Party which settlement includes a provision containing relief to the Class Members which is also relief to the Class Members set forth in this Agreement if: (i) such relief to the Class Members is applicable to Self-Insured Plans under this Agreement, but is not made applicable to Self-Insured Plans in the proposed settlement agreement; (ii) such relief is made applicable to Class Members who are Non-Participating Physicians under this Agreement, but is not made applicable to Non-Participating Physicians in the proposed settlement agreement, or (iii) such relief to Class Members is applicable to programs offered or sponsored by any state or federal governmental entity other than in its capacity as an employer under this Agreement, but is not made applicable to programs offered or sponsored by any state or federal governmental entity other than in its capacity as an employer in the proposed settlement. The Representative Plaintiffs and Signatory Medical Societies further agree that they will not reach a settlement with any defendant in this Action which is not a Party (or which does not become a Party in the future to this Agreement) which settlement imposes a less burdensome obligation on such other defendant regarding the obligations set forth in any of § 7.18, § 7.19 or § 7.20, unless such less burdensome obligation is a result of good faith negotiation between such defendant and the Medical Society in the state where it does business and is an exception similar to those exceptions set forth in Exhibit H. A settlement agreement involving only a financial payment obligation and no prospective relief shall be construed as a less burdensome obligation under this provision.

DELIVERED on April 27, 2007.

EXHIBIT A

CLAIM FORM FOR THE SETTLING BLUE CROSS AND BLUE SHIELD PARTIES' SETTLEMENT FUND AND ELECTION OF CONTRIBUTION TO CHARITABLE ORGANIZATION

You must read the Notice of Proposed Settlement and Claim Form Instructions before completing this Claim Form. The capitalized terms used in this Claim Form are defined in the Settlement Agreement. A Class Member may file only one Claim Form.

SECTION A: CLAIMANT INFORMATION - ALL CLAIMANTS MUST COMPLETE THIS SECTION (EITHER THROUGH A

	ganization	Please indicate the	number of Physician	ns on your list
Physician Group/Or Physician Group or Organi	-	Flease moleate the	number of Enysicial	is on your list
Name and Title of Employee/Representative Filing		Ph	Phone	
claims, along with each Active Phys	n the information that is the	on specified in the Cla the Physician Group/ that is attached to this	im Form Instruction Organization is sub Claim Form, or, al	-
Name of Physician				
Name of Representative (if Physician is Deceased)*	·			Phone
•	r an estate to co	onfirm your status. The o	Tax I.D. requested ir dividual Claiman	th documentation such as a death certificate n Section E is that of the heir or estate.
City	State Zip Code Blue Cross/Blue Shield Provider Number (if applicable)			Shield Provider Number (if applicable)
ndividual Claimants: please a member.	check the appro	ppriate box in SECTION	B or SECTION C to	o indicate the category of which you are
INACTIVE SUBSEQUENT T CLASS MEMBER. ☐ By checking the bo and that I am either a Class	X to the left, I come Member (as de	ertify that I have reviewed escribed in the enclosed error become inactive su	L HEIR OR REPRES ed the enclosed Noti d Notice of Proposed	SENTATIVE OF A DECEASED ice of Proposed Settlement and Claim Form Instruction d Settlement and defined in the Settlement Agreement, 1999 (a "Retired Physician"), or that I am the legal has
wno nas retired from the brai			usequent to May 22	:, 1999 (a "Ketired Physician"), or that I am the lega

ACTIVE PHYSICIANS: Individual Physicians must check ONLY ONE of the boxes below to designate the range of Gross Receipts that are the basis of this claim. Physician Groups and Physician Organizations must attach a list that designates the range of Gross Receipts for each Active Physician for whom you are filing this claim (by using the attached rider, or, alternatively a form that is substantially similar to the one that is attached) Physician Groups and Physician Organizations do not check any boxes below.

•	this box, I certify that I rece to the settling Blue Plans' P		-		
Members during the three ca	this box, I certify that my of alendar year period of 2004, this box, I certify that my of alendar year period of 2004, this box, I certify that my of	2005, and 2006 were Gross Receipts for pi 2005, and 2006 were	at least \$ 5,000 b roviding Covered \$ 50,000 or great	out less than \$ 50,000. Services to the settlinger.	ng Blue Plans' Plan
-	nsecutive three - year period orted by the enclosed docum		-	nber 31, 2006 were in t	he amount shown in
• • • • • • • • • • • • • • • • • • • •	ments from a Blue Cross	-	,	NOT listed in the Not	tice, please check
Box I above.	or III in Section C, please	e move to Section I)		
ii you checked boxes i, ii	or in the decision of pieus.	e move to decitor i	~• ,		
the basis of your claim and received for providing Covernment	Section C, please indicated the check the appropriate ered Services to the settling ou attached in the box below.	box to indicate for g Blue Plans' Plan M	this three-year p	period the range of G	Pross Receipts you
	☐ under \$5,000	\$ 5,000 - \$49,	999	□ \$ 50,000 or ove	r
	Description of the Proof Atta	ached.			
Dates of the Three-Year Period					
SECTION D: INSTRUCTIONS CHECK ONE:	S FOR PAYMENT - ALL	CLAIMANTS MUS	T COMPLETE	THIS SECTION.	-
☐ By checking this box, the Settlement Fund for an elig be an individual Physician or Ph		e., to the Class Mei			
☐ By checking this box, Settlement Fund for an eligible Claim Form Instructions (select		anization that I have			
CLEARLY print the number you are selecting from the Instructions in the box to the	e list on page 3 of th		Charitable Orga	nization Number	
SECTION E: SUBSTITUTE V	N-9 - ALL CLAIMANTS N	MUST COMPLETE	THIS SECTION	l.	
On the appropriate line, enter the name will appear on any check For Physician Groups and Phy	and related Form-1099. F	or individuals, this is	your Social Sec	curity Number (SSN).	
	OR		_		
Social Security Nu		Employer Identi	fication Number	(EIN)	

By signing this Claim Form, I certify that:

The number shown on this form above is the correct Social Security Number or Employer Identification Number for this claimant; and

Document 928-7

The claimant is not subject to backup withholding because the claimant: (a) is exempt from 2. backup withholding, or (b) has not been notified by the Internal Revenue Service (IRS) that the claimant is subject to backup withholding as a result of failure to report all interest or dividends, or (c) the IRS has notified the claimant that the claimant is no longer subject to backup withholding.

Backup withholding is extra tax withholding that occurs when a taxpayer has underreported interest or dividends in a previous year. The IRS notifies taxpayers who are subject to backup withholding. If you (the claimant) have been notified by the IRS that you are subject to backup withholding because you have failed to report all interest and dividends on your tax return, you must cross out item 2 above by placing a line through the section. The IRS does not require your consent to any provision of this document other than the certifications in Section E required to avoid backup withholding.

SECTION F: CERTIFICATION - ALL CLAIMANTS MUST COMPLETE THIS SECTION.

I do declare and certify, under penalties of perjury, as follows:

- I am a Class Member, a legal heir or representative of a deceased Class Member, or an authorized representative of the Physician Group or Physician Organization identified above;
- I am not submitting a claim on behalf of any Class Members who have submitted a request to Opt-Out of the Class and Settlement:
- I am not submitting a claim on behalf of any Active Physicians who are, on their own behalf, submitting separate claims based on the same Covered Services;
- I am not submitting a claim on behalf of any Active Physicians against whom a Blue Plan has obtained a final finding of fraud and/or abuse from a judicial, arbitral, or administrative proceeding and a corresponding final judgment for damages arising from a claim (or claims) for payment during the same time period for which the claim is asserted; and
- All of the statements and information provided in this Claim Form are true, correct and complete, to the best of my knowledge.
- The IRS does not require your consent to any provision of this document other than the certifications in Section E required to avoid backup withholding.

Signature	Date	

Claims should be sent to the Settlement Administrator at:

[ADDRESS]

YOU MUST COMPLETE AND SIGN THIS CLAIM FORM, AND THE ENVELOPE RETURNING YOUR CLAIM FORM MUST BE MAILED TO THE SETTLEMENT ADMINISTRATOR WITH A POSTMARK DATE NO LATER THAN [90 DAYS AFTER NOTICE DATE].

IF YOUR SIGNED CLAIM FORM IS NOT MAILED TO THE SETTLEMENT ADMINISTRATOR BY THIS DEADLINE, YOU WILL BE DEEMED TO HAVE WAIVED YOUR RIGHT TO RECEIVE ANY PAYMENT FROM THE SETTLEMENT FUND.

WE STRONGLY RECOMMEND SENDING YOUR CLAIM FORM VIA REGISTERED OR CERTIFIED MAIL AND RETAINING YOUR RECEIPT AND A COPY OF YOUR CLAIM FORM FOR YOUR RECORDS.

If you have any questions, please call the Settlement Administrator at [PHONE]

RIDER FOR PHYSICIAN GROUPS/ORGANIZATIONS THAT ARE FILING CLAIMS ON BEHALF OF ACTIVE PHYSICIANS (PURSUANT TO SECTION A OF THE CLAIM FORM)

Physician Group/Organization

Physician Group or Organization Name	
Name and Title of Employee/Representative Filing	Phone

List of Active Physician Names and Key Information, For $\underline{\sf Each}$ Active Physician for Whom You Are Submitting Claims:*

Physician Name	Physician Type (MD or DO)	Last four digits of Social Security No.	Range of Gross Receipts (e.g., "Under \$5,000" or "Over \$50,000")

^{*} Please attach additional pages of this form, if necessary.

EXHIBIT B

INSTRUCTIONS REGARDING THE CLAIM FORM FOR THE BLUE CROSS AND BLUE SHIELD PARTIES' SETTLEMENT FUNDAND ELECTION OF CONTRIBUTION TO CHARITABLE ORGANIZATION

It is very important that you read the enclosed Notice of Proposed Settlement in order to fully understand your rights under this Settlement.

DEADLINE FOR CLAIM FORM SUBMISSION: Postmarked by

WE STRONGLY RECOMMEND SENDING YOUR CLAIM FORM VIA REGISTERED OR CERTIFIED MAIL AND RETAINING YOUR RECEIPT AND A COPY OF YOUR CLAIM FORM FOR YOUR RECORDS.

If you validly submit a Claim Form to the Settlement Administrator postmarked no later than ______, you may elect to receive the portion of the Settlement Fund to which you are entitled or you may direct that your portion be paid to one of the charitable organizations listed on the bottom of Page 2 of this form (on the reverse of this page). A Class Member may file only one Claim Form.

If you would like your portion of the Settlement Fund to be donated to a charitable organization, you may do so by selecting from the List of charitable organizations found on the bottom of Page 2. (See Section D on the Claim Form to make your payment election.) The Class Member may only select one charitable organization.

By submitting a Claim Form you are agreeing to be subject to the jurisdiction of the United States District Court for the Southern District of Florida for any proceedings relating to your Claim Form. Capitalized terms used in the Instructions and Claim Form that are not otherwise defined herein are defined in the Settlement Agreement. A copy of the Settlement Agreement can be found on:

[Settlement Administrator's website] <u>www.hmosettlements.com; www.WhatleyDrake.com; www.ArchieLamb.com;</u> and www.kttlaw.com

"Active Physician" is defined in the Settlement Agreement as a "Class Member who is a Physician and who is not a Retired Physician as of the Preliminary Approval Date," i.e., as of [Preliminary Approval Date]. "Retired Physician" is defined in the Settlement Agreement as a "Class Member who subsequent to May 22, 1999, has become an inactive Physician, has retired from the practice of, or has otherwise ceased to practice, medicine or has died as of the date of Preliminary Approval," i.e. as of [Preliminary Approval Date]. The "Class" and "Class Members" are defined in the Settlement Agreement and described in the enclosed Notice of Proposed Settlement.

Mail your completed Claim Form, with any required documentation, to the Settlement Administrator at:

Settlement Administrator

NOTE: YOU MUST NOTIFY THE SETTLEMENT ADMINISTRATOR <u>IMMEDIATELY</u> OF ANY CHANGE IN YOUR ADDRESS, TO AVOID HAVING YOUR MAILED CHECK RETURNED TO THE SETTLEMENT ADMINISTRATOR AND REMITTED TO A CHARITABLE ORGANIZATION, PURSUANT TO SECTION 8.4 OF THE SETTLEMENT AGREEMENT.

SECTION-BY-SECTION INSTRUCTIONS:

Section A: ALL CLAIMANTS MUST COMPLETE THIS SECTION.

PHYSICIAN GROUPS OR ORGANIZATIONS: If you are representing a Physician Group or Physician Organization, in Section A, please write in the Group or Organization name, the name of the person completing the Claim Form, and attach a list of all the Active Physicians for whom you are filing this Claim. Physician Groups and Physician Organizations may submit Claim Forms on behalf of Physicians employed by or working with them without providing individual signatures from the individual Physicians, if authorized to do so by the individual Physicians and if the individual Physicians do not also submit Claim Forms on their own behalf. Your list of Active Physicians should be set forth on the rider attached to the Claim Form, or on a document substantially similar to that which is attached, and must include all of the following information for each Active Physician:

1. Physician name

2. Physician Type (example: MD or DO)

3. Last four digits of each Active Physician's Social Security Number (SSN)

4. Range of Gross Receipts received from all settling Blue Plans (for a list of which, please see Exhibit 1 to the Notice) for each Active Physician during the three-year period from 2004 through 2006: Range I: Under \$5,000; Range II: \$5,000 to \$49,999; Range III: \$50,000 or over Case 1:03-cv-21296-FAM

Document 928-8 Entered on FLSD Docket 04/27/2007 Page 2 of 5 if Active Physician is submitting records for any other consecutive three-year period within the time frame from January 1, 1997 through December 31, 2006, indicate the period, the Range of Gross Receipts, and the form of proof submitted.

Please do not forget to fill out Sections D, E and F all of which are labeled: "All Claimants Must Complete This Section."

Section B: ONLY RETIRED PHYSICIANS MUST COMPLETE THIS SECTION: Individual Class Members (as described in the enclosed Notice of Proposed Settlement) who are Retired Physicians are entitled to receive a *pro rata* amount of the portion of the Settlement Fund that is reserved for retired or deceased Physicians. This amount is determined by multiplying the Settlement Fund by two times the quotient derived by dividing the number of Retired Physicians who file valid Claim Forms by the total number of Class Members.

Section C: ONLY ACTIVE PHYSICIANS MUST COMPLETE THIS SECTION: Individual Class Members (as described in the enclosed Notice of Proposed Settlement) who are Active Physicians are entitled to receive a *pro rata* amount of the portion of the Settlement Fund that is not reserved for retired or deceased Physicians. Your settlement payment will be based upon the amount of payments received by you from settling Blue Plans in payment for services during the three-year period from 2004 to 2006 (or for any consecutive three-year period from 1997 through 2006 if you elect to submit payment records). For purposes of determining which box to check in this section, "Blue Plan" means any of the settling Blue Plans listed on Page 2 of the Notice that was sent along with this Instruction form. You should check ONE of the boxes in Section C, as described below:

Box I: Active Physicians who received no payments from any settling Blue Plan (for a list of which, please see Exhibit 1 to the Notice) or payments from settling Blue Plans of less than \$5,000 during the three-year period from 2004 through 2006should check Box I, and will receive a settlement payment that is equal to the "Base Amount" of the Settlement Fund that is being paid to Active Physicians in the settlement.

Box II: Active Physicians who received payments from settling Blue Pians of \$5,000 or more, and less than \$50,000, during the three year period from 2004 through 2006should check **Box II**, and will receive a settlement payment that is equal to five times the "Base Amount."

Box III: Active Physicians who received payments from settling Blue Plans of \$50,000 or more during the three-year period from 2004 through 2006 should check **Box III**, and will receive a settlement payment that is equal to ten times the "Base Amount."

To simplify the process of obtaining payment from the Settlement Fund, Active Physicians may check off one of the boxes described above in Section C and submit their Claim Forms to the Settlement Administrator without any additional documentation. The amount of the Settlement Fund that each Active Physician is entitled to receive shall be determined based upon the Active Physician's certification as to the amount of his or her Gross Receipts from the settling Blue Plans during the period 2004 through 2006.

Box IV: Alternatively, Active Physicians may elect to submit to the Settlement Administrator <u>proof</u> of their payments from settling Blue Plans, in the form of Form-1099s or other forms of proof, to show the amounts of payments received from settling Blue Plans during any other consecutive three-year period falling within the time frame from January 1, 1997 through December 31, 2006, to establish which of the three Active Physician payment levels is due to such Active Physician from the Settlement Fund. Active Physicians electing to do so should check Box IV and then complete the table indicating the period for which the documentation is being submitted, the range of payments received during that period, and the type of proof being submitted. Active Physicians who have been paid through Physician Organizations or Physician Groups (including without limitation Delegated Entities) may submit to the Settlement Administrator proof (examples include IRS form 1099, Explanations of Benefits or EOBs, or other suitable accounting records) of the amounts received during any other consecutive three-year period falling within the time frame from January 1, 1997 through December 31, 2006 for providing services to members of plans offered or administered by settling Blue Plans. Please note that proof submitted will be used exclusively to assign claimants to one of the three compensation categories for Active Physicians in the Settlement Agreement (e.g., "Under \$5,000," "Over \$50,000"); compensation will not be based on the exact dollar value of the receipts.

The Settlement Administrator will make the final decision on any dispute regarding the eligibility of a Class Member to receive payment from the Settlement Fund or the amount of any such payment.

In determining your Gross Receipts, you should include amounts paid by settling Blue Plans directly or by intermediaries for providing Covered Services to Blue Plan Plan Members. For example, you may have provided services to Blue Plan Plan Members through an intermediary that contracted with a Blue Plan to provide the services, for example, an IPA, medical group, organized delivery system, physician hospital organization, etc. In determining your Gross Receipts for providing Covered Services to Blue Plan Plan Members, you should also include amounts you received from such intermediaries for treating Blue Plan Plan Members.

Active Physicians against whom a Blue Plan has obtained a final finding of fraud and/or abuse (as that term is defined in Section 8.3(h) of the Settlement Agreement) from a judicial, arbitral, or administrative proceeding and a corresponding final judgment for damages arising from a claim (or claims) for payment for the same time period for which a claim may be asserted are not entitled to payment from the Settlement Fund regardless of their status as Class Members.

Any questions about this procedure or proof that will be accepted should be addressed to the Settlement Administrator at:

Section D: ALL CLAIMANTS (PHYSICIAN GROUPS, PHYSICIAN ORGANIZATIONS AND INDIVIDUALS) MUST COMPLETE THIS SECTION: This Section will dictate to whom the payment is addressed. If you submit a Claim Form on behalf of a Physician Group or Physician Organization and do not elect to donate the settlement award to a charitable organization, the payment will be made to the Physician Group or Physician Organization for distribution by the Physician Group or Physician Organization to individual Physicians.

SEARS 1:03-CV-C211/206-F-CMOUPD AND INDIVIDUALS) MUST ETEMPLET ELAND SECTION. 04/27/2007	Page 3 of 5
Section F: ALL CLAIMANTS (GROUPS AND INDIVIDUALS) MUST COMPLETE THIS SECTION.	
IF YOU HAVE QUESTIONS ABOUT THE SETTLEMENT FUND, THE CHARITABLE ORGANIZATIONS, PROCEDURE FOR FILING A CLAIM FORM, CONTACT THE SETTLEMENT ADMINISTRATOR ATCOUNSEL AT	
DO NOT CONTACT THE COURT OR ANY OF THE SETTLING BLUE PLANS WITH OUESTIONS ABOUT THE SE	TTI EMENT

List of Charitable Organizations

National Charitable Organization

1. Physicians' Foundation for Health Systems Innovations

State Charitable Organizations

Alabama

2. Alabama Department of Public Health

California

3. California Medical Association Foundation

Colorado

4. El Paso County Medical Society Foundation

Connecticut

CSMS Physicians' Health and Education Fund (Connecticut State Medical Society)

Florida

- 6. Florida Medical Foundation (Florida Medical Association)
- 7. Florida Academy of Family Physicians Foundation

Hawaii

Hawaii Medical Foundation 8.

Illinois

Illinois State Medical Society Educational and Scientific Foundation

lowa

10. The University of Iowa Foundation (Medical Center for Development)

Georgia

11. Medical Association of Georgia Institute for Excellence in Medicine, Inc.

Louisiana

12. Louisiana State Medical Society Educational and Research Foundation

Massachusetts

- 13. BCBSMA Foundation
- 14. MMS Alliance Charitable Foundation
- 15. Community Health Center Capital Fund

Michigan

16. The Michigan Health & Safety Coalition

17. Minnesota Physicians Foundation (Minnesota Medical Association)

Nebraska

18. Nebraska Medical Foundation (Nebraska Medical Association)

New Hampshire

19. John P. Bowler, M.D., Memorial Library (New Hampshire Medical Society)

New Jersey

20. Institute of Medicine and Public Health of New Jersey, Inc. (Medical Society of New Jersey)

New Mexico

- 21. Physician Outreach Program (University of New Mexico Health Sciences Center)
- 22. Carrie Tingley Foundation (Carrie Tingley Hospital)

New York

23. Medical, Educational and Scientific Foundation of New York, Inc. (Medical Society of the State of New York)

North Carolina

24. North Carolina Medical Society Foundation, Community Practitioners Program

Northern Virginia

25. Medical Society of Northern Virginia Foundation

Oklahoma

26. Oklahoma Medical Research Foundation

27. Oregon Medical Education Foundation

Pennsylvania

- 28. The Foundation of the Pennsylvania Medical Society
- 29. Commission on Economic Opportunity (Free Health Clinics)
- 30. Moses Taylor Hospital (Patient Care for Underinsured/Uninsured)
- 31. Susquehanna Health System (Community Based Health Clinics for Underinsured/Uninsured)
- 32. Pocono Health System (Community Based Health Clinics for Underinsured/Uninsured)

Puerto Rico

33. Fundación del Colegio de Médicos y Cirujans de Puerto Rico

Rhode Island

- 34. Rhode Island Community Health Center Association
- 35. Rhode Island Free Clinic
- 36. Rhode Island Medical Society Foundation

South Carolina

37. South Carolina Medical Association Foundation

South Dakota

- 38. The University of South Dakota Foundation (Medical School Program)
- 39. The University of South Dakota Foundation (Lee Medical Building)

Tennessee

- 40. Tennessee Medical Foundation (Tennessee Medical Association)
- 41. Tennessee Foundation for Quality Patient Healthcare

Texas

42. Texas Medical Association Foundation

Utah

- 43. Utah Medical Association Foundation
- 44. The Fourth Street Clinic

Vermont

45. Vermont Medical Society Education and Research Foundation, Inc.

Virginia

46. Arlington County Medical Society Foundation

Washington

47. Washington State Medical Education and Research Foundation

EXHIBIT C

UNITED STATES DISTRICT COURT FOR THE SOUTHERN DISTRICT OF FLORIDA Miami Division

RICK LOVE, M.D., et al.,)
, , ,	j
Plaintiffs	í
- A) Case No. 03-21296-CIV-
) MORENO/SIMONTON
v.)
**)
BLUE CROSS AND BLUE SHIELD)
ASSOCIATION, et al.,)
)
Defendants)
)
	

COMPLIANCE DISPUTE FORM FOR SETTLEMENT AMONG BLUE PARTIES, PHYSICIANS, PHYSICIAN GROUPS, AND PHYSICIAN ORGANIZATIONS

All capitalized terms used in this form are defined in the Settlement Agreement.

Name
Name of Entity:
Address:
Tax Identification Number:
Blue Cross/Blue Shield Provider Number:(if applicable)
E-mail Address:
Telephone Number:
Signature:
Date:

behalf.

Check	one of the following:
	I am a Class Member bringing this Compliance Dispute on my own behalf.
	I am a Class Member and hereby authorize the following Signatory Medical Society to bring this Compliance Dispute on my behalf:
	It is a Signatory Medical Society authorized to bring this Compliance Dispute on its own

Document 928-9

Set forth in detail below, using particularized facts and dates of occurrence, the specific obligation(s) of the Blue Party owing to you under Section 7 of the Settlement Agreement that you allege the Blue Party has materially failed to perform. Describe how you have been adversely affected by the Blue Party's alleged failure to comply with those specific obligation(s). You may attach supporting documentation or affidavit testimony.

You must complete and submit this Compliance Dispute Form no later than ninety (90) days after the Compliance Dispute first arose or after the Compliance Dispute reasonably could have been known to you, whichever is later, to:

Class Compliance Dispute Facilitator

	15102.00 15102.00	ansara Karaty						92381	11120
				100	Sec. 15.		200		
fatotalist Dearge			Acesty Valida Singal			\$2.5 1825			

EXHIBIT D1

IN THE UNITED STATES DISTRICT COURT FOR THE SOUTHERN DISTRICT OF FLORIDA Miami Division

RICK LOVE, M.D., et al., Plaintiffs,) Case No. 03-21296- CIV-MORENO/SIMONTON
V.)
BLUE CROSS & BLUE SHIELD ASSOCIATION, et al., Defendants.)))

ORDER APPROVING SETTLEMENT AMONG CERTAIN BLUE PARTIES AND PHYSICIANS, PHYSICIAN GROUPS, AND PHYSICIAN ORGANIZATONS, CERTIFYING CLASS, AND DIRECTING ENTRY OF FINAL JUDGMENT

The Court having reviewed and considered the Joint Motion for Final Approval of Settlement Concerning Claims Against Certain Blue Parties dated ______, 2007 in the action styled Love, et al. v. Blue Cross Blue Shield Ass'n, et al., (formerly Thomas, et al. v. Blue Cross Blue Shield Ass'n, et al.), Case No. 03-21296-CIV-MORENO/SIMONTON (the "Action"), and having reviewed and considered the terms and conditions of the proposed settlement (the "Settlement") as set forth in the settlement agreement dated April 27, 2007(the "Settlement Agreement"), 1/a copy of which has been submitted to the Court, and having reviewed and considered the applications of Class Counsel for an award of attorneys' fees and expenses and for an award of fees to Representative Plaintiffs and Representative Plaintiffs in Other Actions, and the Court having held a Settlement Hearing after being satisfied that notice to the Class had been provided in accordance with the Court's Order Preliminarily Approving Proposed Settlement

Capitalized terms used in this Order that are not otherwise defined herein have the meaning assigned to them in the Settlement Agreement.

Page 2 of 19

- The Court has jurisdiction over the subject matter of the Action pursuant to A. 28 U.S.C. §§ 1331 and 1367, and all acts within the Action, and over all the parties to the Action, and all Class Members.
- The Class conditionally certified in the Preliminary Approval Order has В. been appropriately certified, for settlement purposes, under Rules 23(a), 23(b)(2), and 23(b)(3) of the Federal Rules of Civil Procedure.
- Certification for settlement purposes is appropriate under Rule 23(a) 1. because: (i) the Class is so numerous that joinder of all Class Members is impracticable; (ii) there are questions of law or fact common to the Class; (iii) the claims or defenses of the Representative Plaintiffs are typical of the claims of the Class; and (iv) the Representative Plaintiffs will fairly and adequately protect the interests of the Class.
- Certification for settlement purposes is appropriate under Rule 2. 23(b)(2) because the Blue Parties have acted on grounds generally applicable to the Class, thereby making appropriate final injunctive relief with respect to the Class as a whole.
- Certification for settlement purposes is appropriate under Rule 3. 23(b)(3) because questions of law or fact common to all Class Members predominate over any

questions affecting only individual Class Members, and a class action is superior to other available methods for the fair and efficient adjudication of this controversy. In making this determination, the Court has considered the interest of the Class Members in individually controlling the prosecution of separate actions and the extent and nature of any litigation concerning this controversy already commenced by Class Members. The Court notes that because certification of the Class is in connection with the Settlement rather than litigation, it need not consider the desirability of concentrating the litigation of the claims in this particular forum or the manageability presented by certification of the nationwide class proposed in the Complaint in the Action. See Strube v. American Equity Investment Life Ins. Co., 226 F.R.D. 688 (M.D. Fla. 2005) (citing Amchem Prods., Inc. v. Windsor, 521 U.S. 591, 620 (1997)).

- C. Class Counsel has fairly and adequately represented the Class for purposes of entering into and implementing the Settlement.
- D. Notice to the putative Class Members and other potentially interested Persons has been provided in accordance with the notice requirements specified by the Court in the Preliminary Approval Order. Such notice: (i) constituted the best notice to putative Class Members that was practicable under the circumstances; (ii) constituted notice that was reasonably calculated, under the circumstances, to apprise putative Class Members of the pendency and nature of the Action, their right to object and to appear at the Settlement Hearing or to exclude themselves from the Class and the Settlement, and the binding effect of a class judgment; (iii) was reasonable and constituted due, adequate, and sufficient notice to Persons entitled to be provided with notice; and (iv) fully complied with the requirements of due process and the Federal Rules of Civil Procedure.

Page 4 of 19

- E. The Court has held a hearing to consider the fairness, reasonableness, and adequacy of the Settlement, has been advised of all objections to the Settlement and has given fair consideration to such objections.
- F. The Settlement is the product of good faith, arm's length negotiations between the Representative Plaintiffs and the Signatory Medical Societies and their counsel, on one hand, and the Blue Parties and their counsel, on the other hand.
- G. The Settlement, as provided for in the Settlement Agreement, is in all respects fair, reasonable, adequate, is not the product of collusion between the Parties, and is otherwise proper and in the best interests of the Class. In reaching this conclusion, the Court has considered a number of factors, including: (i) an assessment of the likelihood that the Representative Plaintiffs and/or the Class would prevail at trial; (ii) the range of possible recovery available to such plaintiffs as a result of such a trial; (iii) the consideration provided to the Class Members pursuant to the Settlement, as compared to the range of possible recovery discounted for the inherent risks of litigation; (iv) the complexity, expense, and possible duration of such litigation in the absence of a settlement; (v) the nature and extent of any objections to the Settlement; and (vi) the stage of proceedings at which the Settlement was reached. See Sterling v. Stewart, 158 F.3d 1199, 1203 n.6 (11th Cir. 1998) (citing Bennett v. Behring Corp., 737 F.2d 982, 986 (11th Cir. 1984)).
- H. A list of those putative Class Members who or which have timely and properly elected to Opt-Out of the Class and the Settlement and who therefore are not bound by the Settlement, the provisions of the Settlement Agreement, this Order, and the Judgment to be entered by the Clerk of the Court hereon, has been submitted to the Court as an exhibit to the Declaration of [sworn to on [], 2007. A copy of such exhibit is

Page 5 of 19

attached as Exhibit hereto. All Class Members (as permanently certified below) shall be subject to all of the provisions of the Settlement, the Settlement Agreement, this Order, and the Judgment to be entered by the Clerk of the Court.

- I. The bar order provisions of this Order, which prohibit the assertion of certain claims against the Blue Parties and other Released Parties, as set forth below, are a condition of the Settlement and a significant component of the consideration afforded to the Blue Parties in the Settlement, and these provisions are reasonable under the circumstances.
- J. The dismissal with prejudice and entry of Judgment contemplated by the Settlement and this Order will dispose of fewer than all of the claims at issue, or parties to, this Action. The Court finds that there is no just reason for delay in entering judgment in the form attached hereto (the "Judgment") dismissing the Action with prejudice as to the Blue Parties and that entry of the Judgment to that effect, as directed below, is warranted under Rule 54(b) of the Federal Rules of Civil Procedure.

On the basis of the foregoing findings and the submissions and proceedings referred to above, NOW THEREFORE, IT IS HEREBY ORDERED ADJUDGED AND DECREED:

Certification of the Class and Approval of Settlement

- 1. The Settlement and the Settlement Agreement are hereby approved as fair, reasonable, adequate, and in the best interests of the Class, and the requirements of due process and Rule 23 of the Federal Rules of Civil Procedure have been satisfied. The objections to the Settlement and the Settlement Agreement are overruled and denied in all respects.
- The Court having found that each of the elements of Rules 23(a), 23(b)(2), and 23(b)(3) of the Federal Rules of Civil Procedure are satisfied, for purposes of settlement only, pursuant to Federal Rules of Civil Procedure 23(a), 23(b)(2), and 23(b)(3), solely with respect to the Blue Parties, as well as the other Released Parties, the Action is permanently certified as a class

action on behalf of the following Persons (collectively, the "Class" and each member of the Class a "Class Member"):

> any and all Physicians, Physician Groups and Physician Organizations who provided Covered Services to any Plan Member or services to any individual enrolled in or covered by a Plan offered or administered by any Person named as a defendant in the Complaint or by any other primary licensee of the BCBSA or by any of their respective current or former subsidiaries or Affiliates, in each case from May 22, 1999 through the [Preliminary Approval Date]. The Class shall exclude: (i) all Persons who, in accordance with the terms of this Agreement, execute a timely request for exclusion (Opt-Out) from the Class; and (ii) the Blue Parties, their Affiliates and any of their officers, directors and employees.

The putative Class Members identified on the list submitted to the Court (and attached hereto as an exhibit) as having timely and properly elected to Opt-Out of the Class and the Settlement are hereby excluded from the Class and shall not be entitled to any of the monetary or other benefits afforded to the Class under the Settlement Agreement.

- For purposes of the Settlement only, Representative Plaintiffs are appointed 3. as representatives of the Class and Class Counsel are appointed and designated as counsel to the Class.
- Notwithstanding the certification of the foregoing Class, the appointment of 4. Representative Plaintiffs as representatives of the Class, and the appointment and designation of Class Counsel as counsel for the Class for purposes of effecting the Settlement, if this Order is reversed on appeal or the Settlement Agreement is terminated or is not consummated for any reason, the foregoing certification of the Class, appointment of the Representative Plaintiffs as representatives for the Class, and appointment and designation of Class Counsel as counsel for the Class shall be void and of no further effect and the Parties to the Settlement and/or any Blue Party

terminating its participation in the Settlement Agreement in accordance with its terms, shall be returned to the status each occupied before entry of this Order, without prejudice to any legal argument that any of the Parties to the Settlement Agreement might have asserted but for the Settlement Agreement.

Release and Injunctions Against Released Claims

5. Upon the Effective Date, the "Released Parties," which shall include the Blue Parties and each of their present and former parents, divisions and Affiliates and each of their respective current or former officers, directors, employees, agents, insurers and attorneys (and the predecessors, heirs, executors, administrators, legal representatives, successors and assigns of each of the foregoing), and all Persons who provided claims processing services, software, proprietary guidelines or technology to the Blue Parties, their subsidiaries or Affiliates, and those contracted agents processing claims on their behalf (including, without limitation, NASCO), together with each such Person's predecessors or successors (but only to the extent of such Person's services and work was done pursuant to contract with the Blue Parties or their subsidiaries or Affiliates), but excluding all Delegated Entities, shall be released and forever discharged by the Signatory Medical Societies and all Class Members, and by their respective current and former officers, directors, employees, attorneys, heirs, executors, administrators, agents, legal representatives, professional corporations, partnerships, assigns and successors, but only to the extent claims are derived by contract or operation of law from the claims of Class Members (collectively, the "Releasing Parties"), from any and all causes of action, judgments, liens, indebtedness, costs, damages, obligations, attorneys' fees, losses, claims, liabilities, and demands of whatever kind, source, or character whether arising under any federal or state law, which (consistent with the Parties' understanding of the settlements in Shane) includes, but is not

limited to, Racketeer Influenced and Corrupt Organizations Act, antitrust and other statutory and common law claims, intentional or nonintentional, (each a "Claim"), arising on or before the Effective Date, that are, were, or could have been asserted against any of the Released Parties by reason of, arising out of, or in any way related to any of the facts, acts, events, transactions, occurrences, courses of conduct, business practices, representations, omissions, circumstances, or other matters referenced in the Action, or addressed in the Settlement Agreement, whether any such Claim was or could have been asserted by any Releasing Party on its own behalf or on behalf of other Persons, or to the business practices that are the subject of Section 7 of the Settlement Agreement. This includes, without limitation and as to Released Parties only, any aspect of any fee for service claim submitted by any Class Member to the Blue Plans, and any claims of any Class Member related to or based upon any Capitation agreement between the Blue Plans and any Class Member or other Person or entity, or the delay, nonpayment, or amount of any Capitation payments by the Blue Plans, and any allegation the Blue Parties have conspired with, aided and abetted, or otherwise acted in concert with other managed care organizations, other health insurance companies, Delegated Entities, and/or other third parties with regard to any of the facts, acts, events, transactions, occurrences, courses of conduct, business practices, representations, omissions, circumstances, or other matters related to the Action, or with regard to the Blue Parties' liability for any other demands for payment submitted by any Class Member to such other managed care organizations, health insurance companies, Delegated Entities, and/or other third parties.

6. Except for Claims for damages against Defendants in the Action that are not Parties, relating to providing Covered Services to Blue Plans' Plan Members in connection with the BlueCard Program and other similar national account delivery programs governed by BCBSA

Page 9 of 19

(including but not limited to NASCO-to-NASCO arrangements), the Releasing Parties are deemed to have forever abandoned and discharged any and all Claims that exist now or that might arise in the future against any other Persons which Claims arise from, or are based on, conduct by any of the Released Parties that occurred on or before the Effective Date and are, or could have been, alleged in the Complaint, whether any such Claim was or could have been asserted by any Releasing Party on its own behalf or on behalf of other Persons. This Order does not relieve any Person or entity that is not a Released Party from responsibility for its own conduct or conduct of other Persons who are not Released Parties, or preclude any Representative Plaintiff from introducing any competent and admissible evidence to the extent consistent with Sections 14 and 16 of the Settlement Agreement.

- 7. Any Claims released or discharged pursuant to Sections 13.1(a) and 13.1(b) of the Settlement Agreement, subject to the exception regarding Retained Claims contained in Section 13.6 of the Settlement Agreement, shall be referred to collectively as "Released Claims."
- 8. Notwithstanding the foregoing, the Releasing Parties are not deemed to have released claims for payment (each a "Retained Claim" and, collectively, the "Retained Claims") for Covered Services provided to Plan Members prior to or on the Effective Date as to which, as of the Effective Date: (i) no claim with respect to such Covered Services has been submitted to a Blue Plan, provided that the applicable period for filing such claim has not elapsed; or (ii) a claim with respect to such Covered Services has been filed with a Blue Plan but such claim has not been finally adjudicated by the Blue Plan. For purposes of clause (ii), above, final adjudication shall mean completion of the Blue Plan's internal appeals process. In the event that a claim referred to in clause (ii) is finally adjudicated less than thirty (30) days prior to the Effective Date, such claim shall constitute a Retained Claim if a Physician seeks relief under Section 7.10 of

the Settlement Agreement not later than ninety (90) days after notice of such final adjudication, but otherwise such claim shall constitute a Released Claim. Disputes relating to Retained Claims shall be resolved exclusively pursuant to the appropriate remedial provisions of the Settlement Agreement.

- 9. The Releasing Parties are permanently enjoined from: (i) filing, commencing, prosecuting, intervening in, participating in (as class members or otherwise), or receiving any benefits from any lawsuit, arbitration, administrative or regulatory proceeding, or order in any jurisdiction based on any or all Released Claims against one or more Released Parties; (ii) instituting, organizing class members in, joining with class members in, amending a pleading in, or soliciting the participation of class members in, any action or arbitration, including but not limited to a purported class action, in any jurisdiction against one or more Released Parties based on, involving, or incorporating, directly or indirectly, any or all Released Claims; and (iii) filing, commencing, prosecuting, intervening in, participating in (as class members or otherwise), or receiving any benefits from any lawsuit, arbitration, administrative or regulatory proceeding, or order in any jurisdiction based on an allegation that an action of the Blue Parties, which is in compliance with the provisions of the Settlement Agreement, violates any legal right of any Class Member.
- 10. Upon the Effective Date and through the Termination Date, the Releasing Parties and each of them shall be deemed to have agreed and covenanted not to sue or prosecute, institute or cooperate in the institution, commencement, filing, or prosecution of any suit or proceeding against any Released Party, in any forum (i) any Retained Claim or (ii) any Compliance Dispute (whether or not asserted), which respectively shall be asserted and pursued only pursuant to Sections 12 and 13.6 of the Settlement Agreement. This agreement and covenant

Page 11 of 19

- Notwithstanding any other provision of this Order or the Settlement 11. Agreement (including, without limitation, Section 13.2), nothing in this Order or the Settlement Agreement shall be deemed to in any way impair, limit, or preclude the Releasing Parties' rights to enforce any provision of the Settlement Agreement, or any court order implementing the Settlement Agreement, in a manner consistent with the terms of the Settlement Agreement.
- 12. Except as provided in Section 13.1 of the Settlement Agreement, nothing in this Order or the Settlement Agreement prevents the Representative Plaintiffs and the Class from pursuing claims to hold any Person that is not a Released Party liable for damages caused by that Person's conduct in a conspiracy involving any Released Party.
- If any Releasing Parties pursue claims against any Person for damages 13. allegedly caused by any Released Party, any finding, judgment, opinion, or other result from such proceeding under any circumstances (i) shall not be deemed, construed, or asserted as a finding, judgment, opinion, or result against any Released Party; (ii) shall not be deemed, construed, or asserted as res judicata, collateral estoppel, or similar doctrines against any Released Party; and (iii) shall not be admitted or considered as evidence against or used for any purpose against any Released Party in any judicial, administrative, regulatory, arbitration proceeding, or any other forum.

- The Action shall be dismissed with prejudice as to Released Parties, and a 14. Judgment Dismissing Certain Blue Parties with Prejudice (the "Judgment") shall be entered in form attached as Exhibit 1 hereto, which shall constitute a final judgment on the merits to which the principles of res judicata shall apply to the fullest extent of the law as to the Released Parties.
- 15. The Parties stipulate that the affected Blue Parties shall suffer irreparable harm if a Releasing Party takes action inconsistent with Sections 13.1, 13.2, 13.3, 13.4 and/or 13.7 of the Settlement Agreement, and that in that event the affected Blue Parties may seek an injunction from the Court as to such action without further showing of irreparable harm.
- With respect to all Released Claims, the Releasing Parties and each of them 16. are deemed to have expressly and unconditionally waived and relinquished to the fullest extent permitted by law: (a) the provisions, rights, and benefits conferred by Section 1542 of the California Civil Code, which provides:

"A GENERAL RELEASE DOES NOT EXTEND TO CLAIMS WHICH THE CREDITOR DOES NOT KNOW OR SUSPECT TO EXIST IN HIS OR HER FAVOR AT THE TIME OF EXECUTING THE RELEASE, WHICH IF KNOWN BY HIM OR HER MUST HAVE MATERIALLY AFFECTED HIS OR HER SETTLEMENT WITH THE DEBTOR."

and (b) any law of any state or territory of the United States, federal law or principle of common law, or of international or foreign law, which is similar, comparable, or equivalent to Section 1542 of the California Civil Code. Each Class Member and each Signatory Medical Society may hereafter discover facts other than or different from those which he, she, or it knows or believes to be true with respect to the claims which are the subject matter of the provisions of Section 13 of the Settlement Agreement, but each such Class Member and each Signatory Medical Society is deemed to have expressly and unconditionally waived and fully, finally, and forever settled and released, upon the entry of this Order, any known or unknown, suspected or unsuspected,

contingent or non contingent Claim, with respect to the subject matter of the provisions of Section 13 of the Settlement Agreement, whether or not concealed or hidden, without regard to the discovery or existence of such different or additional facts.

All Persons who are, have been, could be, or could have been alleged to be 17. joint tortfeasors, co-tortfeasors, co-conspirators, or co-obligors with the Released Parties or any of them respecting the Released Claims or any of them, are hereby, to the maximum extent permitted by law, barred and permanently enjoined from making, instituting, commencing, prosecuting, participating in, or continuing any Claim, claim-over, cross-claim, action, or proceeding, however denominated, regardless of the allegations, facts, law, theories, or principles on which they are based, in this Court or in any other court or tribunal, against the Released Parties or any of them with respect to the Released Claims, including without limitation equitable, partial, comparative, or complete contribution, set-off, indemnity, or otherwise, whether by contract, common law, or statute, arising out of or relating in any way to the Released Claims. All such claims are hereby fully and finally barred, released, extinguished, discharged, satisfied, and made unenforceable to the maximum extent permitted by law, and no such claim may be commenced, maintained, or prosecuted against any Released Party. Any judgment or award obtained by a Class Member against any such Person shall be reduced by the amount or percentage, if any, necessary under applicable law to relieve any Released Party of all liability to such Person on such barred claims. Such judgment reduction, partial or complete release, settlement credit, relief, or setoff, if any, shall be in an amount or percentage sufficient under applicable law as determined by the Court to compensate such Person for the loss of any such barred claims against any Released Party. Where the claims of a Person who is, has been, could be, or could have been alleged to be a joint tortfeasor, co-tortfeasor, co-conspirator, or co-obligor with a Released Party respecting the Released Claims

have been barred and permanently enjoined by this Order, the claims of Released Parties against that Person respecting those Released Claims are similarly fully and finally barred, released, extinguished, discharged, satisfied, and made unenforceable to the maximum extent permitted by law.

- 18. In contemplation of the dismissal with prejudice of all claims that would constitute Released Claims as to the Blue Parties or any other Released Party pursuant to this Order, all proceedings are stayed as to the Blue Parties or any other Released Party who is a defendant in any action brought by or on behalf of Class Members that asserts any claim that as of the date of this Order would constitute a Released Claim that that has been, or will in the future, be designated as a tag-along action or consolidated with the Provider Track actions under MDL Docket No. 1334 (the "Tag-Along Actions" (a list of such pending Tag-Along Actions is attached as Exhibit N to the Settlement Agreement)), provided, however, that this stay in contemplation of dismissal shall not apply to any such action to the extent that a named plaintiff in such action has timely elected to Opt-Out of the Settlement and the Class.
- 19. In accordance with the terms of the Settlement Agreement, and except as set forth in Exhibit O to the Settlement Agreement with respect to BCBSA, the Releasing Parties and Class Counsel are barred from pursuing discovery against the Released Parties in the Action and in any action brought by or on behalf of Class Members that asserts any claim that as of the date of this Order would constitute a Released Claim. It is the intent of the Parties to limit, to the maximum extent appropriate under law, further discovery burdens on the Released Parties in the Action. Accordingly, Persons not released by this Order will be permitted to obtain discovery from the Released Parties in the Action and in any action brought by or on behalf of Class Members that asserts any claim that as of the date of this Order would constitute a Released Claim

Page 15 of 19

that has been, or will in the future be assigned to this Court under Case No.

03-21296-CIV-MORENO or MDL Docket No. 1334 only upon first moving the Court for leave to obtain said discovery and demonstrating (a) good cause for said discovery and (b) that such discovery cannot be reasonably obtained from any other source.

In addition to the restrictions on discovery from Released Parties discussed 20. above, and notwithstanding any of the other provisions of this Order, the Released Parties shall have no obligation to preserve documents and evidence with respect to Released Claims, including any obligations under the Agreed Order for Preservation of Records as entered in and applicable to the Action, and the Releasing Parties and Class Counsel shall not pursue any spoliation claims or other actions or sanctions against a Released Person with respect to documents or evidence related to the Released Claims. This Order does not relieve any Party from their obligations under the Agreed Order for Preservation of Records as entered in and applicable to Solomon, et al. v. Blue Cross and Blue Shield Ass'n, et al., Case No. 03-22935-CIV-MORENO/SIMONTON.

Applications for Attorneys' Fees and Representative Plaintiff Fees

21. The Court has reviewed the application for an award of fees and expenses submitted by Class Counsel and the exhibits, memoranda of law, and other materials submitted in support of that application. The Court recognizes that in the Settlement Agreement the Blue Parties have agreed not to oppose an award of fees and expenses to Class Counsel up to \$49,061,965, to be paid by the Blue Plans up to that amount. This agreement is in addition to the monetary consideration and other benefits to be provided to Class Members under the Settlement Agreement. On the basis of its review of the foregoing, the Court hereby awards fees and expenses to Class Counsel in the aggregate amount of \$_____, to be paid by the Blue Plans in accordance with the provisions of the Settlement Agreement.

Page 16 of 19

Representative Plaintiffs and Representative Plaintiffs in Other Actions. The Court recognizes that in the Settlement Agreement the Blue Parties have agreed not to oppose an award of fees up to Seven Thousand Five Hundred Dollars (\$7,500.00) for each Representative Plaintiff and each Representative Plaintiff in Other Actions, to be paid by the Blue Parties up to that amount. This agreement is in addition to the monetary consideration and other benefits to be provided to Class Members under the Settlement Agreement. On the basis of its review of the foregoing, the Court hereby awards a fee of \$ ________ to each Representative Plaintiff and each Representative Plaintiff in Other Actions, to be paid by the Blue Parties in accordance with the provisions of the Settlement Agreement.

Case 1:03-cv-21296-FAM

Other Provisions

therein, nor any negotiations, statements, or proceedings in connection therewith shall be construed as, or be deemed to be evidence of, an admission or concession on the part of any of the Representative Plaintiffs, the Signatory Medical Societies, Class Counsel, any members of the Class, the Blue Parties, or any other Person of any liability or wrongdoing by them, or that the claims and defenses that have been, or could have been, asserted in the Action are or are not meritorious, and this Order, the Settlement, the Settlement Agreement, or any such communications shall not be offered or received in evidence in any action or proceeding, or be used in any way as an admission or concession or evidence of any liability or wrongdoing of any nature or that Representative Plaintiffs, the Signatory Medical Societies, any Class Member, the Blue Parties, or any other Person has or has not suffered any damage; provided, however, that the Settlement Agreement, this Order, and the Judgment to be entered thereon may be filed in any

Page 17 of 19

action by the Blue Parties or any Released Party seeking to enforce the Settlement Agreement, this Order or the Judgment by injunctive or other relief or to assert defenses, including but not limited to res judicata, collateral estoppel, release, good faith settlement, judgment bar or reduction, or any theory of claim preclusion or issue preclusion or similar defense or counterclaim. The terms of the Settlement Agreement and of this Order and the Judgment shall be forever binding on, and shall have res judicata and preclusive effect in, all pending and future lawsuits or other proceedings that are subject to the release and other prohibitions that are set forth in this Order that are maintained by, or on behalf of, the Releasing Parties or any other Person subject to those provisions of this Order.

- 24. The following Persons or entities are hereby granted limited immunity from liability for their actions under the Agreement, except for any willful misconduct or gross negligence: the Billing Dispute External Reviewer; the Independent Review Organization or Organizations (and members and agents, if any); the Class Compliance Dispute Facilitator (and agents, if any); the Blue Compliance Dispute Facilitator (and agents, if any); and the Compliance Dispute Officer (and agents, if any), including individual Compliance Dispute Facilitators and/or Compliance Dispute Officers selected in accordance with Section 12.7 of the Agreement.
- 25. In the event that the Effective Date cannot occur, or the Settlement Agreement is canceled or terminated by Plaintiffs or by all Blue Parties in accordance with the terms and provisions of the Settlement Agreement, then this Order and the Judgment shall be rendered null and void and be vacated and all orders entered in connection therewith by this Court shall be rendered null and void, subject to Section 14.6 of the Settlement Agreement. In the event any individual Blue Party terminates its participation in the Settlement Agreement in accordance

Page 18 of 19

with its terms and provisions, then this Order and the Judgment and all orders entered in connection therewith, shall be rendered null and void as to the terminating Blue Party.

Entry of Judgment; Continuing Jurisdiction

- The Clerk of the Court is directed to enter the Judgment in the form attached 26. to this Order dismissing all Released Claims with prejudice as to the Blue Parties pursuant to Rule 54(b) of the Federal Rules of Civil Procedure.
- Without in any way affecting the finality of this Order and the Judgment, 27. this Court hereby retains jurisdiction as to all matters relating to (a) the interpretation, administration, and consummation of the Settlement Agreement and (b) the enforcement of the injunctions in this Order.

SO ORDERED this ____ day of _______, 2007.

HON. FEDERICO A. MORENO	

UNITED STATES DISTRICT COURT

EXHIBIT D2

IN THE UNITED STATES DISTRICT COURT FOR THE SOUTHERN DISTRICT OF FLORIDA Miami Division

RICK LOVE, M.D., et al., Plaintiffs,)) Case No. 03-21296-) CIV-MORENO/SIMONTON
v.))
BLUE CROSS & BLUE SHIELD	j j
ASSOCIATION, et al.,)
Defendants.)
)

JUDGMENT DISMISSING CERTAIN DEFENDANT **BLUE PARTIES WITH PREJDUICE**

In accordance with, and for the reasons set forth in, the Order Approving Settlement Among Certain Blue Parties, Physicians, Physician Groups and Physician Organizations, Certifying Class and Directing Entry of Final Judgment, entered on ______, 2007, all claims against Blue Cross and Blue Shield Association, Blue Cross and Blue Shield of Alabama, Premera, Premera Blue Cross, PremeraFirst, Inc., LifeWise Health Plan of Washington, LifeWise Health Plan of Oregon, Inc., LifeWise Health Plan of Arizona, Inc. Premera Blue Cross Blue Shield of Alaska, Corp., Medical Services Corporation of Eastern Washington, NorthStar Administrators, Inc., CareFirst, Inc., Group Hospitalization and Medical Services, Inc., Access America, Inc., The GHMSI Companies, Inc., CareFirst BlueChoice, Inc., Capital Care, Inc., CareFirst of Maryland, Inc., CFS Health Group, Inc., Blue Cross and Blue Shield of Florida, Inc., Health Options, Inc., Hawaii Medical Service Association, The Regence Group, Regence BlueShield of Idaho, Inc., Regence BlueCross BlueShield of Utah, Regence BlueCross BlueShield of Oregon, Regence BlueShield, Regence Life and Health Insurance

Company, RegenceCare, Regence HMO Oregon, Regence Health Maintenance of Oregon, Inc., Healthwise, Asuris Northwest Health, Wellmark, Inc. d/b/a Wellmark Blue Cross and Blue Shield of Iowa, Wellmark Health Plan of Iowa, Inc., Wellmark Community Insurance, Inc., Wellmark of South Dakota, Inc. dba Wellmark Blue Cross and Blue Shield of South Dakota, Louisiana Health Service & Indemnity Company dba Blue Cross and Blue Shield of Louisiana, HMO Louisiana, Inc., Blue Cross and Blue Shield of Massachusetts, Inc., Blue Cross and Blue Shield of Massachusetts HMO Blue, Inc., Blue Cross Blue Shield of Michigan, BCBSM, Inc. dba BlueCross BlueShield of Minnesota, HMO Minnesota dba Blue Plus, Comprehensive Care Services, Inc., Blue Cross & Blue Shield of Mississippi, HMO of Mississippi, Inc., Blue Cross and Blue Shield of Montana, Inc., Horizon Healthcare Services, Inc. d/b/a Horizon Blue Cross Blue Shield of New Jersey, Horizon Healthcare Plan Holding Company, Inc., Horizon Healthcare Insurance Company of New York, Horizon Healthcare of New Jersey, Inc., Horizon Healthcare of New York, Inc., Enterprise Holding Company, Inc., AtlantiCare Administrators, Inc., Horizon Healthcare Administrators, Inc., Horizon AtlantiCare LLC, Horizon Healthcare of Delaware, Inc., NASCO of New Jersey, Inc., Empire HealthChoice Assurance, Inc., d/b/a Empire Blue Cross Blue Shield, Empire HealthChoice HMO, Inc., WellChoice Insurance of New Jersey, Inc., WellChoice Holdings of New York, Inc., WellPoint Holding Corp., Blue Cross and Blue Shield of North Carolina, Hospital Service Association of Northeastern Pennsylvania, HMO of Northeastern Pennsylvania, Triple-S, Inc.; Triple-S, Inc., of Puerto Rico, Triple-C, Inc., Blue Cross Blue Shield of Rhode Island, Blue Cross and Blue Shield of South Carolina, BlueChoice HealthPlan of South Carolina, Inc. f/k/a Companion HealthCare Corporation, Planned Administrators, Inc., Thomas H. Cooper & Co., Inc., BlueCross BlueShield of Tennessee, Inc., Tennessee Health Care Network, Inc., Health Care Service Corporation, Group

Health Maintenance Organization, Inc. d/b/a Bluelincs HMO, Group Health Service of Oklahoma, Inc. d/b/a Blue Cross Blue Shield of Oklahoma, HMO New Mexico, Inc., New Mexico Blue Cross and Blue Shield, Inc. d/b/a Blue Cross and Blue Shield of New Mexico, New Mexico Blue Cross and Blue Shield, Inc., Hallmark Services Corporation, BCI HMO, Inc. and HCSC Insurance Services Corporation are dismissed with prejudice as to the Representative Plaintiffs and as to all Class Members pursuant to Rule 54(b) of the Federal Rules of Civil Procedure, provided that the claims of those putative Class Members who elected to Opt-Out of the Class (identified on the attached list) are dismissed without prejudice.

Date:	
	Clerk of the Court

Page 1 of 9

UNITED STATES DISTRICT COURT FOR THE SOUTHERN DISTRICT OF FLORIDA **Miami Division**

EXHIBIT E

RICK LOVE, M.D., et al.,)	
Plaintiffs)	Case No. 03-21296-CIV-MORENO/SIMONTON
v.)	Case No. 05-21270-CIV-MOREMO/SHITONION
BLUE CROSS AND BLUE SHIELD ASSOCIATION, et al.,)	
Defendants)))	
	<u>_</u>	

NOTICE OF PROPOSED SETTLEMENT OF CLASS ACTION WITH CERTAIN BLUE CROSS AND BLUE SHIELD PARTIES, OF SETTLEMENT HEARING TO CONSIDER THE PROPOSED SETTLEMENT, AND OF YOUR RIGHTS CONCERNING THE PROPOSED SETTLEMENT

IF YOU ARE A PHYSICIAN, PHYSICIAN GROUP, OR PHYSICIAN ORGANIZATION WHO PROVIDED COVERED SERVICES TO ANY INDIVIDUAL ENROLLED IN OR COVERED BY BLUE CROSS AND BLUE SHIELD PLANS AT ANY TIME BETWEEN MAY 22, 1999 AND [PRELIMINARY APPROVAL DATE], PLEASE READ THIS NOTICE CAREFULLY.

THIS CLASS ACTION AND THE PROPOSED SETTLEMENT MAY AFFECT YOUR RIGHTS.

I. WHY SHOULD YOU READ THIS NOTICE?

If you are or have been a Physician, Physician Group, or Physician Organization 1/ who or which practiced in the United States since May 22, 1999, your rights may be affected by a proposed settlement with certain Blue Cross and Blue Shield parties ("Settlement") in the class action lawsuit known as Love, et al. v. Blue Cross and Blue Shield Ass'n, et al., (formerly, Thomas, et al. v. Blue Cross and Blue Shield Ass'n, et al.), Case No. 03-21296-CIV-MORENO/SIMONTON (the "Action"), which is pending in the U.S. District Court for the Southern District of Florida (the "Court"). The proposed Settlement and the Action involve the Blue Cross and Blue Shield Association ("BCBSA") and certain Blue Cross and Blue Shield Plans and certain current and former Subsidiaries and Affiliates (the "Blue Parties").

The Blue Cross and Blue Shield Plans and their current and former Subsidiaries and Affiliates involved in the proposed Settlement are: Blue Cross and Blue Shield of Alabama, Premera, Premera Blue Cross, PremeraFirst, Inc., LifeWise Health Plan of Washington, LifeWise Health Plan of Oregon, Inc., LifeWise Health Plan of Arizona, Inc., Premera Blue Cross Blue Shield of Alaska, Corp., Medical Services Corporation of Eastern Washington, NorthStar Administrators, Inc., CareFirst, Inc., Group Hospitalization and Medical Services, Inc., Access America, Inc., The GHMSI Companies, Inc., CareFirst BlueChoice, Inc., Capital Care, Inc., CareFirst of Maryland, Inc., CFS Health Group, Inc., Blue Cross and Blue Shield of Florida, Inc., Health Options, Inc., Hawaii Medical Service Association, The Regence Group, Regence BlueShield of Idaho, Inc., Regence BlueCross BlueShield of Utah, Regence BlueCross BlueShield of Oregon, Regence BlueShield, Regence Life and Health Insurance Company, RegenceCare, Regence HMO Oregon, Regence Health Maintenance of Oregon, Inc., Healthwise, Asuris Northwest Health, Wellmark, Inc. d/b/a Wellmark Blue Cross and Blue Shield of Iowa, Wellmark Health Plan of Iowa, Inc., Wellmark Community Insurance, Inc., Wellmark of South Dakota, Inc. dba Wellmark Blue Cross and Blue Shield of South Dakota, Louisiana Health Service & Indemnity Company dba Blue Cross and Blue Shield of Louisiana, HMO Louisiana, Inc., Blue Cross and Blue Shield of Massachusetts, Inc., Blue Cross and Blue Shield of Massachusetts HMO Blue, Inc., Blue Cross Blue Shield of Michigan, BCBSM, Inc. dba BlueCross BlueShield of Minnesota, HMO Minnesota dba Blue Plus, Comprehensive Care Services, Inc., Blue Cross & Blue Shield of Mississippi, HMO of Mississippi, Inc., Blue Cross and Blue Shield of Montana, Inc., Horizon Healthcare Services, Inc. d/b/a Horizon Blue Cross Blue Shield of New Jersey, Horizon Healthcare Plan Holding Company, Inc., Horizon Healthcare Insurance Company of New York, Horizon Healthcare of New Jersey, Inc., Horizon Healthcare of New York, Inc., Enterprise Holding Company, Inc., AtlantiCare Administrators, Inc., Horizon Healthcare Administrators, Inc., Horizon AtlantiCare LLC, Horizon Healthcare of Delaware, Inc., NASCO of New Jersey, Inc., Empire HealthChoice Assurance, Inc., d/b/a Empire Blue Cross Blue Shield, Empire HealthChoice HMO, Inc., WellChoice Insurance of New Jersey, Inc., WellChoice Holdings of New York, Inc., WellPoint Holding Corp., Blue Cross and Blue Shield of North Carolina, Hospital Service Association of Northeastern Pennsylvania, HMO of Northeastern Pennsylvania, Triple-S, Inc.; Triple-S, Inc., of Puerto Rico, Triple-C, Inc., Blue Cross Blue Shield of Rhode Island, Blue Cross and Blue Shield of South Carolina, BlueChoice HealthPlan of South Carolina, Inc. f/k/a Companion HealthCare Corporation, Planned Administrators, Inc., Thomas H. Cooper & Co., Inc., BlueCross BlueShield of Tennessee, Inc., Tennessee Health Care Network, Inc., Health Care Service Corporation, Group Health Maintenance Organization, Inc. d/b/a Bluelincs HMO, Group Health Service of Oklahoma, Inc. d/b/a Blue Cross Blue Shield of Oklahoma, HMO New Mexico, Inc., New Mexico Blue Cross and Blue Shield, Inc. d/b/a Blue Cross and Blue Shield of New Mexico, New Mexico Blue Cross and Blue Shield, Inc., Hallmark Services Corporation, BCI HMO, Inc., HCSC Insurance Services Corporation. (the "settling Blue Plans").

You may be a Class Member (described below) in the Action if you have provided Covered Services to Plan Members enrolled in or covered by a plan offered or administered by any of these Blue Parties or by other primary licensees of the BCBSA.

The Representative Plaintiffs have agreed to settle all claims against the Blue Parties in the Action in exchange for the Blue Parties' commitments regarding their business practices, and the establishment of a Settlement Fund with respect to which Class Members can make claims for settlement payment. The Court has scheduled a hearing to consider the fairness, reasonableness, and adequacy of the proposed settlement with the Blue Parties, together with certain other matters, on [DATE], at [TIME], at the United States Courthouse, Courtroom IV, Tenth Floor, Federal Justice Building, 99 Northeast Fourth Street, Miami, Florida, 33132 (the "Settlement Hearing"). Please note that the Court may adjourn the Settlement Hearing without further written notice to putative Class Members.

You may be a Class Member who would be entitled to receive the benefits of the proposed Settlement. As a Class Member, however, you will also be bound by the release and other provisions of the Settlement if it is approved by the Court. You may elect to Opt-Out of the Class and the Settlement, as explained below. You may also have a right to object to the Settlement or to the applications for attorneys' fees and Representative Plaintiffs' fees that Class Counsel intend to make to the Court, but only if you comply with the procedures described in this notice. BECAUSE YOUR RIGHT TO PURSUE CERTAIN TYPES OF CLAIMS AGAINST THE BLUE PARTIES, AND CERTAIN OTHERS AFFILIATED WITH THE BLUE PARTIES, MAY BE AFFECTED BY THE SETTLEMENT, YOU SHOULD READ THIS NOTICE CAREFULLY.

2

All capitalized terms in this notice are defined in the Settlement Agreement, copies of which are available as described below.

II. WHAT IS THIS LITIGATION ABOUT?

The Complaint in the Action alleges, among other things, that between 1999 and the present, the Blue Parties, among others, engaged in a conspiracy to improperly deny, delay, and/or reduce payments to physicians, physician groups, and physician organizations by engaging in several types of allegedly improper conduct, including but not limited to:

- Misrepresenting and/or failing to disclose the use of edits to unilaterally "bundle," "downcode," and/or reject claims for medically necessary covered services;
- Failing to pay for "medically necessary" services in accordance with member plan documents;
- Failing and/or refusing to recognize CPT® modifiers;
- Concealing and/or misrepresenting the use of improper guidelines and criteria to deny, delay, and/or reduce payment for medically necessary covered services;
- Misrepresenting and/or refusing to disclose applicable fee schedules; and
- Failing to pay claims for medically necessary covered services within the required statutory and/or contractual time periods.

The Complaint in the Action claims that the conduct described above violated the federal statute entitled the Racketeer Influenced and Corrupt Organizations Act, 18 U.S.C. § 1961, et seq. The Blue Parties deny the factual allegations and legal claims asserted in the Complaint in the Action, and deny any liability or wrongdoing relating thereto. If you would like further information about the claims asserted in the Action, you can review a copy of the Complaint at: [Insert the Settlement Administrator's Website;] www.hmosettlements.com; www.WhatleyDrake.com; www.ArchieLamb.com; and www.kttlaw.com

The Action asserting these and other claims against the Blue Parties and other Blue Cross and Blue Shield licensed plans began in approximately 2003 before the federal court in Miami. Since the initial complaint was filed, substantial proceedings have occurred, including the production of a significant number of documents by the defendants and the depositions of various witnesses. As part of the Settlement, the Blue Parties would no longer actively participate in the Action. Other defendants in the Action comprising WellPoint, Inc. and the current and former direct and indirect subsidiaries and affiliates of WellPoint, Inc. and the former Anthem, Inc. agreed to a separate settlement with, among others, the Representative Plaintiffs that was given final approval by the Court in an order entered on December 22, 2005 (and amended on January 3, 2006). Other defendants in the Action who are not WellPoint entites or Parties to this proposed Settlement may continue to defend the claims asserted in the Action on various grounds.

III. WHAT ARE THE TERMS OF THE PROPOSED SETTLEMENT?

In a settlement agreement dated April 27, 2007 (the "Settlement Agreement"), the Representative Plaintiffs and the Signatory Medical Societies have agreed to settle all claims that were or could have been asserted against the Blue Parties and certain of their Affiliates and Subsidiaries in the Action, in exchange for the Blue Parties' commitments regarding their business practices and for certain monetary consideration. The terms of the Settlement Agreement are summarized in this notice, but a copy of the entire Settlement Agreement can be reviewed at [Insert the Settlement Administrator's Website]. It is also available at www.hmosettlements.com, www.WhatleyDrake.com, www.ArchieLamb.com, and www.kttlaw.com.

A. The Settlement Class

The proposed Settlement with the Blue Parties will be on behalf of the following Persons (collectively, the "Class" and each member of the Class a "Class Member"), which Class has been conditionally certified for settlement proposes:

> "Class" means any and all Physicians, Physician Groups and Physician Organizations who provided Covered Services to any Plan Member or services to any individual enrolled in or covered by a Plan offered or administered by any Person named as a defendant in the Complaint or by any other primary licensee of the BCBSA or by any of their respective current or former subsidiaries or Affiliates, in each case from May 22, 1999 through [Preliminary Approval Date]. The Class shall exclude: (i) all Persons who, in accordance with the terms of this Agreement, execute a timely request for exclusion (Opt-Out) from the Class; and (ii) the Blue Parties, their Affiliates and any of their officers, directors, and employees.

B. The Settlement Consideration

If the Settlement is approved by the Court, the Settlement Agreement provides for both monetary and other benefits to be provided by the Blue Parties to Class Members.

Page 4 of 9

C. Business Practice Commitments

Document 928-12

As a part of the Settlement, the Blue Parties have agreed to commitments regarding their business practices. For example, the Blue Parties have agreed, subject to implementation dates described in the Settlement Agreement to: (a) include in their contracts with Physicians a definition of Medical Necessity that bases Medical Necessity determinations on factors including generally accepted standards of medical practice; (b) use clinical guidelines that are based on credible scientific evidence published in peer reviewed medical literature (taking into account Physician Specialty Society recommendations, the views of Physicians practicing in the relevant clinical areas, and other relevant factors) when making Medical Necessity determinations; (c) provide Class Members access to the applicable Blue Party's Medical Necessity external review process; (d) establish an external review process for resolving Billing Disputes with Class Members; (e) make investments designed to facilitate the automated adjudication of claims submitted by Physicians and thereby reduce the average time taken by the Blue Parties to pay valid claims; (f) continue to fund initiatives to reduce the percentage of resubmitted claims for Covered Services; (g) not automatically reduce the intensity coding of evaluation and management codes billed for certain covered services; (h) disclose payment rules and conform bundling and other Edit practices and proceduress as specified in the Settlement Agreement; (i) not include "gag clauses" in its contracts with Class Members; (j) continue to devote resources to improve accuracy of information about eligibility of Plan Members; (k) where all necessary information is available to the applicable Blue Plan, ensure the payment of valid Complete Claims within 15 business days for electronicallysubmitted claims and 30 days for paper claims; (1) provide relevant Class Members with the ability to view applicable fee schedule amounts for billing codes related to their practice; (m) follow protocols for the reimbursement of Physicians providing mental health care services as specified in the Settlement Agreement; and (n) establish a Compliance Dispute resolution mechanism to address disputes regarding the Blue Parties' compliance with the Settlement Agreement. In addition, the Blue Parties will disclose additional information about their claim administration policies and procedures on their existing websites. These changes, as well as others, are more fully described in the Settlement Agreement.

D. The Settlement Fund

As a part of the Settlement, the Blue Parties have agreed to make a settlement payment of \$128,315,907 which, together with accrued interest from [30 days after the Preliminary Approval Date], will be distributed to Physicians who are Class Members and who timely file a Claim Form. If the Settlement is approved by the Court, these Class Members will be entitled to payments from the Settlement Fund in accordance with formulas that are set forth in the Settlement Agreement.

- A Physician who provided any Covered Services reimbursed by primary licensees of the BCBSA since May 22, 1999 but who has since become an inactive Physician, retired from the practice of, or otherwise ceased to practice, medicine, or has died as of [the Preliminary Approval Date] (a "Retired Physician"), will receive a pro rata portion of the amount of the Settlement Fund that has been allocated to such Retired Physicians.
- An actively-practicing Physician (an "Active Physician") whose Gross Receipts for the calendar years 2004, 2005, and 2006 for providing Covered Services to the settling Blue Plans' Plan Members were in the aggregate zero or less than \$5,000 will receive a single Base Amount (determined pro rata according to the claims that are filed) from the amount of the Settlement Fund that is allocated to Active Physicians.
- An Active Physician whose Gross Receipts for the calendar years 2004, 2005, and 2006 for providing Covered Services to the settling Blue Plans' Plan Members were in the aggregate at least \$5,000 but less than \$50,000 will receive a pro rata portion of five (5) times the Base Amount from the amount of the Settlement Fund that is allocated to Active Physicians.
- An Active Physician whose Gross Receipts for the calendar years 2004, 2005, and 2006 for providing Covered Services to the settling Blue Plans' Plan Members were in the aggregate \$50,000 or greater, will receive a pro rata portion of ten (10) times the Base Amount from the amount of the Settlement Fund that is allocated to Active Physicians.
- Alternatively, an Active Physician may establish, through the submission of billing records or similar information, that his or her settlement compensation category (of the three provided and described above) should be based on aggregate payments received for providing Covered Services to the settling Blue Plans' Plan Members over any consecutive three-year period from May 22, 1999 through December 31, 2006.
- In determining Gross Receipts, Physicians should include amounts paid by the settling Blue Plans directly or by intermediaries for providing Covered Services to the settling Blue Plans' Plan Members. For example, a Physician may have provided services through an intermediary that contracted with a Blue Plan to provide the services (for example, an IPA, medical group, organized delivery system, Physician hospital organization, etc.). In determining Gross Receipts for providing Covered Services to the settling Blue Plans' Plan Members, Physicians should also include amounts they received from such intermediaries for treating the settling Blue Plans' Plan Members.
- Active Physicians against whom a Blue Plan has obtained a final finding of fraud and/or abuse (as that term is defined in Section 8.3(h) of the Settlement Agreement) from a judicial, arbitral, or administrative proceeding and a corresponding final judgment for damages arising from a claim (or claims) for payment for the same time period for which a claim may be asserted are not entitled to payment from the Settlement Fund regardless of their status as Class Members.

Page 5 of 9

- Physician Groups and Physician Organizations may submit claims on behalf of individual Physicians employed by or otherwise working with them at the time that the claims are made, without the necessity of individual signatures from the individual Physicians, if authorized to do so by such Physicians or the compensation for the Covered Services provided by such Physicians belonged to the Physician Group or Physician Organization. A Physician Group or Physician Organization may only submit claims on behalf of individual Physicians, not on behalf of the Physician Group or Physician Organization itself. No Covered Service provided by an individual Physician may be the subject of more than one submission, either by the individual Physician or by a Physician Group or Physician Organization on the Physician's behalf, but not both. Physician Groups and Physician Organizations may make claims on behalf of individual Physicians employed by or otherwise working with them who are Class Members, but not on behalf of, or with respect, to Covered Services provided by any individual Physicians that choose to Opt-Out of the Class and the Settlement. Nor may Physician Groups or Physician Organizations make claims on behalf of any individual Active Physicians against whom a Blue Plan has obtained a final finding of fraud and/or abuse (as that term is defined in Section 8.3(h) of the Settlement Agreement) from a judicial, arbitral, or administrative proceeding and a corresponding final judgment for damages arising from a claim (or claims) for payment for the same time period for which a claim may be asserted.
- The Settlement Fund is described in detail in Section 8 of the Settlement Agreement.

If you are a Physician, the heir of a deceased Physician, or a Physician Group or Physician Organization submitting a claim on behalf of an individual Physician, who is eligible to participate in the Settlement Fund, you may submit a claim using the Claim Form and Claim Form Instructions enclosed with this notice, to the Settlement Administrator at the following address:

Blue Parties' Settlement Administrator [Address]

E. Charitable Organizations

Class Members may elect to have the amount of their settlement payment from the Settlement Fund (discussed in the prior section) contributed on their behalf to a charitable organization set forth on Exhibit L to the Settlement Agreement. A list of the charitable organizations that may receive your donation is also attached to the Claim Form Instructions enclosed with this notice.

F. The Release and Dismissal with Prejudice

Upon final approval of the Settlement, the Action will be dismissed with prejudice as to the Blue Parties. In addition, Blue Parties, and certain others affiliated with the Blue Parties (collectively, the "Released Parties"), will receive a release and discharge from the Class (which would not include putative Class Members who or which timely elect to Opt-Out of the Class and the Settlement, as discussed below) of any and all causes of action, judgments, liens, indebtedness, costs, damages, obligations, attorneys' fees, losses, claims, liabilities and demands of whatever kind, source or character whether arising under any federal or state law, which (consistent with the Parties' understanding of the settlements in Shane) includes, but is not limited to, the Racketeer Influenced and Corrupt Organizations Act, antitrust and other statutory and common law claims, intentional or nonintentional, (each a "Claim"), arising on or before the Effective Date, that are, were or could have been asserted against any of the Released Parties by reason of, arising out of, or in any way related to any of the facts, acts, events, transactions, occurrences, course of conduct, business practices, representations, omissions, circumstances, or other matters referenced in the Action, or addressed in the Settlement Agreement, whether any such Claim was or could have been asserted by any Releasing Party on its own behalf or on behalf of other Persons. The provisions regarding the discharge of all Released Claims is discussed in Section 13.1 of the Settlement Agreement, to which you should refer if you have any questions as to its applicability.

Excluded from the release are claims for certain Covered Services that are or were in the process of being submitted to, adjudicated or paid by the Blue Parties at or around the time that the Settlement Agreement was entered into and the Court approval process was begun. This provision regarding "Retained Claims" is discussed in more detail in Section 13.6 of the Settlement Agreement, to which you should refer if you have any questions as to its applicability.

Except as described above, nothing in the Settlement Agreement is intended to relieve any Person that is not a Released Party from responsibility for its own conduct or the conduct of other Persons or entities who are not Released Parties for claims that are not Released Claims, or to preclude any Representative Plaintiff from introducing any competent and admissible evidence to the extent consistent with the Settlement Agreement. In addition, except as described above, nothing in the Settlement Agreement prevents the Representative Plaintiffs from pursuing claims to hold any Person that is not a Released Party liable for damages caused by that Party's conduct in a conspiracy involving any Released Party.

Page 6 of 9

Finally, the Settlement Agreement includes a covenant not to sue the Released Parties (including the Blue Parties) for Claims that are subject to the release, with certain limited exceptions which are described in detail in Sections 13.6 and 13.7 of the Settlement Agreement, to which you should refer if you have any questions as to their applicability.

The release provided for in the Settlement Agreement applies to Claims whether they are known or unknown. In this regard, each Class Member and the Class shall be deemed expressly to have waived and relinquished: (a) the provisions, rights and benefits conferred by California Civil Code § 1542, which reads:

"Section 1542. General Release - Claims Extinguished. A general release does not extend to claims which the creditor does not know or suspect to exist in his favor at the time of executing the release, which if known by him must have materially affected his settlement with the debtor;"

and (b) any law of any state or territory of the United States, federal law, or principle of common law, or of international or foreign law which is similar, comparable, or equivalent to § 1542 of the California Civil Code.

IV. WHAT WILL HAPPEN AT THE SETTLEMENT HEARING?

Case 1:03-cv-21296-FAM

As noted above, the Settlement Hearing will be held on [DATE], at [TIME], at the United States Courthouse, Courtroom IV, Tenth Floor, Federal Justice Building, 99 Northeast Fourth Street, Miami, Florida, 33132. However, the order scheduling that hearing also provides that it may be adjourned by the Court without any additional notice to putative Class Members other than an announcement in open court.

At the Settlement Hearing, the Court will consider several different issues, including:

First, the Court will consider whether the proposed Settlement of the Action with the Blue Parties that is reflected in the Settlement Agreement is fair, reasonable and adequate to putative Class Members.

Second, the Court will consider whether it should certify the Class for settlement purposes pursuant to Rules 23(a), 23(b)(2) and 23(b)(3) of the Federal Rules of Civil Procedure. Among other things, this will require the Court to determine (i) whether questions of law or fact common to the putative Class Members predominate over questions affecting only individual putative Class Members, and (ii) whether a class action is superior to other available methods for fair and efficient adjudication of the controversy. If the Court certifies the Class for settlement purposes, putative Class Members who or which have timely elected to Opt-Out of the Class by following the procedures described below will be excluded from it.

Third, the Court will consider whether to enter orders that would prevent Class Members and certain other Persons, including the defendants in the Action other than the Blue Parties, from asserting certain claims against the Blue Parties in the future.

Fourth, the Court will consider the applications for a payment of fees by the Blue Parties to the Representative Plaintiffs and to the Representative Plaintiffs in Other Actions, which is discussed in more detail below.

Fifth, the Court will consider an application by counsel to the Class for attorneys' fees and expenses to be paid by the Blue Parties, which is also discussed in more detail below.

V. CAN I PARTICIPATE IN THE SETTLEMENT HEARING?

Any putative Class Member, or other interested Person, who or which objects to the proposed Settlement with the Blue Parties, the Settlement Agreement, the applications for attorneys' fees, or the other matters to be considered at the Settlement Hearing may appear and present such objections, provided, however, that Persons who or which have elected to Opt-Out of the Class and the Settlement will not be entitled to object. In order to be permitted to object to the proposed Settlement, however, you must, on or before [75 days after Notice Date] comply fully with the following requirements:

- File with the Court a written statement setting forth your objections to the matters to be considered and the basis for those objections, together with any documentation you want the Court to consider. If you intend to appear at the Settlement Hearing, you must also file a written notice of intention to appear at this same time; and
- Serve copies of all such materials either by hand or overnight delivery upon the following counsel:

Edith M. Kallas, Joe R. Whatley, Jr., Joseph P. Guglielmo WHATLEY DRAKE & KALLAS LLC 1540 Broadway, 37 th Floor New York, NY 10036	Archie Lamb, Jr. LAW OFFICES OF ARCHIE LAMB, LLC 2017 Second Avenue North, 2 nd Floor Birmingham, Al 35203	Harley S. Tropin Kozyak Tropin & Throckmorton, P.A. 2525 Ponce de Leon Blvd., 9 th Floor Miami, Fl 33134
Michael A. Pope MCDERMOTT WILL & EMERY LLP 227 W. Monroe Street Chicago, IL 60606	Chester T. Kamin JENNER & BLOCK LLP 330 North Wabash Chicago, IL 60611	

If you do not comply with the foregoing procedures and deadlines for filing and serving a written statement setting forth your objections, and a written notice of your intention to appear at the Settlement Hearing, if applicable, you may lose substantial legal rights, including but not limited to: the right to appear and be heard at the Settlement Hearing; the right to contest approval of the proposed Settlement or the application for an award of attorneys' fees and expenses to Class Counsel; the right to contest approval of the application for the payment of fees to the Representative Plaintiffs and the Representative Plaintiffs in Other Actions; and the right to contest any other orders or judgments of the Court entered in connection with the proposed Settlement.

If the Court does not approve the proposed Settlement, the Settlement Agreement will be null and void. If there are further actions taken in the Action that affect your rights, you will receive notice as determined by the Court.

VI. HOW DO I FILE A CLAIM?

As discussed above, the proposed Settlement contemplates certain settlement payments to Class Members with respect to Claim Forms that are timely submitted to the Settlement Administrator. In order to qualify for a settlement payment, you must complete the enclosed Claim Form, sign the form, and mail the completed and signed form by NO LATER THAN [90 DAYS AFTER NOTICE DATE] to:

> Blue Parties' Settlement Administrator [Address]

IN ORDER TO BE ENTITLED TO RECEIVE ANY PAYMENT FROM THE SETTLEMENT FUND, YOU MUST COMPLETE AND SIGN THE CLAIM FORM, AND THE ENVELOPE RETURNING YOUR CLAIM FORM MUST BE MAILED TO THE SETTLEMENT ADMINISTRATOR WITH A POSTMARK DATE NO LATER THAN [90 DAYS AFTER NOTICE DATE]. IF YOUR SIGNED CLAIM FORM IS NOT MAILED TO THE SETTLEMENT ADMINISTRATOR BY THIS DEADLINE, YOU WILL BE DEEMED TO HAVE WAIVED YOUR RIGHT TO RECEIVE ANY PAYMENT FROM THE SETTLEMENT FUND.

IF YOU ARE AN ACTIVE PHYSICIAN AGAINST WHOM A BLUE PLAN HAS OBTAINED A FINDING OF FRAUD AND/OR ABUSE (AS THAT TERM IS DEFINED IN SECTION 8.3(h) OF THE SETTLEMENT AGREEMENT) FROM A JUDICIAL, ARBITRAL, OR ADMINISTRATIVE PROCEEDING AND A CORRESPONDING FINAL JUDGMENT FOR DAMAGES FOR THE SAME TIME PERIOD FOR WHICH A CLAIM MAY BE ASSERTED, YOU ARE NOT ENTITLED TO PAYMENT FROM THE SETTLEMENT FUND, REGARDLESS OF YOUR STATUS AS A CLASS MEMBER.

If you submit a claim, or if a Physician Group or Physician Organization submits a claim on your behalf, you will be electing to be a Class Member and will be bound by all proceedings, orders, and judgments entered in connection with the proposed Settlement and the Settlement Agreement, including the release, covenant not to sue, and dismissal with prejudice described above. Physician Groups and Physician Organizations may make claims on behalf of individual Physicians employed by or otherwise working with them who are Class Members, but not on behalf of, or with respect to Covered Services provided by, any individual Physicians who choose to Opt-Out of the Class and Settlement, or on behalf of individual Physicians who are submitting claims on their own behalf for the same Covered Services.

VII. WHAT IF I DO NOT WANT TO BE PART OF THE SETTLEMENT?

If you do <u>not</u> want to be a Class Member and participate in the proposed Settlement, then BY NO LATER THAN [60 DAYS AFTER NOTICE DATE], the Notice Administrator must receive your written request to Opt-Out of the Class and the Settlement, including your name, business address, telephone number, Federal Tax Identification Number, and signature. Physician Groups or Physician Organizations may exclude themselves, as distinct legal entities, from the Class by submitting a complete and timely request to Opt-Out. Physician Groups or Physician Organizations may not request to Opt-Out individual Physicians, unless the Physician Group or Physician Organization has written authorization to act on behalf of those Physicians, which authorization has been submitted to the Notice Administrator. Individual Physicians seeking to Opt-Out must otherwise submit their own individual, complete and timely request to Opt-Out. Opt-Out requests must be submitted to:

Blue Parties' Notice Administrator [Address]

TO BE CONSIDERED TIMELY AND TO EFFECTIVELY OPT-OUT OF THE CLASS AND THE SETTLEMENT, YOUR COMPLETE AND SIGNED OPT-OUT REQUEST MUST BE RECEIVED BY THE NOTICE ADMINISTRATOR BY NO LATER THAN [60 DAYS AFTER NOTICE DATE]. IF IT IS NOT POSTMARKED BY THAT DATE, YOUR RIGHT TO OPT-OUT WILL BE DEEMED WAIVED AND YOU WILL BE BOUND BY ALL ORDERS AND JUDGMENTS ENTERED IN CONNECTION WITH THE SETTLEMENT.

If you choose to Opt-Out of the Settlement and the Class, you will not be entitled to receive the benefits of the proposed Settlement with the Blue Parties, including any payment from the Settlement Fund and any benefit from the business practice commitments specified in the Settlement Agreement. Your claims against the Blue Parties will not be released and you will be free to pursue any claims you believe you have by filing a separate action or request for arbitration if you are subject to an arbitration agreement.

Any putative Class Member who or which timely submits a request to Opt-Out of the Class and the Settlement will have until the Settlement Hearing to deliver to the Notice Administrator a written revocation of the request to Opt-Out and shall thereby become a Class Member.

VIII. WHAT ABOUT ATTORNEYS' FEES, COSTS, AND EXPENSES?

If the Court approves the proposed Settlement, Class Counsel will apply to the Court for an award of attorneys' fees, including costs and expenses. In the Settlement Agreement, the Blue Parties have agreed not to oppose such an application in the aggregate amount of up to \$49,061,965. If the Court awards attorneys' fees and expenses in an amount no greater than that amount, the Blue Plans will pay the amount awarded by the Court to Class Counsel. This payment is in addition to the settlement consideration to the Class Members that is described above and will not reduce the amount available to Class Members if the Settlement is approved.

IX. WHAT ARE THE REPRESENTATIVE PLAINTIFFS' FEES?

In addition to the application by Class Counsel for attorneys' fees and expenses described in the preceding section, in connection with the Court's consideration of the Settlement, the Representative Plaintiffs and the Representative Plaintiffs in Other Actions intend to seek an award from the Court of fees in the amount of up to Seven Thousand and Five Hundred Dollars (\$ 7,500) for each Representative Plaintiff and each Representative Plaintiff in Other Actions, which, if awarded, would be in addition to the settlement consideration that will be available to Class Members generally. In the Settlement Agreement, the Blue Parties have agreed not to oppose such an application up to seven thousand and five hundred dollars (\$ 7,500) for each Representative Plaintiff and each Representative Plaintiff in Other Actions. If the Court awards Representative Plaintiffs and Representative Plaintiffs in Other Actions a fee up to that amount, the Blue Parties will pay such amount to the Representative Plaintiffs and the Representative Plaintiffs in Other Actions.

X. WHOM CAN I CONTACT WITH QUESTIONS?

If you have questions regarding this notice, the proposed Settlement with the Blue Parties, the Settlement Agreement, or the Action generally, you can obtain additional information from the following sources:

On the Internet, at any of these sites: [Insert Settlement Administrator's Website]; www.hmosettlements.com; www.WhatleyDrake.com; www.ArchieLamb.com; www.kttlaw.com

By Telephone:

1-[NUMBER]

(Settlement Administrator)

By Mail:

Whatley Drake & Kallas, LLC 1540 Broadway, 37th Floor New York, New York 10036 Attention: Edith Kallas Law Offices of Archie Lamb, LLC 2017 2nd Avenue Birmingham, Alabama 35203

Attention: Archie Lamb

Kozyak Tropin & Throckmorton 2525 Ponce de Leon Blvd., 9th Floor Miami, Florida 33134

Attention: Harley Tropin

PLEASE DO NOT CALL THE BLUE PARTIES, A BLUE PLAN'S PROVIDER RELATIONS REPRESENTATIVE, THE COURT, OR THE CLERK'S OFFICE.

XI. EXAMINATION OF PAPERS

This notice is a summary and does not describe all details of the Settlement with the Blue Parties, the Settlement Agreement, or the proceedings in the Action generally. Complete copies of the Settlement Agreement and certain pleadings and papers filed in the Action can be found for review on the following website:

[Insert the Settlement Administrator's Website]; www.WhatleyDrake.com; www.WhatleyDrake.com; www.WhatleyDrake.com; www.WhatleyDrake.com; www.kttlaw.com; www.kttlaw.com;

In addition, you may review the complete files of papers submitted in the Action at the office of the Clerk of the Court, United States Courthouse, U.S. District Court for the Southern District of Florida, 301 North Miami Avenue, Miami, Florida, 33128, during regular business hours.

XII. REQUEST TO FORWARD THIS NOTICE

If you would be a Class Member described in this Notice but you have assigned any claim that might be covered by the proposed Settlement or the release described above, please forward this Notice to the appropriate Person as soon as possible.

BY ORDER OF THE UNITED STATES DISTRICT COURT FOR THE SOUTHERN DISTRICT OF FLORIDA

EXHIBIT F

Page 1 of 14

UNITED STATES DISTRICT COURT FOR THE SOUTHERN DISTRICT OF FLORIDA Miami Division

	,
RICK LOVE, M.D., et al.,)
Plaintiffs)) Case No. 03-21296-CIV-
v.) MORENO/SIMONTON)
BLUE CROSS AND BLUE SHIELD ASSOCIATION, et al.,))
Defendants)))
	í

ORDER PRELIMINARILY APPROVING PROPOSED SETTLEMENT AMONG CERTAIN BLUE PARTIES AND PHYSICIANS, PHYSICIAN GROUPS, AND PHYSICIAN ORGANIZATIONS, SETTING FORM AND CONTENT OF NOTICE TO THE CLASS, AND SCHEDULING SETTLEMENT HEARING

The Court having reviewed and considered the Joint Motion for Preliminary Approval of Settlement ("Joint Motion"), filed on April 27, 2007, and having reviewed and considered the terms and conditions of the proposed settlement (the "Settlement") as set forth in the settlement agreement dated April 27, 2007 (the "Settlement Agreement"), ½/ a copy of which has been submitted as Exhibit 1 to the Joint Motion, and on the basis of such submissions, together with any other submissions by the Parties in support of the Joint Motion, and all prior proceedings had in this litigation, good cause for this Order having been shown,

NOW, THEREFORE, IT IS HEREBY ORDERED:

Capitalized terms used in this Order that are not otherwise defined herein have the meaning assigned to them in the Settlement Agreement.

1. The terms of the Settlement Agreement are hereby preliminarily approved, subject to further consideration at the Settlement Hearing provided for below. The Court concludes that the Settlement is sufficiently within the range of reasonableness to warrant the conditional certification of the settlement Class (defined below), the scheduling of the Settlement Hearing and the circulation of notice to putative Class Members (defined below), each as provided for in this Order.

Conditional Certification of the Class

2. For purposes of settlement only, pursuant to Federal Rules of Civil Procedure 23(a), 23(b)(2), and 23(b)(3), solely with respect to the Blue Cross and Blue Shield Association ("BCBSA") and Blue Cross and Blue Shield of Alabama, Premera, Premera Blue Cross, PremeraFirst, Inc., LifeWise Health Plan of Washington, LifeWise Health Plan of Oregon, Inc., LifeWise Health Plan of Arizona, Inc., Premera Blue Cross Blue Shield of Alaska, Corp., Medical Services Corporation of Eastern Washington, NorthStar Administrators, Inc., CareFirst, Inc., Group Hospitalization and Medical Services, Inc., Access America, Inc., The GHMSI Companies, Inc., CareFirst BlueChoice, Inc., Capital Care, Inc., CareFirst of Maryland, Inc., CFS Health Group, Inc., Blue Cross and Blue Shield of Florida, Inc., Health Options, Inc., Hawaii Medical Service Association, The Regence Group, Regence BlueShield of Idaho, Inc., Regence BlueCross BlueShield of Utah, Regence BlueCross BlueShield of Oregon, Regence BlueShield, Regence Life and Health Insurance Company, RegenceCare, Regence HMO Oregon, Regence Health Maintenance of Oregon, Inc., Healthwise, Asuris Northwest Health, Wellmark, Inc. d/b/a Wellmark Blue Cross and Blue Shield of Iowa, Wellmark Health Plan of Iowa, Inc., Wellmark Community Insurance, Inc., Wellmark of South Dakota, Inc. dba Wellmark Blue Cross and Blue Shield of South Dakota, Louisiana Health Service & Indemnity Company dba Blue Cross and Blue

Shield of Louisiana, HMO Louisiana, Inc., Blue Cross and Blue Shield of Massachusetts, Inc., Blue Cross and Blue Shield of Massachusetts HMO Blue, Inc., Blue Cross Blue Shield of Michigan, BCBSM, Inc. dba BlueCross BlueShield of Minnesota, HMO Minnesota dba Blue Plus, Comprehensive Care Services, Inc., Blue Cross & Blue Shield of Mississippi, HMO of Mississippi, Inc., Blue Cross and Blue Shield of Montana, Inc., Horizon Healthcare Services, Inc. d/b/a Horizon Blue Cross Blue Shield of New Jersey, Horizon Healthcare Plan Holding Company, Inc., Horizon Healthcare Insurance Company of New York, Horizon Healthcare of New Jersey, Inc., Horizon Healthcare of New York, Inc., Enterprise Holding Company, Inc., AtlantiCare Administrators, Inc., Horizon Healthcare Administrators, Inc., Horizon AtlantiCare LLC, Horizon Healthcare of Delaware, Inc., NASCO of New Jersey, Inc., Empire HealthChoice Assurance, Inc., d/b/a Empire Blue Cross Blue Shield, Empire HealthChoice HMO, Inc., WellChoice Insurance of New Jersey, Inc., WellChoice Holdings of New York, Inc., WellPoint Holding Corp., Blue Cross and Blue Shield of North Carolina, Hospital Service Association of Northeastern Pennsylvania, HMO of Northeastern Pennsylvania, Triple-S, Inc.; Triple-S, Inc., of Puerto Rico, Triple-C, Inc., Blue Cross Blue Shield of Rhode Island, Blue Cross and Blue Shield of South Carolina, BlueChoice HealthPlan of South Carolina, Inc. f/k/a Companion HealthCare Corporation, Planned Administrators, Inc., Thomas H. Cooper & Co., Inc., BlueCross BlueShield of Tennessee, Inc., Tennessee Health Care Network, Inc., Health Care Service Corporation, Group Health Maintenance Organization, Inc. d/b/a Bluelincs HMO, Group Health Service of Oklahoma, Inc. d/b/a Blue Cross Blue Shield of Oklahoma, HMO New Mexico, Inc., New Mexico Blue Cross and Blue Shield, Inc. d/b/a Blue Cross and Blue Shield of New Mexico, New Mexico Blue Cross and Blue Shield, Inc., Hallmark Services Corporation, BCI HMO, Inc., HCSC Insurance Services Corporation (collectively, the "Blue Parties"), and the other Released Parties, the action styled as

Love, et al. v. Blue Cross and Blue Shield Ass'n, et al., (formerly, Thomas, et al. v. Blue Cross and Blue Shield Ass'n, et al.), Case No. 03-21296-CIV-MORENO/SIMONTON (the "Action") is conditionally certified as a class action on behalf of the following persons (collectively, the "Class" and each member of the Class a "Class Member"):

> any and all Physicians, Physician Groups and Physician Organizations who provided Covered Services to any Plan Member or services to any individual enrolled in or covered by a Plan offered or administered by any Person named as a defendant in the Complaint or by any other primary licensee of the BCBSA or by any of their respective current or former subsidiaries or Affiliates, in each case from May 22, 1999 through the Preliminary Approval Date. The Class shall exclude: (i) all Persons who, in accordance with the terms of this Agreement, execute a timely request for exclusion (Opt-Out) from the Class; and (ii) the Blue Parties, their Affiliates and any of their officers, directors and employees.

Representative Plaintiffs are conditionally appointed as representatives of the Class. Class Counsel are appointed and designated as counsel for the Class. This conditional certification of the Class appointment of the Representative Plaintiffs as representatives of the Class and appointment and designation of Class Counsel as counsel for the Class are solely for purposes of effectuating the Settlement. If the Settlement Agreement is terminated or is not consummated for any reason, the foregoing conditional certification of the Class, appointment of the Representative Plaintiffs as representatives of the Class, and appointment and designation of Class Counsel as counsel for the Class shall be void and of no effect and the Parties to the Settlement Agreement and/or any Blue Party terminating its participation in the Settlement Agreement in accordance with its terms, shall be returned to the status each occupied before entry of this Order without prejudice to any legal argument that any of the Parties to the Settlement Agreement might have asserted but for the Settlement Agreement.

Page 5 of 14

Based on the Court's review of the Joint Motion and supporting materials, the Court conditionally finds that the proposed Class satisfies Rule 23(a), 23(b)(2), and 23(b)(3) of the Federal Rules of Civil Procedure.

Settlement Hearing; Right to Appear and Object

- A Settlement Hearing shall take place before the Court on [DATE], at [TIME] in 3. the United States Courthouse, Courtroom IV, Tenth Floor, Federal Justice Building, 99 Northeast Fourth Street, Miami, Florida, 33132 to determine:
- whether the Court should certify the Class and whether the Court should (a) find the Representative Plaintiffs and Class Counsel as adequate representatives of the Class and counsel for the Class, respectively;
- (b) whether the Settlement, on the terms and conditions provided for in the Settlement Agreement, should be finally approved by the Court as fair, reasonable, and adequate;
- whether the Released Claims of the Class Members and Signatory Medical (c) Societies asserted in the Action should be dismissed on the merits and with prejudice as to the Blue Parties;
- whether the Court should permanently enjoin the assertion of any Claims (d) that arise from or relate to the subject matter of the Action against Released Parties by the Class Members or any other Persons;
- whether the application for attorneys' fees, costs, and expenses to be (e) submitted by Class Counsel in connection with the Settlement Hearing should be approved;
- whether the application for payments to Representative Plaintiffs and (f) Representative Plaintiffs in Other Actions to be submitted in connection with the Settlement Hearing should be approved;

Page 6 of 14

- (g) whether each Tag-Along Action should be dismissed with prejudice as to Released Claims against the Blue Parties, with the exception of any Tag-Along Action with respect to any named plaintiff who or which has timely submitted an Opt-Out request;
- (h) whether the Court should permanently enjoin the assertion of Claims by the Releasing Parties that arise from or relate to conduct by the Released Parties that occurred on or before the Effective Date of the Settlement Agreement, except for Claims for damages against Defendants in the Action that are not Parties, relating to providing Covered Services to Blue Plans' Plan Members in connection with the BlueCard Program and other similar national account delivery programs governed by BCBSA (including but not limited to NASCO-to-NASCO arrangements; and
 - (i) such other matters as the Court may deem necessary or appropriate.
- 4. The Court may finally approve the Settlement Agreement at or after the Settlement Hearing with any appropriate modifications agreed to by the Parties and without further notice to the Class Members.
- 5. Any putative Class Member who or which has not timely and properly provided notice of an election to Opt-Out of the Class and the Settlement Agreement in the manner set forth below, and any other interested Person, may appear at the Settlement Hearing in person or by counsel and be heard, to the extent allowed by the Court, either in support of or in opposition to the matters to be considered at the Settlement Hearing, provided, however, that no Person who or which has elected to Opt-Out of the Class and the Settlement Agreement shall be entitled to object; and provided further that no Person shall be heard, and no papers, briefs, or other submissions shall be considered by the Court in connection with the Court's consideration of those matters, unless on or before a date seventy five (75) days after the Notice Date, such Person:

- (a) files with the Court a written statement of objection setting forth such

 Person's objections to the matters to be considered and the basis therefore, together with all other support, papers, or briefs that he, she, or it wishes the Court to consider. If any Person making an objection wishes to appear at the Settlement Hearing, they must also file a written notice of intention to appear; and
- (b) serve copies of all such materials either by hand or overnight delivery upon the following counsel:

Edith M. Kallas Joe R. Whatley, Jr. Joseph P. Guglielmo WHATLEY DRAKE & KALLAS LLC 1540 Broadway, 37th Floor New York, New York 10036

Archie Lamb, Jr. LAW OFFICES OF ARCHIE LAMB, LLC 2017 Second Avenue North, 2nd Floor Birmingham, Alabama 35203

Harley S. Tropin Kozyak Tropin & Throckmorton, P.A. 2525 Ponce de Leon Blvd., 9th Floor Miami, Florida 33134

And

Michael A. Pope MCDERMOTT WILL & EMERY LLP 227 W. Monroe Street Chicago, IL 60606

Chester T. Kamin JENNER & BLOCK LLP 330 North Wabash Chicago, IL 60611

All responses by the Parties to objections shall also be served by hand or overnight delivery on the objecting Person or his, her, or its attorney no later than fifteen (15) days before the Settlement Hearing. Objectors' papers may be supplemented ten (10) days prior to the Settlement Hearing

following the filing of any responsive papers by the Parties and served in the manner and on counsel described above.

6. The Court may adjourn the Settlement Hearing, or any adjournment thereof, including the consideration of the application for attorneys' fees, costs and expenses, without any additional notice to putative Class Members other than an announcement in open court.

Form and Timing of Notice

- 7. No earlier than thirty (30) days and no later than sixty (60) days after entry of this Order, Class Counsel and the Blue Parties through the Notice Administrator shall cause copies of the Notice of Class Action and Proposed Settlement in the form attached as Exhibit E to the Settlement Agreement (the "Mailed Notice"), the form of which is hereby approved, and in the case of Class Members who are individuals, including the claim form substantially in the form attached as Exhibit A to the Settlement Agreement (the "Claim Form"), the form of which is hereby approved, to be mailed by first-class mail, postage pre-paid, to all putative Class Members who can be identified with reasonable diligence from (a) solely in the Blue Plans' discretion, the Blue Plans' books and records, at their last known address appearing in such books and records, and/or (b) records maintained by the American Medical Association (the "AMA") and/or a vendor associated therewith, to the extent that such records can be obtained for this purpose within the schedule for notice provided in this Order.
- 8. No earlier than thirty (30) days and no later than sixty (60) days after entry of this Order, Class Counsel and the Blue Parties through the Notice Administrator shall cause to be published three times in the legal notices sections of the national editions of *The Wall Street Journal* and *USA Today*, a summary notice substantially in the form attached as Exhibit G to the Settlement Agreement (the "Published Notice"), the form of which is hereby approved. To the

extent reasonably feasible within the schedule contemplated by this Order, Class Counsel and the Blue Parties through the Notice Administrator shall cause the Published Notice to be published in a nationwide periodical addressing issues of concern to physicians, such as The Journal of the American Medical Association or The American Medical News. If publication in one or more said publications on the foregoing schedule is not practicable, then either Class Counsel or the Blue Parties may apply to the Court for alternative notice by publication. In addition, no later than thirty (30) days prior to the Opt-Out Deadline and continuing through the Final Order Date, each Blue Plan shall publish the Mailed Notice on its Public Website.

- 9. Beginning on the earliest date that notice is provided pursuant to paragraphs 6 and 7 of this Order and continuing through the deadline for putative Class Members to return a completed Claim Form (as that date is specified in the Settlement Agreement), Blue Parties and Class Counsel shall make available to putative Class Members printable versions of the Mailed Notice and the Claim Form on the public websites maintained by or on behalf of the Blue Plans and at least three of the law firms serving as Class Counsel in the Action.
- 10. Prior to the Settlement Hearing, Class Counsel and the Blue Parties shall serve and file a sworn statement attesting to compliance with paragraphs 6, 7, and 8 of this Order. Costs of providing the notice to the Class that is specified in this Order shall be paid as set forth in the Settlement Agreement.
- 11. The notice to be provided as set forth in paragraphs 6, 7, and 8 of this Order is hereby found to be the best means of notice to putative Class Members that is practicable under the circumstances and, when completed, shall constitute due and sufficient notice of the Settlement, the Settlement Agreement and the Settlement Hearing to all Persons affected by and/or entitled to

Page 10 of 14

participate in the Settlement, the Settlement Agreement or the Settlement Hearing, in full compliance with the requirements of due process and the Federal Rules of Civil Procedure.

12. All notices to any Party (including without limitation any designations made by Class Counsel pursuant to the Settlement Agreement) required under the Settlement Agreement shall be sent by first class U.S. Mail, by hand delivery, or by facsimile, to the recipients designated in the Settlement Agreement. Unless otherwise specifically stated, timeliness of all submissions and notices shall be measured by the date of the postmark. The Persons designated to receive notices under the Settlement Agreement are as follows (unless the addressee refuses or intentionally or willfully delays receipt, or unless notification of any change to such designated is given to each other Party to the Settlement Agreement pursuant to Section 5 of the Settlement Agreement).

Representative Plaintiffs and Signatory Medical Societies: Notice to be given to Class Counsel on behalf of Representative Plaintiffs and Signatory Medical Societies.

Class Counsel:

Edith M. Kallas Joe R. Whatley, Jr. Joseph P. Guglielmo WHATLEY DRAKE & KALLAS LLC 1540 Broadway, 37th Floor New York, New York 10036

Archie Lamb, Jr. LAW OFFICES OF ARCHIE LAMB, LLC 2017 Second Avenue North, 2nd Floor Birmingham, Alabama 35203

Harley S. Tropin Kozyak Tropin & Throckmorton, P.A. 2525 Ponce de Leon Blvd., 9th Floor Miami, Florida 33134

Page 11 of 14

Blue Parties:

Michael A. Pope MCDERMOTT WILL & EMERY LLP 227 W. Monroe Street Chicago, IL 60606

Chester T. Kamin JENNER & BLOCK LLP 330 North Wabash Chicago, IL 60611

Ability of Putative Class Members to Opt-Out of Settlement Class

- 13. All putative Class Members who or which wish to Opt-Out of the Class and the Settlement must do so by sending written notice of their election to Opt-Out to the Notice Administrator at the address and in the manner set forth in the notices to be provided pursuant to paragraphs 6, 7, and 8 of this Order. To be considered proper and timely, a complete, signed Opt-Out request must be postmarked by no later than 60 days after the Notice Date (the "Opt-Out Deadline"). Prior to the Settlement Hearing, Class Counsel or their designee shall submit to the Court a sworn statement setting forth the names and addresses of each putative Class Member who or which has properly and timely elected to Opt-Out of the Class and the Settlement.
- 14. Any putative Class Member who or which does not properly and timely Opt-Out of the Class shall be included in such Class and, if the Settlement is approved and becomes effective, shall be bound by all the terms and provisions of the Settlement Agreement, including but not limited to the releases, waivers, and covenants not to sue described therein, whether or not such Person shall have objected to the Settlement and whether or not such Person makes a claim upon, or participates in, the Settlement Fund or the other benefits to the Class to be provided under the Settlement Agreement.

Other Provisions

- All discovery and any other proceedings in the Action against or concerning the 15. Released Parties in the Action, except with respect to discovery from the BCBSA as set forth in Exhibit O to the Settlement Agreement, and other than proceedings as may be necessary to carry out the terms and conditions of the Settlement, are hereby stayed and suspended until further order of the Court. Pending final determination of whether the Settlement and the Settlement Agreement should be finally approved and the Class permanently certified, all putative Class Members and Signatory Medical Societies are hereby barred and enjoined from commencing or prosecuting any action asserting any Released Claims, and any other action or proceedings brought by any putative Class Members or Signatory Medical Societies asserting any Released Claims are hereby stayed and suspended until further order of the Court.
- 16. All discovery and any other proceedings against or concerning the Blue Parties in Tag-Along Actions asserting any Released Claim, other than proceedings as may be necessary to carry out the terms and conditions of the Settlement, are hereby stayed and suspended until further order of the Court.
- 17. No discovery with regard to the Settlement or the Settlement Agreement shall be permitted other than as may be directed by the Court upon a proper showing by the Person seeking such discovery by motion properly noticed and served in accordance with this Court's Local Rule 7.1.
- 18. Pending entry of an Order setting forth the final determination of the Court with respect to the Parties' application for final approval of the Settlement, the Released Parties are hereby relieved from the date of this Order and going forward, of any and all obligations to preserve documents and evidence with respect to Released Claims, including any obligations to

preserve Future Documents under the Agreed Order For Preservation of Records entered on April 27, 2006 in this Action, or any similar obligations to preserve future documents relating to Released Claims in any Tag-Along Action. This Order does not relieve any Party from their obligations under the Agreed Order for Preservation of Records as entered in and applicable to Solomon, et al. v. Blue Cross and Blue Shield Ass'n, et al., Case No. 03-22935-CIV-MORENO/SIMONTON.

- 19. The Blue Parties are authorized to communicate with putative Class Members and Signatory Medical Societies regarding the provisions of the Settlement Agreement (including, but limited to, the Blue Parties' right to withhold all Settlement benefits from putative Class Members who or which Opt-Out of the Class and the Settlement) so long as such communications are not inconsistent with the agreed upon communications concerning the Settlement Agreement. The Blue Parties shall refer to the Settlement Administrator or to Class Counsel any inquiries from Class Members about claims to be filed under the Settlement Agreement.
- 20. Any putative Class Member may enter an individual appearance in this Action, at his, her, or its own expense. Any putative Class Member who or which does not enter an individual appearance will be represented by Class Counsel.
- 21. Neither the Settlement, nor the Settlement Agreement nor any provision therein, nor any negotiations, statements, or proceedings in connection therewith shall be construed as, or be deemed to be evidence of, an admission or concession on the part of any of the Representative Plaintiffs, the Signatory Medical Societies, Class Counsel, any putative Class Members, the Blue Parties, or any other Person of any liability or wrongdoing by them, or that the claims and defenses that have been, or could have been, asserted in the Action were or were not meritorious, and neither the Settlement, the Settlement Agreement, nor any such communications shall be offered

or received in evidence in any action or proceeding, or be used in any way as an admission or concession or evidence of any liability or wrongdoing of any nature or that Representative Plaintiffs, the Signatory Medical Societies, any Class Member, the Blue Parties, or any other Person has or has not suffered any damage.

22. In the event that the Settlement Agreement is terminated by Class Counsel or by all Blue Parties or is not consummated for any reason, the Settlement and all proceedings had in connection therewith shall be null and void, except to the extent expressly provided to the contrary in the Settlement Agreement, and without prejudice to the rights of the Parties to the Settlement Agreement before it was executed. In addition, in the event any individual Blue Party terminates its participation in the Settlement Agreement in accordance with its terms and provisions, then this Order and all orders entered in connection therewith, shall be rendered null and void as to the terminating Blue Party.

DONE AND ORDERED after a nearing	g in open court at the United	1 States District
Courthouse in Miami, Florida, on	, and signed this	day of
, 2007.		
7		

HON. FEDERICO A. MORENO UNITED STATES DISTRICT JUDGE

EXHIBIT G

Page 1 of 3

UNITED STATES DISTRICT COURT FOR THE SOUTHERN DISTRICT OF FLORIDA MIAMI DIVISION

RICK LOVE, M.D., et al.,)
Plaintiffs)) Case No. 03-21296-CIV-) MORENO/SIMONTON
v.)
BLUE CROSS AND BLUE SHIELD ASSOCIATION, et al.,))
Defendants)
	,)

NOTICE OF PROPOSED SETTLEMENT WITH CERTAIN BLUE CROSS AND BLUE SHIELD PARTIES

IF YOU ARE A PHYSICIAN, PHYSICIAN GROUP, OR PHYSICIAN ORGANIZATION WHO PROVIDED COVERED SERVICES TO ANY INDIVIDUAL ENROLLED IN OR COVERED BY BLUE CROSS AND BLUE SHIELD PLANS AT ANY TIME BETWEEN MAY 22, 1999 AND [PRELIMINARY APPROVAL DATE], PLEASE READ THIS NOTICE CAREFULLY. THE CLASS ACTION AND THE PROPOSED SETTLEMENT MAY AFFECT YOUR RIGHTS.

If you are or have been a Physician, Physician Group, or Physician Organization I/ who or which practiced in the United States since May 22, 1999, your rights may be affected by a proposed settlement with certain Blue Cross and Blue Shield parties ("Settlement"), in the class action lawsuit known as Love, et al. v. Blue Cross and Blue Shield Ass'n, et al., (formerly, Thomas, et al. v. Blue Cross and Blue Shield Ass'n, et al.), Case No. 03-21296-CIV-MORENO/SIMONTON (the "Action"), which is pending in the U.S. District Court for the Southern District of Florida (the "Court"). The Representative Plaintiffs have agreed to settle all claims against the Blue Cross and Blue Shield Association ("BCBSA") and certain Blue Cross and Blue Shield Plans and certain current and former Subsidiaries and Affiliates (collectively, the "Blue Parties") in the Action in exchange for the Blue Parties' commitments regarding their business practices and the establishment of a Settlement Fund with respect to which Class Members (described below) can make claims for settlement payment. The Court has scheduled a hearing to consider the fairness, reasonableness, and adequacy of the proposed Settlement with the Blue Parties. together with If you are or have been a Physician, Physician Group, or Physician Organization 1/2 who or which reasonableness, and adequacy of the proposed Settlement with the Blue Parties, together with certain other matters, to be held on [DATE], at [TIME], at the United States Courthouse, Courtroom IV, Tenth Floor, Federal Justice Building, 99 Northeast Fourth Street, Miami, Florida, 33132 (the "Settlement Hearing"). You may be a Class Member who would be entitled to receive the benefits of the proposed Settlement. As a Class Member, however, you will also be bound by the release and other provisions of the Settlement if it is approved by the Court. You may elect to Opt-Out of the Class and the Settlement, as explained below. You may also have a right to object to the Settlement or to the applications for attorneys' fees and Representative Plaintiffs' fees that

^{1/} All capitalized terms in this notice are defined in the Settlement Agreement.

Class Counsel intend to make to the Court, but only if you comply with the procedures described or referenced in this notice.

WHAT IS THE LITIGATION ABOUT?

The Complaint in the Action alleges, among other things, that between 1999 and the present, the Blue Parties, among others, engaged in a conspiracy to improperly deny, delay, and/or reduce payment to physicians, physician groups, and physician organizations by engaging in several types of allegedly improper conduct. The Complaint in the Action claims that this conduct violated the federal statute entitled the Racketeer Influenced and Corrupt Organizations Act, 18 U.S.C. § 1961, et seq. The Blue Parties deny the factual allegations and legal claims asserted in the Complaint in the Action, and deny any liability or wrongdoing relating thereto.

WHO IS INVOLVED?

The proposed Settlement with the Blue Parties will be on behalf of a settlement class (collectively the "Class" and each member of the Class a "Class Member") consisting of any and all Physicians, Physician Groups, and Physician Organizations who provided Covered Services to any Plan Member or services to any individual enrolled in or covered by a Plan offered or administered by any Person named as a defendant in the Complaint or by any other primary licensee of the BCBSA or by any of their respective current or former subsidiaries or Affiliates, in each case from May 22, 1999 through [Preliminary Approval Date]. The Class shall exclude: (i) all Persons who, in accordance with the terms of the Settlement, execute a timely request for exclusion (Opt-Out) from the Class; and (ii) the Blue Parties, their Affiliates, and any of their officers, directors, and employees.

THE PROPOSED SETTLEMENT

In a settlement agreement dated April 27, 2007 (the "Settlement Agreement"), the Blue Parties have made commitments regarding their business practices. In addition, the Blue Parties will fund a Settlement Fund with respect to which individual Physicians who are Class Members will be able to make claims for a settlement payment according to a formula that is set forth in the Settlement Agreement. This settlement consideration is described more fully in a longer notice that has been mailed directly to putative Class Members.

YOUR RIGHTS

If you want to participate in the proposed Settlement Fund, your completed, signed Claim Form must be mailed to the Settlement Administrator with a postmark NO LATER THAN [90 DAYS AFTER NOTICE DATE]. The Claim Form, the address to which it must be sent, and other details can be obtained at [Insert Settlement Administrator Website], www.hmosettlements.com, <a h

☐ If you want to be excluded from the proposed Settlement, you must submit a written request to Opt-Out of the Class and the Settlement, including your name, business address, telephone number, Federal Tax Identification Number, and signature, to:

[NOTICE ADMINISTRATOR CONTACT INFO.]

The complete, signed Opt-Out request must be mailed to the Notice Administrator with a postmark NO LATER THAN [60 DAYS AFTER NOTICE DATE].

- The Court has scheduled a Settlement Hearing to consider the proposed Settlement and other matters, which will take place on [DATE], at [TIME], at the United States Courthouse, Courthouse IV, Tenth Floor, Federal Justice Building, 99 Northeast Fourth Street, Miami, Florida, 33132. At that hearing, the Court will determine, among other things, whether the proposed Settlement with the Blue Parties should be approved as fair, reasonable, and adequate, whether the proposed Class should be certified for settlement purposes, whether the Court should award attorneys' fees and expenses to Class Counsel or to the Representative Plaintiffs in the Action and to Representative Plaintiffs in Other Actions, and certain other matters. You may attend this hearing if you wish, but you do not have to attend in order to participate in the proposed Settlement.
- If you wish to appear at the Settlement Hearing or to object to the proposed Settlement, you must file and serve, NO LATER THAN [75 DAYS AFTER NOTICE DATE], a written objection, and a notice of intention to appear at the Settlement Hearing, if applicable, in accordance with the procedures and deadlines that are described in the longer notice that has been mailed directly to putative Class Members. If you have not received that longer notice, you should request it from one of the law firms identified below, or obtain it from the website listed below. If these procedures and deadlines are not followed, you may lose significant legal rights, including, but not limited to, the right to have your objections considered by the Court.

For Complete Information and a Copy of the Full Notice:

Call:		

Or Write to:

Edith Kallas, Esq. WHATLEY DRAKE & KALLAS LLC 1540 Broadway, 37th Floor New York, NY 10036

Archie C. Lamb, Jr. Law Offices of Archie C. Lamb, L.L.C. 2017 Second Avenue North Birmingham, AL 35203

Harley S. Tropin Kozyak Tropin & Throckmorton, PA 2525 Ponce de Leon Blvd., 9th Floor Miami, FL 33134

Or visit:

[Settlement Administrator Website]

www.hmosettlements.com www.WhatleyDrake.com www.kttlaw.com www.ArchieLamb.com

Please do not contact the Court directly, or the Blue Parties, or a Blue Plan's Provider Relations Representative.

Page 1 of 18

Exhibit H Party Specific Provisions

H.1. § 7.8 Disclosure of and Commitments Concerning Claims Payment Practices

- (a) The Blue Plan Empire Blue Cross and Blue Shield (hereinafter as used in this Exhibit H "Empire") shall not be subject to the requirements of § 7.8(b), regarding consistency across product lines, solely with respect to its products in New Jersey; provided, however, that it is the intent of Empire to terminate its products in New Jersey as of the end of first quarter 2008; and provided, further, if such products are not terminated as of such date, and new Plan Members are added, the obligations under § 7.8(b) shall apply thereafter, and continue to apply for a period beyond the Termination Date equal to the time between the date on which the obligation was to commence (under the applicable Implementation Date), and the date when it is actually implemented.
- (b) The Blue Plan Blue Cross Blue Shield of North Carolina (hereinafter as used in this Exhibit H "BCBSNC") recognizes the benefits of greater transparency with respect to automated "rebundling" and other claims payment rules, and to that end, currently makes available on its Provider Website, McKesson's Clear Claim Connection ("C3") software, a web-based code auditing disclosure tool. BCBSNC may elect to satisfy the disclosure obligations set forth in § 7.8(d) of the Agreement regarding "Significant Edits" by maintaining the C3 software on its Provider Website during the term of the Agreement. Notwithstanding this provision, BCBSNC will publish and annually update in a location that can be accessed by both Participating and Non-Participating Physicians a searchable list of Significant Edits that are expected to result in denial or reduction in payment for a particular CPT® Code or HCPCS Level II Code more than two thousand (2000) times per year.
- (c) The General Counsel of the North Carolina Medical Society (hereinafter as used in this Exhibit H "NCMS") shall have access to BCBSNC's Provider Website for the purpose of monitoring issues related to the implementation of this Settlement Agreement. Access to BCBSNC's Provider Website, and any information derived therefrom, may be used by the NCMS solely for the purpose identified herein, and may not be used or disclosed for any other purpose.
- (d) BCBSNC is pursuing actions reasonably designed to allow the electronic submission of Clinical Information by physicians, will continue to work on this initiative throughout 2007, and anticipates beginning implementation of this initiative by June 1, 2008.

H.2. § 7.9 Physician Advisory Committee

(a) The Blue Plan Louisiana Health Service & Indemnity Company d/b/a Blue Cross and Blue Shield of Louisiana (hereinafter as used in this Exhibit H "BCBSLA") shall be permitted to continue its existing physician advisory committee activities and shall not be subject to the provisions of § 7.9 of the Agreement.

- The Blue Plan Blue Cross Blue Shield of Massachusetts (hereinafter as used in this (b) Exhibit H "BCBSMA") shall be permitted to continue its existing physician advisory committee activities and shall not be subject to the provisions of § 7.9 of the Agreement.
- The Blue Plan BCBSM, Inc. dba BlueCross BlueShield of Minnesota (hereinafter as used in this Exhibit H "BCBSMN") shall be permitted to continue its existing physician advisory committee activities and shall not be subject to the provisions of § 7.9 of the Agreement.
- The Blue Plan Blue Cross & Blue Shield of Mississippi (hereinafter as used in this (d) Exhibit H "BCBSMS") shall be permitted to continue its existing physician advisory committee activities and shall not be subject to the provisions of § 7.9 of the Agreement.
- The Blue Plan The Regence Group (including Regence BlueCross BlueShield of Oregon, Regence BlueShield, Regence BlueCross BlueShield of Utah, and Regence BlueShield of Idaho, Inc.) (hereinafter as used in this Exhibit H "Regence") will comply with § 7.9, except as follows:
 - Regence BlueShield of Idaho shall be permitted to continue its existing physician (1)advisory committee activities and shall not be subject to the provisions of § 7.9 of the Agreement; and
 - Regence will continue to negotiate with Class Counsel for thirty (30) days after the Execution Date regarding the physician advisory committee in its Washington service area. Regence will negotiate with the respective state medical societies in Oregon and Utah for thirty (30) days after the Execution Date regarding the physician advisory committees in its Oregon and Utah service areas. In the event that Regence reaches agreements regarding its Washington, Oregon and Utah service areas, then the provisions of § 7.9 shall not apply to Regence. In the event such agreements are not reached, the provisions of § 7.9 shall apply to Regence.
- The Blue Plan Blue Cross and Blue Shield of South Carolina (hereinafter as used in this (f) Exhibit H "BCBSSC") shall be permitted to continue its existing physician advisory committee activities and shall not be subject to the provisions of § 7.9 of the Agreement.
- The Blue Plan Blue Cross and Blue Shield of Alabama (hereinafter as used in this Exhibit (g) H "BCBS Ala") shall not be subject to the provisions of § 7.9 of the Agreement and shall instead be subject to the following provisions:
 - In order to provide Physicians in the State of Alabama the opportunity to provide meaningful advice with respect to the policies of BCBS Ala that affect physicians in the state, BCBS Ala will continue to maintain and implement the following practices and initiatives that provide meaningful opportunity for input to matters impacting upon the practice of medicine and provision of care to Plan Members:
 - BCBS Ala currently has a physician advisory committee ("Preferred Medical Doctor Physicians' Advisory Committee") under which Participating Physicians in its Preferred Medical Doctor ("PMD") Program elect seven (7) practicing physicians from within Alabama to provide advice to BCBS Ala. The current Preferred Medical Doctor

Physicians' Advisory Committee structure, election process and operating guidelines and procedures, as established and amended from time to time by the BCBS Ala Preferred Medical Doctor Physicians' Advisory Committee, will continue to be maintained;

- BCBS Ala will continue to meet with elected officials and established committees of the Medical Association of Alabama ("MASA"), on a periodic basis to discuss and seek input and comment upon BCBS Ala's various policies, initiatives and matters impacting upon medical practice and provision of care;
- BCBS Ala will continue to publish on its Provider Website proposed new medical policies and proposed revisions to existing medical policies for advance notice and comment by Participating Physicians in its PMD Program and will continue to assess and evaluate all comments received thereon prior to finalizing such medical policies. Upon request, BCBS Ala will meet or confer with those representatives of Alabama specialty societies identified in (4) below regarding new medical policies or changes in medical policies; and
- BCBS Ala will meet twice a year with the Presidents and Presidents-Elect or immediate past Presidents of the Alabama specialty societies for (i) ENTs, (ii) one surgical specialty (the surgical specialty shall change each year and rotate among the various surgical specialties), (iii) Cardiologists, (iv) Orthopedics, and (v) others that BCBS Ala determines to have a significant number of members to discuss proposed policies and other issues and initiatives that are important to those specialties.
- In the event that the Blue Plan Wellmark, Inc. d/b/a Wellmark Blue Cross and Blue (h) Shield of Iowa (hereinafter as used in this Exhibit H "Wellmark") and the Iowa Medical Society agree within sixty (60) days after the Execution Date on the method for selecting members of the Wellmark physician advisory committee, the Blue Plan shall be permitted to continue its existing physician advisory committee activities, except as modified by such agreement with the Iowa Medical Society, and shall not be subject to the provisions of § 7.9 of the Agreement. In the event such an agreement is not reached within 60 days after the Execution Date, the provisions of § 7.9 of the Agreement shall thereafter apply to Wellmark with respect to the Iowa service area.
- (i) In the event that the Blue Plan Wellmark of South Dakota, Inc. d/b/a Wellmark Blue Cross and Blue Shield of South Dakota (hereinafter as used in this Exhibit H "Wellmark South Dakota"), and the South Dakota State Medical Association agree within sixty (60) days after the Execution Date on the method for selecting members of the Wellmark South Dakota physician advisory committee, Wellmark South Dakota shall be permitted to continue its existing physician advisory committee activities, except as modified by such agreement with the South Dakota State Medical Association, and shall not be subject to the provisions of § 7.9 of the Agreement. In the event such an agreement is not reached within 60 days after the Execution Date, the provisions of § 7.9 of the Agreement shall thereafter apply to Wellmark South Dakota.
- The Blue Plan Hospital Service Association of Northeastern Pennsylvania (hereinafter as (i) used in this Exhibit H "BCNEPA") shall not be subject to the provisions of § 7.9 of the Agreement. BCNEPA will continue with its existing Quality Improvement Committee, subject

Page 4 of 18

to the following provisions regarding its size and composition: BCNEPA will select four (4) Participating Physician members and the Pennsylvania Medical Society will select four (4) Participating Physician members. Those eight (8) members will select any remaining members not to exceed four (4) in number. The Pennsylvania Medical Society will choose physicians who practice in BCNEPA's service area. If BCNEPA decides not to accept a recommendation of the Quality Improvement Committee, BCNEPA shall communicate that decision in writing to the Committee with an explanation of BCNEPA's reasons. BCNEPA agrees to post on its Provider Website a listing of all Quality Improvement Committee recommendations made to BCNEPA and BCNEPA's responses to such recommendations.

(k) With respect to the Blue Plan Health Care Service Corporation (as used hereinafter in this Exhibit H "HCSC"), the fourth sentence of § 7.9(b) of the Agreement is deleted and is replaced by the following:

HCSC shall establish a mechanism by which the state medical societies in its service area shall select four (4) members of HSCS's Physician Advisory Committee with consultation, as each state society deems necessary, with appropriate organized medical and physician professional organizations.

- (I) In lieu of Empire's compliance with § 7.9, Empire and the Medical Society of New York each may add a representative to the Physician Advisory Committee created by the settlement agreement between WellPoint, Inc. and the representative plaintiffs named in the actions styled Shane, et al. v. Humana, Inc., et al., Master File No. 00-1334-MD-Moreno and Shane, et al. v. Humana, Inc., et al., Case No. 04-21589 Civ-Moreno, as well as this Action (the "Wellpoint Settlement Agreement").
- (m) BCBCNC will comply with the obligations under § 7.9, including (1) a summary of proceedings of such meetings and recommendations approved by its members shall be posted on the BCBSNC website, and (2) members shall not be prohibited from communicating with their colleagues and medical associations regarding any information or data disclosed at the meetings not identified by BCBSNC as proprietary.
- (n) The Blue Plan Horizon Blue Cross Blue Shield of New Jersey (hereinafter as used in this Exhibit H "Horizon") shall cause the chairman of the Physician Advisory Committee to propose that the Physician Advisory Committee consider the issues relating to (1) improving the time for making payments to physicians where a delay relates to obtaining information from treating physicians regarding possible pre-existing conditions of Plan Members and coordination of benefits; and (2) improving the time needed to make system adjustments to assure that payments to Plan Members for services provided by non-participating physicians are at the appropriate rates after a physician has left Horizon's physician networks.
- (o) With respect to the Blue Plan CareFirst, Inc., CareFirst of Maryland, Inc. and Group Hospitalization and Medical Services, Inc. (hereinafter as used in this Exhibit H collectively "CareFirst"), § 7.9 is amended by replacing the phrase "state medical society" with the phrase state medical society and the Northern Virginia Medical Societies.

Page 5 of 18

- With respect to the Blue Plan Triple-S, Inc. (hereinafter as used in this Exhibit H "Triple-(p) S"), the term "state medical society" shall refer to Colegio De Medicos Y Cirujanos De Puerto Rico.
- The Blue Plan Blue Cross Blue Shield of Michigan (hereinafter as used in this Exhibit H "BCBSMI") shall be permitted to continue its existing physician advisory committee activities and shall not be subject to the provisions of § 7.9 of the Agreement.

H.3. § 7.10 New Dispute Resolution Process for Physician Billing Disputes

Document 928-15

- The requirements of § 7.10 with regard to external review shall not apply to the Blue Plan (a) Blue Cross Blue Shield of Montana (hereinafter as used in this Exhibit H "BCBSMT") with respect to Plan Members enrolled under the Montana state employees health benefits program.
- (b) Horizon agrees that, to the extent that it utilizes a billing dispute external review process and organization provided by New Jersey state law to adjudicate billing disputes with physicians, Horizon will comply with the provisions of § 7.10.
- With respect to the Blue Plan BlueCross BlueShield of Tennessee, Inc. (hereinafter as (c) used in this Exhibit H "BCBSTN"), the first sentence of § 7.10(c) of the Agreement is deleted and is replaced by the following:

BCBSTN agrees that any individual Physician or Physician Group may submit a Billing Dispute to the Billing Dispute External Reviewer after the Physician or Physician Group exhausts BCBSTN's internal appeals process, and when the amount in dispute exceeds \$350.

- (d) The Blue Plan Premera Blue Cross (hereinafter as used in this Exhibit H "Premera") confirms that, under § 7.10 of the Agreement, an individual Physician, Physician Group or Physician Organization practicing in the State of Washington has the option of using either § 7.10 of the Agreement or pursuing a judicial remedy. Premera will not take the position that it is unable to comply with § 7.10 pursuant to Washington state law.
- Regence confirms that, under § 7.10 of the Agreement, an individual Physician, (e) Physician Group or Physician Organization practicing in the State of Washington has the option of using either § 7.10 of the Agreement or pursuing a judicial remedy. Regence will not take the position that it is unable to comply with § 7.10 pursuant to Washington state law.
- BCBSNC confirms that a Physician will not be required to obtain Plan Member (f) authorization in order to pursue internal appeals of payment disputes in patient-specific factual situations. This provision does not apply to appeals of medical necessity or experimental and investigational determinations covered by § 7.11 or disputes related to Clinical Information submission requirements.

H.4. § 7.11 Determinations Related to Medical Necessity

- (a) The requirements of § 7.11 with regard to external review as to BCBSMT shall not apply to Plan Members enrolled under the Montana state employees health benefits program.
- (b) Horizon agrees that, to the extent that it utilizes a medical review external review process and organization provided by New Jersey state law to make determinations relating to whether proposed health care services or supplies are Medically Necessary or experimental or investigational in nature, Horizon will comply with the provisions of § 7.11.
- (c) CareFirst shall not be obligated to comply with the requirements of § 7.11 with respect to the Maryland state employee health benefits program.

H.5. § 7.13(a) Credentialing

The Blue Plan Hawaii Medical Service Association (hereinafter as used in this Exhibit H "HMSA") is subject to the following amendment to the second sentence of § 7.13(a):

HMSA shall complete primary source verification and notify the Physician as to whether he or she is credentialed within thirty-five (35) days of receiving a Physician's completed application to be a Participating Physician unless in spite of HMSA's best efforts and because of a failure of a third party to provide necessary documentation, HMSA cannot obtain the necessary information to make a decision within thirty-five (35) days. The thirty-five day period shall be tolled for: 1) any time period HMSA is waiting on information from the Physician; and 2) the time period primary source verification is pending with CVS.

H.6. § 7.13(b) All Products Clauses

- (a) BCBSNC shall comply with § 7.13(b), except with respect to Medicare Advantage where: (1) it can be demonstrated that physicians have been unwilling to contract for the BCBSNC Medicare Advantage plan after good faith efforts to establish such a network; and (2) BCBSNC is subject to a competitive disadvantage as a result.
- (b) With respect to the Blue Plan Blue Cross Blue Shield of Rhode Island (hereinafter as used in this Exhibit H "BCBSRI"), § 7.13(b) of the Agreement is amended by adding the following sentence at the end of the section:

Nothing in § 7.13(b) of the Agreement shall be construed to alter BCBSRI's obligation to comply with R.I. A.D.C. 15 020.006, Section 0348.65.05.

H.7. § 7.14(b) Payment Rules for Injectibles, DME, Administration of Vaccines and **Review of New Technologies**

- BCBSNC will pay at least the average sales price as provided by RJ Health of the vaccine product and reasonable costs of administration as required by § 7.14(b) if the vaccine is a Covered Service and not otherwise paid for by the Vaccines for Children Program under the Universal Childhood Vaccine Distribution Program.
- With respect to BCBSTN, the second sentence of § 7.14(b) of the Agreement is deleted (b) and is replaced by the following:

BCBSTN agrees to pay for the vaccines and injectibles themselves whether ordered by a Participating Physician in single or multiple doses.

H.8. § 7.16(a) Medical Necessity

With respect to BCBSNC:

As North Carolina state law requires BCBSNC to apply a statutory definition of "medical necessity" (or comparable term) for one or more of its products or Plans, BCBSNC may apply that statutory definition in lieu of the definition contained in § 7.16(a) to any other of its products or Plans.

H.9. § 7.16(c) Future Consideration of Administrative Exemption Program

BCBSTN shall not be subject to the provisions of § 7.16(c) of the Agreement and shall instead be subject to the following provision:

BCBSTN shall consider the feasibility and desirability of exempting certain Participating Physicians from certain administrative requirements based on criteria such as the Participating Physician's delivery of quality and cost effective medical care and accuracy and appropriateness of claims submissions. In connection with the above, BCBSTN intends to expand its current goldcard program. To the extent that the Tennessee Medical Association submits written recommendations to BCBSTN regarding issues related to any expansion of its goldcard program, BCBSTN will respond to such recommendations in writing within sixty (60) days of receipt. To the extent a recommendation proposed by the Tennessee Medical Association is rejected by BCBSTN, it will post on its Public Website Tennessee Medical Association's proposed recommendation and BCBSTN's response. BCBSTN shall not be obliged to implement any exemption process under this § 7.16(c) during the term hereof, and this § 7.16(c) is not intended and shall not be construed to limit BCBSTN's ability to implement any such program on a pilot or experimental basis, base exemptions on any BCBSTN determined basis, or otherwise to implement one or more programs in only some markets.

H.10. § 7.18 Timelines for Processing and Payment of Complete Claims

- (a) With respect to BCBSTN, § 7.18 of the Agreement is amended by adding new subsections (g) and (h) as follows:
 - BCBSTN agrees not to defer payment of claims that would otherwise be (g) Complete Claims submitted by any Participating Physician, Participating Physician Group or Participating Physician Organization currently treating or that has treated the Plan Member where that provider does not have the medical records necessary to determine if a Plan Member's condition is preexisting under the applicable Plan Document, provided however, that BCBSTN reserves the right to recoup any Overpayments made on claims related to pre-existing conditions, as otherwise appropriate under the terms of this Agreement. BCBSTN further agrees that if it is investigating a pre-existing condition issue that may affect payment of a claim, it will promptly notify the appropriate Participating Physician, Participating Physician Group or Participating Physician Organization whose payment is affected of that fact in writing.
 - (h) Notwithstanding the provisions of § 7.18(a) of the Agreement, BCBSTN will, with respect to the first claim for a Plan Member by a Participating Physician, Participating Physician Group or Participating Physician Organization, pay a claim that would otherwise be a Complete Claim submitted by any Participating Physician, Participating Physician Group or Participating Physician Organization within sixty (60) days following BCBSTN's receipt of such claim to obtain information regarding coordination of benefits for that Plan Member. BCBSTN further agrees that if it is investigating a coordination of benefits issue that may affect the timely payment of a claim, it will promptly notify the appropriate Participating Physician, Participating Physician Group or Participating Physician Organization of that fact in writing.
- CareFirst will not be required to follow § 7.18 for the Maryland state employee health (b) benefits program.
- (c) BCBSNC shall not be subject to the obligations in § 7.18 in connection with its administration of the North Carolina Teachers' and State Employees' Plan (the "Plan"); provided, however, that BCBSNC will comply with all applicable state laws, and will endeavor to comply with the obligations in § 7.18 regarding claims processing timelines unless the State of North Carolina and/or the Plan interferes with BCBSNC's efforts to comply.
- (d) Triple-S is not subject to the provisions of § 7.18.

H.11. §§ 7.8, 7.18, 7.19 & 7.20 Bundling and Claim Editing Systems (excluding Legacy **Exceptions**)

BCBSMI is migrating its local claims system to its NASCO based system. BCBSMI (a) will, upon becoming aware of a deficiency in its claim's system with respect to §§ 7.19 and 7.20 of the Agreement, cure the deficiency on a going forward basis so that the system complies with the § 7 standards.

H.12 §§ 7.8, 7.18, 7.19 & 7.20 Bundling and Claim Editing Systems (including Legacy Exceptions)

(a) The following Blue Plans will comply with the obligations under § 7 except with respect to the specified sub-sections for the identified claim systems:

Plan	System Name	Sections of the Agreement for which the Blue Plan has no obligation to comply	
Premera	(1) BlueChip	(1) §§ 7.8(d), 7.18, 7.19, 7.20	
	(2) GBAS	(2) §§ 7.8(d), 7.18, 7.19, 7.20	
BCBSNC	AMYSIS	§§ 7.8(b)(d), 7.18, 7.19, 7.20 (Moreover, BCBSNC agrees that, during the Effective Period, the number of covered lives for which claims are processed on Amysis will not exceed 350,000)	
Empire	NASCO	§§ 7.8, 7.10, 7.18, 7.19, 7.20, 7.21; provided, however, that these systems are intended to operate as legacy systems with no new group enrollment being administered with these systems, and the systems will not be used for purposes of processing claims for any groups not processed on such systems as of the Effective Date.	
Blue Cross and Blue Shield of Florida, Inc.	MHS System	§§ 7.8(d), 7.18, 7.19, and 7.20. The MHS System will only be used for the existing business being processed on the system.	

- (b) BCBS Ala shall not be required to comply with §§ 7.8, 7.19, and 7.20 in administering claims under the Medicare Advantage Program for so long as it uses the CCI in processing such Medicare Advantage claims and CCI is the sole system of Edits applied to such claims.
- (c) With respect to BCBS Ala, one of the significant benefits of this Agreement for Physicians providing services to Plan Members is that, in implementing the Agreement, BCBS Ala will change its policy from one of generally not recognizing modifiers 25 and 59 to a policy where it generally will recognize those modifiers when billed in accordance with AMA's CPT as provided in the Agreement, with limited exceptions, consistent with §§ 7.8 and 7.20. In the event BCBS Ala determines, after the Implementation Date applicable to §§ 7.8 and 7.20 and during the remaining term of this Agreement, that additional code combination exceptions to recognition of modifier 25 and/or 59 are necessary to be implemented, BCBS Ala will give advance notice via its Provider Website of the proposed additional exceptions, will seek advance comments thereon, and will meet or confer with its Preferred Medical Doctor Physicians'

Advisory Committee prior to implementing such additional exceptions, which implementation shall otherwise be consistent with §§ 7.8 and 7.20. In addition, upon request, BCBS Ala will meet or confer with elected officials and established committees of MASA as referenced in § 7.9(g)(2) of this Exhibit H above, and the specialty representative(s) and other representatives referred in § 7.9(g)(4) of this Exhibit H above whose specialties are predominantly affected by its proposed additional code combination exceptions, and discuss and obtain input thereon prior to implementing any such additional exceptions. The exceptions that BCBS Ala decides to implement after going through this process will be disclosed to Physicians consistent with § 7.8. Any Compliance Dispute with respect to additional exceptions to modifiers 25 and 59 will be limited to a determination that BCBS Ala complied with this process and, if it did not, what the appropriate remedy would be for BCBS Ala to fulfill its obligations under this paragraph.

- BCBSSC shall not be required to comply with §§ 7.8, 7.19, and 7.20 in administering claims under the Medicare Advantage Program to the extent it uses CCI for such claims and CCI differs from the rules in §§ 7.19 and 7.20.
- BCBSTN shall not be required to comply with §§ 7.8, 7.19, and 7.20 in administering (e) claims under the Medicare Advantage Program to the extent it uses CCI for such claims and CCI differs from the rules in §§ 7.19 and 7.20.

H.13. §§ 7.8, 7.14, 7.18, 7.20, 7.21 Bundling and Claim Editing Systems (Special Implementation)

For its LRSP system, BCBSNC will implement the obligations of §§ 7.8 (b), 7.8 (d), (d)(i), (d)(iii), 7.14(d), 7.18, 7.20 and 7.21 in accordance with Schedule 1 of Exhibit H.

H.14. § 7.21 EOB and Remittance Advice

(a) BCBSMI shall comply with the requirements of § 7.21(a) of the Agreement except that BCBSMI need not provide a separate identification number as provided in clause (i) of § 7.21(a).

H.15. § 7.22 Overpayment Recovery Procedures

- (a) BCBS Ala shall not be required to comply with the obligations under § 7.22 of the Agreement with respect to the time limit for Overpayment recovery efforts for claims that are paid under a Plan that are later determined to be the responsibility of a workers compensation carrier.
- CareFirst shall comply with the obligations under § 7.22 except with respect to the 18-(b) month time limit for Overpayment recovery, which shall not apply to the Maryland state employee health benefits program.
- HMSA shall comply with the obligations under § 7.22 except with respect to the time limit for Overpayment recovery efforts for claims that are paid under a Plan that are later determined to be the responsibility of a workers compensation carrier. For claims that are determined to be the responsibility of a workers compensation carrier, HMSA shall have four years after the payment was received by the Physician to recover from the Physician the

difference between what the Physician was paid by HMSA for the claim and the amount the claim is determined to be payable under workers compensation.

- HMSA will satisfy the notice requirements of § 7.22 for Medicare cross over claims and (d) Medigap claims by providing on the Report to Provider (i) the patient's name; (ii) claim identification number (iii) the service date; (iv) the payment amount received by Physician; and (v) a reasonably specific explanation of the change (including the procedure code, where appropriate).
- The following provision of § 7.22 applies to BCBSNC's Fully-Insured and Self-Insured (e) Plans: "If a Physician requests an appeal within thirty (30) days of receipt of a request for repayment of an Overpayment, no Blue Plan shall require such Physician to repay the alleged Overpayment before such appeal is concluded."
- With respect to the provision of the thirty (30) days written notice under § 7.22, the Implementation Date for BCNEPA shall be extended by an additional six (6) months.

H.16. § 7.25 Effect of Blue Plan Confirmation of Patient Procedure/Medical Necessity

With respect to BCBSNC, the first sentence of § 7.25 of the Agreement is deleted and is replaced by the following:

If BCBSNC has predetermined that a Covered Service is Medically Necessary, it will not revoke the Medical Necessity determination if (1) the Clinical Information recorded at the time of the service is consistent with the basis for the Medical Necessity predetermination, (2) the Plan Member is eligible for the Covered Service at the time the service is provided, and (3) the determination was not based on a material misrepresentation about the Plan Member's health condition that was knowingly made by the Plan Member or the Physician who provided the service.

H.17. § 7.29(c) Arbitration

- With respect to HCSC, notwithstanding the requirements of § 7.29(c)(ii)(F), HCSC may require arbitration to occur: (1) in the following cities, which are closest to the principal office of the Physicians, Physicians Groups, or Physicians Organizations located in Illinois: Aurora, Arlington Heights, Bloomington, Chicago, Carbondale, Elgin, Joliet, Springfield, Champaign-Urbana, Peoria, Rock Island, Rockford, and Quincy; (2) in the following cities, which are closest to the principal office of the Physicians, Physicians Groups, or Physicians Organizations located in Texas: Abilene, Amarillo, Austin, Corpus Christi, Dallas, El Paso, Houston, Lubbock, Lufkin, Midland-Odessa, San Angelo, San Antonio, Texarkana, Waco, Wichita Falls, and Brownsville; and (3) for Physicians, Physicians Groups, or Physicians Organizations located in Oklahoma, no more than 150 miles from the principal office of the Physician, Physicians Group, or Physicians Organization.
- With respect to BCBSMN, § 7.29(c)(ii)(F) of the Agreement is amended to read: "(F) (b) requiring that any arbitration proceeding occur more than 150 miles from the principal office of the Physician, Physician Group, or Physician Organization."

- (c) With respect to Regence, § 7.29 (c) is not applicable to Physicians practicing in Washington state with whom Regence has contracted because of the ruling of the Washington State Supreme Court voiding the arbitration provisions in the Physicians' contracts. If Regence utilizes arbitration provisions in its contracts with Physicians practicing in Washington state that are compliant with the ruling of the Washington State Supreme Court, then such a process will comply with § 7.29. With respect to all other Physicians contracting with Regence, § 7.29(c)(ii)(F) of the Agreement is amended to read: "(F) requiring that any arbitration proceeding occur more than 150 miles from the principal office of the Physician, Physician Group, or Physician Organization, except that for Physicians, Physicians Groups, or Physicians Organizations located in Idaho, arbitration may be required in Boise, Pocatello, or Lewiston, Idaho or in Spokane, Washington, which location is closest to the principal office of the Physicians, Physicians Groups, or Physicians Organizations."
- (d) With respect to BCBSTN, § 7.29(c)(i) of the Agreement shall be amended by adding the following:

"BCBSTN agrees that with respect to any arbitration proceeding between BCBSTN and a Participating Physician Group of six or more Physicians, the Participating Physician Group shall have the right to elect in advance of the proceeding that the prevailing party in the arbitration have any applicable filing fees and arbitrators' fees paid by the losing party. If the Participating Physician Group does not make such election, the parties shall bear their own filing fees and arbitrators' fees, as applicable."

- (e) HMSA may require arbitration proceedings to occur in Honolulu, Hawaii. Physicians who have an office on an island other than Oahu may, at the physician's election, participate in arbitrations by telephone or in person.
- (f) BCBSMT will not require any arbitration proceeding to occur more than 150 miles from the principal office of the Physician, Physician Group, or Physician Organization and may conduct arbitration proceedings telephonically in accordance with the Medicare rules.
- (g) Premera will not require any arbitration proceeding to occur more than 150 driving miles from the principal office of the Physician, Physician Group, or Physician Organization in the state of Alaska.

H.18. § 7.29(s) Meeting with Tennessee Medical Association

With respect to BCBSTN, § 7.29 of the Agreement is amended by adding a new subsection (s) as follows:

BCBSTN further agrees that Tennessee Medical Association will have access to all information disclosed on its Public Website which will include information on medical policies, claims edits, provider manuals, and appeals information related to §§ 7.10 and 7.11, as well as its general criteria and policies for goldcarding Participating Physicians.

Page 13 of 18

BCBSTN agrees to meet with representatives of Tennessee Medical Association as soon as is practicable to engage in good faith discussions to address issues concerning coordination of benefits, goldcarding and other matters of concern identified by the Tennessee Medical Association in advance of any such meeting.

BCBSTN agrees to continue with its current timeframe for credentialing physicians in its network through a process designed to enable it to make a decision regarding a physician's application within forty-five (45) days of receipt of all required information.

BCBSTN agrees to continue with its current policy of disclosing to Participating Physicians on its Public Website the identities of those entities which it provides access to its network of physicians and for which it does not adjudicate claims. The foregoing shall not apply to the BlueCard program or any similar national account delivery program governed by the Blue Cross and Blue Shield Association (hereinafter "BCBSA").

H.19. § 7.30 Compliance with State Law and Applicable Government Contracts

Each of the Blue Plans and entities listed below are participants in the Regional Joint (a) Venture for Medicare Advantage – Region 19, a contractual joint venture for the offering of a Medicare Advantage preferred provider organization ("PPO") program in the CMS geographic area designated as Region 19 (hereinafter the contractual joint venture participants are respectively referred to as the "Joint Venture Participants"). The terms of the Agreement shall only apply to each Joint Venture Participant who executed this Agreement with respect to its participation in the Joint Venture the later of (i) sixty (60) days following notice from Class Counsel that all of the Joint Venture Participants listed below have entered into a settlement agreement in the Action, or (ii) the applicable date(s) set forth in the Agreement with respect to a particular obligation of such Blue Plans relating to the Medicare Advantage PPO program, and which applies obligations equivalent to the obligations of this § 7.30(a) of this Exhibit H of this Agreement to all Joint Venture Participants. The obligations of a Blue Plan with respect to its participation in the joint venture shall be applicable only to those functions or activities performed by the Blue Plan or one or more of its Affiliates only so long as the Blue Plan remains a Joint Venture Participant. If any of the Blue Plans ceases to be a member of the joint venture, then the terms of the Agreement shall apply to any Medicare Advantage PPO program that the Blue Plan operates or administers. Where any of or part of any of the functions or activities of the joint venture are performed pursuant to any contract with any third party that is not an Affiliate of a Joint Venture Participant, the Joint Venture Participant shall make a good faith effort to include in contracts entered into with such third party terms that are substantially equivalent to the terms of this Agreement, provided that a Joint Venture Participant shall not be liable under this Agreement in the event that any such third party acts in a manner inconsistent with the terms of this Agreement. Nothing in this Agreement shall require a Joint Venture Participant to take any action, with regard to an agreement with a third party that is not an Affiliate and which agreement is in existence as of the Preliminary Approval Date, to amend or terminate any such agreement unless such action may be taken without any liability or additional costs to the Joint Venture Participant. In addition, the provisions of § 7.8 shall not apply to the Joint Venture Participants with respect to claims processing for the PPO so long as claims administration has been delegated to a third party vendor which is not an Affiliate of a Joint

Page 14 of 18

Venture Participant and provided that the vendor utilizes CCI for claim processing for the Medicare Advantage PPO program.

Joint Venture Participants:

Wellmark, Inc. d/b/a Wellmark Blue Cross and Blue Shield of Iowa

Wellmark of South Dakota, Inc. d/b/a Wellmark Blue Cross and Blue Shield of South Dakota

BCBSM, Inc. dba BlueCross BlueShield of Minnesota

Blue Cross and Blue Shield of Montana, Inc.

Blue Cross and Blue Shield of Nebraska

Noridian Mutual Insurance Company d/b/a Blue Cross Blue Shield of North Dakota, a

North Dakota Mutual Insurance Company

Blue Cross Blue Shield of Wyoming

- (b) BCBSTN is not required to comply with the obligations of § 7 of the Agreement with respect to the TennCare program.
- (c) HCSC is not required to comply with the obligations of § 7 of the Agreement with respect to the Indian Health Services program.
- (d) BCBSMN is not required to comply with the obligations of § 7 of the Agreement with respect to the Indian Health Services program.
- (e) As long as BCBSMI's Medicare Advantage programs apply the federal rules for "deemed contract providers" in Michigan (which rules, in summary, permit all willing providers meeting CMS standards to receive direct payment for services to Medicare Advantage enrollees), BCBSM's Medicare Advantage programs are excluded from the application of this Agreement.
- (f) HMSA shall not be required to comply with the obligations under § 7 with respect to the QUEST program. HMSA agrees that for the QUEST program:
 - 1. HMSA shall comply with § 7.13(b) and will not require a Participating Physician to participate in the QUEST program in order to participate in Product Networks in which such Participating Physician is compensated on a fee for service basis; and
 - 2. HMSA shall comply with § 7.20 solely by utilizing Medicare CCI Edits in processing QUEST claims and the CCI Edits are the sole system of Edits applied to such claims.
- (g) Products providing Health Care Reform Program, Reforma de la Salud, 24 LPRA sec. 7001, et seq. ("Reforma") coverage shall not be subject to the provisions of § 7, but notwithstanding this exclusion, the provisions of § 7, as herein agreed for Triple-S, Inc., and its affiliated companies ("Triple-S") shall apply to those Reforma claims paid by Triple-S under a fee for service schedule.

H.20. § 7.33 Managed Care Issues Relating to Mental Health and Substance Abuse

BCBSNC shall not be subject to the obligations in § 7.33 in connection with its administration of the North Carolina Teachers' and State Employees' Plan.

H.21. § 7.36 Competitive Flexibility – Modification for Specific Circumstances

(a) Regence is subject to the following addendum in § 7.36(c):

> In the event of any settlement between the Class and Blue Cross of Idaho that contains any business practice initiative that differs in any respect from the provision(s) in § 7 of this Agreement dealing with the same subject matter (a "Different Business Practice"), Regence may modify its undertakings in the corresponding portion of its service area pursuant to § 7.36(c) such that its obligations in that territory will be identical (except as to duration) to the Different Business Practice agreed to by Blue Cross of Idaho. It shall be conclusively presumed for purposes of § 7.36(c) that any such modification is "reasonably necessary" for Regence to compete in the affected part of Idaho. If Regence invokes this provision, it shall give notice as provided in § 7.36(c). Within ten (10) days of receipt of such notice, Class Counsel shall inform Regence whether they consider the Different Business Practice to be more beneficial to Physicians. If Class Counsel conclude that the Different Business Practice provides greater benefits to Physicians, then Regence may implement that Different Business Practice without any changes. If Class Counsel determines that the Different Business Practice provides lesser benefits to Physicians, then Class Counsel will identify to Regence any additional business practice obligation(s) (an "Additional Practice") that (a) the other Idaho plan agreed to adopt in exchange for the Class accepting the lesser business practice, and (b) is reasonably proportional to the lesser business practice. After receiving that information, if Regence wishes to employ any Different Business Practice(s) that provides lesser benefits to Physicians, it may do so only if it also adopts the corresponding Additional Business Practice(s). Any dispute as to whether an Additional Business Practice is reasonably proportional to a lesser business practice shall be resolved pursuant to § 12 of this Agreement.

(b) Triple-S is not subject to the provisions of § 7.36.

H.22. § 13.1 Discharge of Released Claims

The claims alleged in the action entitled Washington State Medical Ass'n, et al. v. Regence Blue Shield, pending in the Superior Court of the State of Washington for King County, are not Released Claims under this Agreement.

H.23. § 14.2 Right to Terminate this Agreement

With respect to BCBSMN, § 14.2 of the Agreement is amended by deleting § 14.2(d) and substituting the new subsection (d) as follows:

(d) If the number of putative Class Members within the State of Minnesota submitting Opt-Out requests exceeds five percent (5%) of the total number of putative Class Members in Minnesota, BCBSMN may reduce the amount of its Settlement Fund Payment under § 8.2 by a percentage equal to the percentage of Opt-Out Requests from putative Class Members within the State of Minnesota in excess of five percent (5%) (the "Settlement Reduction Amount") and to reduce the amount of Attorneys' Fees payable by it under § 9.1 by a percentage equal to the percentage of Opt-Out Requests from putative Class Members within the State of Minnesota in excess of five percent (5%) (the "Attorneys Fee Reduction Amount"). Such election shall be made by delivering a notice of such election to the Class Counsel within thirty (30) days after receipt of the complete list of Opt-Out requests from the Notice Administrator referred to in § 6. In the event BCBSMN makes such election, the Settlement Fund Amount shall be reduced by the Settlement Reduction Amount, and the Unopposed Amount shall be reduced by the Attorneys Fee Reduction Amount.

H.24. § 28 Effect of Other Settlements

Blue Party Blue Cross and Blue Shield Association is not a party to, and shall have no rights under § 28 of the Agreement.

Additional Non-Section Specific Exceptions

H.25. Empire

With regard to selection of the Billing Dispute External Reviewer, Medical Necessity Independent Review Organization, the Compliance Dispute Officer and Compliance Dispute Facilitator, Empire may use the same individuals and/or entities as were selected pursuant to the Wellpoint Settlement Agreement.

H.26. BCBSNC/NCMS

BCBSNC and the NCMS have agreed to engage in a high-level meeting in the near future in order to discuss issues of mutual interest. This meeting will include, but not be limited to, good faith discussions on the following items: coordination of benefits; any outstanding issues regarding immunizations and administration; physician contracts; non-participating physician issues; and medical necessity appeals. At the time the Parties file their Motion for Preliminary Approval, BCBSNC and the NCMS will issue a joint press release regarding their joint efforts and commitments, including through this Agreement and the upcoming meeting described herein, to promote better health and affordable quality health care.

H.27. BCBSMI/Michigan Health Insurance Company

Michigan Health Insurance Company's membership is being folded into BCBSMI over the next months, and when that membership becomes part of BCBSMI, Covered Services provided by Physicians to this membership will then be subject to the Agreement. In addition, the parties agree that Class Members who had provider agreements with Michigan Health Insurance

Company and BCBSMI may, for the purposes of § 8.3(c), aggregate gross receipts received from Michigan Health Insurance Company for Covered Services with the gross receipts received from BCBSMI for Covered Services for each of the years 2004, 2005, and 2006.

H.28. HMSA - 65C Plus Program

This provision clarifies that HMSA's provision of Medicare part B services under its 65C Plus program is included within the exemption of 7.30(a) for a Blue Party's role as a carrier providing administrative services for Medicare Part B business.

Page 18 of 18

Schedule 1 to Exhibit H

BCBSNC's Implementation Dates for LRSP for Select § 7 Provisions

BCBSNC's obligations under the following provisions of § 7, with respect to LRSP, will commence on the later of the Start Date in Exhibit I of the Agreement or the LRSP Start Date identified herein.

Section	Obligation	Exhibit I Start Date	BCBSNC's LRSP Start Date
§ 7.8	Disclosure of and Commitments Concerning Claims Payment Practices		
§ 7.8(b)	Consistency of Claim Payment Practices	Final Order Date + 12 months	1/01/2009
§ 7.8(d)	Disclosure of Significant Edits on Provider Website	Final Order Date + 6 months (or as soon thereafter as practicable)	1/01/2009
§ 7.8(d)(i)	Publish Customized Edits on Provider Website	Final Order Date + 6 months (or as soon thereafter as practicable)	1/01/2009
§ 7.8(d)(iii)	Disclosures regarding Modifiers 25 and 59	Final Order Date + 6 months	1/01/2009
§ 7.14	Fee Schedule Changes	-	-
§ 7.14(d)	Usual, Reasonable, and Customary Determinations	Final Order Date + 3 months	1/01/2009
§ 7.18	Timeliness for Processing and Payment of Complete Claims	_	-
§ 7.18(a)	15 Day Period for Processing and Payment of Electronically Submitted Complete Claims	Effective Date + 1 year	1/01/2009
§ 7.18(b)	Timing of Payment for Complete Claims and Payment of Interest (15 day Period)	Effective Date + 1 year	1/01/2009
§ 7.20	Bundling and Other Computerized Claim Editing	_	_
§ 7.20(a)-(h)	Modifications to Payment Policies	Final Order Date + 9 months	1/01/2009
§ 7.20(i)	Updating Claims Editing Software	Final Order Date + 6 months (or as soon thereafter as practicable)	1/01/2009
§ 7.21	EOB and Remittance Advice Content		-
§ 7.21(a)	Content of EOB Forms	Final Order Date + 6 months (or as soon thereafter as practicable)	1/01/2009
§ 7.21(a)	Content of Remittance or Similar Forms	Final Order Date + 6 months (or as soon thereafter as practicable)	1/01/2009

$Exhibit\ I-Implementation\ Dates\ of\ Blue\ Plan's\ Obligations\ Under\ \S 7\ of\ the\ Agreement$

Pursuant to §7, the implementation dates of the Blue Plans' obligations under §7 of the Agreement are as follows, except as otherwise provided on Exhibit H.

Section	Obligation	Start Date	
§ 7.3	Availability of Fee Schedules and Scheduled Payment Dates – Complete Fee Information	Final Order Date + 12 months	
§ 7.3	Availability of Fee Schedules and Scheduled Payment Dates – Up to 100 CPT Codes	Final Order Date	
§ 7.5	Reduced Precertification Requirements – Posting on Provider Website	Final Order Date + 3 Months	
§ 7.5	Reduced Precertification Requirements – Proposals to Self-Insured Plans	Contracts Issued or Renewed After Final Order Date	
§ 7.6	Greater Notice of Policy and Procedure Changes	Final Order Date	
§ 7.8	Disclosure of and Commitments Concerning Claims Payment Practices	_	
§ 7.8(b)	Consistency of Claim Payment Practices	Final Order Date + 12 months	
§ 7.8(d)	Disclosure of Significant Edits on Provider Website	Final Order Date + 6 months (or as soon thereafter as practicable)	
§ 7.8(d)(i)	Publish Customized Edits on Provider Website	Final Order Date + 6 months (or as soon thereafter as practicable)	
§ 7.8(d)(ii)	Commitments regarding Submission of Clinical Information	Final Order Date	
§ 7.8(d)(iii)	Disclosures regarding Modifiers 25 and 59	Final Order Date + 6 months	
§ 7.9	Physician Advisory Committee	_	
§ 7.9(a)	Establishment of Physician Advisory Committee	The later of the Final Order Date + 3 months or the selection of the members of the Physician Advisory Committee in accordance with § 7.9(b)	
§ 7.9(b)	Selection of Members by Blue Plan	Preliminary Approval Date + 60 days	
§ 7.9(b)	Selection of Members by Physicians or State Medical Societies	Preliminary Approval Date + 60 days	
§ 7.9(b)	Selection of Remaining Members	Preliminary Approval Date + 120 days	
§ 7.10	New Dispute Resolution Process for Physician Billing Disputes	_	
§ 7.10(a)	Establishment of Billing Dispute Reviewer ("BDR")	Final Order Date + 4 months	
§ 7.10(d)	Selection of BDR	Preliminary Approval Date + 120 days	
§ 7.11	Determinations Related to Medical Necessity or the Experimental or Investigational Nature of Any Proposed Health Care Service or Supply	_	
§ 7.11(a)	Procedures regarding Initial Determinations	Final Order Date	
§ 7.11(b)(ii)	Establishment of Internal and External Review Process	Final Order Date + 12 months	
§ 7.13	Participating in Blue Plans' Network	_	
§ 7.13(a)	Credentialing of Physicians	Final Order Date + 6 months	
§ 7.13(b)	General Commitments regarding All Products Clauses	Final Order Date	
§ 7.13(c)	Termination Without Cause	Final Order Date	
§ 7.14	Fee Schedule Changes	_	
§ 7.14(a)	Establishing Standard Fee Schedules	Final Order Date	
§ 7.14(a)	Limitations regarding Reduction of Fee Schedules	January 1 of the Year following the Effective Date	
§ 7.14(b)	Payment Rules for Injectibles, DME, Administration of Vaccines, and Review of New Technologies	Final Order Date	
§ 7.14(c)	Usual, Reasonable, and Customary Appeals	Final Order Date + 3 months	
§ 7.14(d)	Usual, Reasonable, and Customary Determinations	Final Order Date + 3 months	
§ 7.16	Application of Clinical Judgment	-	
§ 7.16(a)	Medical Necessity Definition – Current Agreements	Final Order Date	
§ 7.16(a)	Medical Necessity Definition – Future Agreements	Agreements Issued After Final Order Date	
§ 7.16(a)	Adverse Determination Denial Rate	For the calendar year beginning after the Final Order Date	
§ 7.16(b)	Policy Issues Involving Clinical Judgment	Final Order Date	

Exhibit I – Start Dates of Defendants' Obligations Under Section 7 of the Agreement

Section	Obligation	Start Date	
§ 7.16(c)	Consideration of an Administrative Exemption	Final Order Date + 6 months	
	Program		
§ 7.17	Billing and Payment	-	
§ 7.17(a)	Time Period for Submission of Claims	Final Order Date	
§ 7.17(b)	Claims Submission (re: acceptance of certain forms	Final Order Date	
	and submissions of Clinical Information)		
§ 7.18	Timeliness for Processing and Payment of Complete	_	
	Claims		
§ 7.18(a)	30 Day Period for Processing and Payment of	Final Order Date + 9 months	
	Complete Claims	7700 1 79 1	
§ 7.18(a)	15 Business Day Period for Processing and Payment of Electronically Submitted Complete Claims	Effective Date + 1 year	
§ 7.18(b)-(c)	Timing of Payment for Complete Claims and Payment	Final Order Date + 9 months	
	of Interest (30 Day Period for Processing and Payment		
	of Completed Claims)		
§ 7.18(b)-(c)	Timing of Payment for Complete Claims and Payment	Effective Date + 1 year	
	of Interest (15 Business Day Period for Processing and		
0.7.10(1)	Payment of Electronically Submitted Claims)	Einel Onder Detect Oursette	
§ 7.18(d)	Commitments regarding Claim Handling	Final Order Date + 9 months	
§ 7.18(e)	Exceptions	Final Order Date	
§ 7.19	No Automatic Downcoding of Evaluation and	Final Order Date	
\$ 7.00	Management Claims Bundling and Other Computerized Claim Editing		
§ 7.20 § 7.20(a)	Modifier 51-exempt codes	Final Order Date + 9 months	
§ 7.20(a) § 7.20(b)	Add-on codes	Final Order Date + 9 months Final Order Date + 9 months	
§ 7.20(b)	Modifiers 25 and 57	Final Order Date + 9 months Final Order Date + 9 months	
§ 7.20(d)	Supervision or interpretation or radiologic guidance	Final Order Date + 9 months Final Order Date + 9 months	
§ 7.20(u)	codes	Timal Order Date + 9 months	
§ 7.20(e)	Indented codes	Final Order Date + 9 months	
§ 7.20(f)	Modifier 59	Final Order Date + 9 months	
§ 7.20(g)	Global periods	Final Order Date + 9 months	
§ 7.20(h)	Intensity codes	Final Order Date + 9 months	
§ 7.20(i)	Updating Claims Editing Software	Final Order Date + 6 months (or as soon thereafter as	
0 ()		practicable)	
§ 7.21	EOB and Remittance Advice Content	_	
§ 7.21(a)	Content of EOB Forms	Final Order Date + 6 months (or as soon thereafter as	
		practicable)	
§ 7.21(a)	Content of Remittance or Similar Forms	Final Order Date + 6 months (or as soon thereafter as	
		practicable)	
§ 7.22	Overpayment Recovery Procedures	Final Order Date	
§ 7.23	Improve Accuracy of Eligibility Information	Final Order Date	
§ 7.25	Confirmation of Medical Necessity	Final Order Date	
§ 7.27	Information about Physicians Provided by Blue Plan	Final Order Date	
§ 7.29	Miscellaneous		
§ 7.29(a)	No "Gag Clauses"	Final Order Date	
§ 7.29(c)	Arbitration	Final Order Date	
§ 7.29(d)	Impact on Standard Form Agreements and Individually	_	
	Negotiated Contracts		
§ 7.29(d)(i)	Standard Form Agreements	Final Order Date	
§ 7.29(d)(ii)	Current Individually Negotiated Contracts	Preliminary Approval Date	
§ 7.29(d)(iii)	Future Individually Negotiated Contracts	Preliminary Approval Date	
§ 7.29(e)	Impact on Covered Services	Final Order Date	
§ 7.29(e)	Recommendation to Self-Insured Plans	Contracts Issued or Renewed After Final Order Date	

Exhibit I - Start Dates of Defendants' Obligations Under Section 7 of the Agreement

Section	Obligation	Start Date
§ 7.29(i)	Pharmacy Provisions	Final Order Date
§ 7.29(j)	Restrictive Endorsements	Preliminary Approval Date
§ 7.29(k)	Scope of Blue Parties' Responsibilities	_
§ 7.29(k)(i)	Future Contracts with Delegated Entities	Contracts Entered Into After Final Order Date
§ 7.29(k)(ii)	Application to BlueCard Program or Similar Programs	Final Order Date
§ 7.29(k)(iii)	BlueCard Program Initiatives and Investments	Final Order Date
§ 7.29(k)(iv)	BCBSA Rules and Regulations	Preliminary Approval Date
§ 7.29(1)	Copies of Contracts	Final Order Date
§ 7.29(m)	State and Federal Laws and Regulations	Preliminary Approval Date
§ 7.29(o)	Limitations on Obligations of Non-Participating	Preliminary Approval Date
	Physicians	
§ 7.29(q)	Effect of Assignment of Benefits	Preliminary Approval Date
§ 7.29(r)	"Most Favored Nations" Clauses	Contracts Issued or Renewed After Final Order Date
§ 7.30	Compliance with Applicable Law and Requirements of	Preliminary Approval Date
	Government Contracts	
§ 7.31	Estimated Value of Section 7 Initiatives	-
§ 7.32	Force Majeure	Preliminary Approval Date
§ 7.33	Mental Health and Substance Abuse Provisions	
§ 7.33(a)	Medical Necessity Definitions – Current Agreements	Final Order Date
§ 7.33(a)	Medical Necessity Definitions – Future Agreements	Agreements Issued After Final Order Date
§ 7.33(b)	Listing of Participating Psychiatrists; Referrals	Final Order Date
§ 7.33(c)	Payment for Medically Necessary Covered Services	Final Order Date
§ 7.33(d)	"Prudent Lay Person" Standard	Final Order Date
§ 7.33(e)	Posting of Authorization Form	Final Order Date
§ 7.35	Use of Third Party Intellectual Property	Preliminary Approval Date
§ 7.36	Competitive Flexibility	
§ 7.36(a)	Negotiated Purchaser	Preliminary Approval Date
§ 7.36(b)	Ability of Blue Parties to Substitute Business Practices	Preliminary Approval Date
§ 7.36(c)	Modification for Specific Circumstances	Preliminary Approval Date

EXHIBIT J

Page 2 of 2

INTENTIONALLY LEFT BLANK

EXHIBIT K

PLAN	%	Settlement Fund Amount
Blue Cross and Blue Shield of Alabama	6.158536525	\$7,902,382
Blue Cross and Blue Shield of Florida, Inc.	5.877914264	\$7,542,299
Hawaii Medical Service Association	1.591806540	\$2,042,541
Wellmark, Inc.	3.088739419	\$3,963,344
Health Care Service Corporation	17.198225470	\$22,068,059
Louisiana Health Services and Indemnity Company	1.623208727	\$2,082,835
CareFirst, Inc.	5.124468317	\$6,575,508
Blue Cross & Blue Shield of Massachusetts, Inc.	6.031419783	\$7,739,271
Blue Cross and Blue Shield of Michigan	8.954196926	\$11,489,659
BCBSM, Inc.	4.735037255	\$6,075,806
Blue Cross & Blue Shield of Mississippi and HMO of Mississippi, Inc.	1.510511086	\$1,938,226
Blue Cross and Blue Shield of Montana, Inc.	0.531341761	\$681,796
Horizon Blue Cross and Blue Shield of New Jersey	5.789881530	\$7,429,339
Empire HealthChoice HMO, Inc.	7.865204896	\$10,092,309
Blue Cross and Blue Shield of North Carolina	4.856747028	\$6,231,979
Hospital Service Association of Northeastern PA	0.808780473	\$1,037,794

PLAN	<u>%</u>	Settlement Fund Amount
Triple-S, Inc., of Puerto Rico	2.038417575	\$2,615,614
Blue Cross & Blue Shield of Rhode Island	1.491935057	\$1,914,390
Blue Cross and Blue Shield of South Carolina	2.642897579	\$3,391,258
Blue Cross Blue Shield of Tennessee, Inc.	4.543583205	\$5,830,140
Premera	2.392930909	\$3,070,511
Regence Blue Cross Blue Shield of Oregon	1.879027360	\$2,411,091
Regence Blue Shield (Washington)	2.064281866	\$2,648,802
Regence Blue Shield of Idaho	0.458657865	\$588,531
Regence Blue Cross Blue Shield of Utah	0.742248582	\$952,423
TOTAL	100%	\$128,315,907

Page 1 of 5

EXHIBIT L

List of Charitable Organizations

Pursuant to §§ 8.3 (g), 8.3 (h) and 8.4, the list of charitable foundations is as follow:

National Charitable Organization

1. Physicians' Foundation for Health Systems Innovations

Document 928-19

State Charitable Organizations

Alabama

2. Alabama Department of Public Health

California

3. California Medical Association Foundation

Colorado

4. El Paso County Medical Society Foundation

Connecticut

5. CSMS Physicians' Health and Education Fund (Connecticut State Medical Society)

Florida

- 6. Florida Medical Foundation (Florida Medical Association)
- 7. Florida Academy of Family Physicians Foundation

Hawaii

8. Hawaii Medical Foundation

Illinois

9. Illinois State Medical Society Educational and Scientific Foundation

Iowa

10. The University of Iowa Foundation (Medical Center for Development)

Georgia

11. Medical Association of Georgia Institute for Excellence in Medicine, Inc.

Page 2 of 5

Louisiana

12. Louisiana State Medical Society Educational and Research Foundation

Massachusetts

- 13. BCBSMA Foundation
- 14. MMS Alliance Charitable Foundation
- 15. Community Health Center Capital Fund

Michigan

16. The Michigan Health & Safety Coalition

Minnesota

17. Minnesota Physicians Foundation (Minnesota Medical Association)

Nebraska

18. Nebraska Medical Foundation (Nebraska Medical Association)

New Hampshire

19. John P. Bowler, M.D., Memorial Library (New Hampshire Medical Society)

New Jersey

20. Institute of Medicine and Public Health of New Jersey, Inc. (Medical Society of New Jersey)

New Mexico

- 21. Physician Outreach Program (University of New Mexico Health Sciences Center)
- 22. Carrie Tingley Foundation (Carrie Tingley Hospital)

New York

23. Medical, Educational and Scientific Foundation of New York, Inc. (Medical Society of the State of New York)

North Carolina

24. North Carolina Medical Society Foundation, Community Practitioners Program

Northern Virginia

25. Medical Society of Northern Virginia Foundation

Page 3 of 5

Oklahoma

26. Oklahoma Medical Research Foundation

Oregon

27. Oregon Medical Education Foundation

Pennsylvania

- 28. The Foundation of the Pennsylvania Medical Society
- 29. Commission on Economic Opportunity (Free Health Clinics)
- 30. Moses Taylor Hospital (Patient Care for Underinsured/Uninsured)
- 31. Susquehanna Health System (Community Based Health Clinics for Underinsured/Uninsured)
- 32. Pocono Health System (Community Based Health Clinics for Underinsured/Uninsured)

Puerto Rico

33. Fundación del Colegio de Médicos y Cirujans de Puerto Rico

Rhode Island

- 34. Rhode Island Community Health Center Association
- 35. Rhode Island Free Clinic
- 36. Rhode Island Medical Society Foundation

South Carolina

37. South Carolina Medical Association Foundation

South Dakota

- 38. The University of South Dakota Foundation (Medical School Program)
- 39. The University of South Dakota Foundation (Lee Medical Building)

Tennessee

- 40. Tennessee Medical Foundation (Tennessee Medical Association)
- 41. Tennessee Foundation for Quality Patient Healthcare

Texas

42. Texas Medical Association Foundation

Page 4 of 5

Utah

- 43. Utah Medical Association Foundation
- 44. The Fourth Street Clinic

Vermont

45. Vermont Medical Society Education and Research Foundation, Inc.

Virginia

46. Arlington County Medical Society Foundation

Washington

47. Washington State Medical Education and Research Foundation

Exhibit M

List of Parties Making Payments to Representative Plaintiffs in Other Actions

Pursuant to §9.4, the Blue Parties listed below shall pay such fees to the Representative Plaintiffs in Other Actions as designated, not later than ten (10) days after the Effective Date, in the amount awarded by the Court up to but not exceeding the unopposed amount set forth in the Agreement.

Blue Party Responsible for Payment	Representative Plaintiffs in Other Action
Hawaii Medical Service Association	Maxwell Cooper, M.D.
Hawaii Medical Service Association	Michon Morita, M.D.
Empire Blue Cross Blue Shield	Andrew Cheng, M.D.
Blue Cross Blue Shield South Carolina	Joseph K. Newsom, M.D.
Blue Cross Blue Shield South Carolina	Joseph K. Newsom, Jr.
Blue Cross Blue Shield South Carolina	Robert Waters, M.D.
Blue Cross Blue Shield South Carolina	John R. Nobles, M.D.
Blue Cross Blue Shield Tennessee	Zachary Rosenberg, M.D.
Blue Cross Blue Shield Tennessee	Dewayne P. Darby, M.D.
Blue Cross Blue Shield of North	Donald S. Horner, M.D.
Carolina	
Blue Cross Blue Shield of North	William Haggerson, M.D.
Carolina	
Louisiana Health Service & Indemnity	Lakeland Anesthesia, Inc.
Company	

Exhibit N

Tag-Along Actions

List of pending Tag-Along Actions (§ 15.1 of the Agreement)

No.	Case Caption		
1.	Decatur Healthcare, LLC, et al. v. Blue Cross and Blue Shield Association, together		
	with its subsidiaries, licensees, and affiliates, and Health Care Service Corporation,		
	D/B/A Blue Cross Blue Shield of Illinois		
2.	Donald S. Horner, M.D., et al. v. Blue Cross Blue Shield of North Carolina		
3.	Paul S. Howard, M.D. v. Blue Cross Blue Shield of Alabama		
4.	North Carolina Medical Society v. Blue Cross Blue Shield of North Carolina		
5.	Charles Stein, M.D. v. Blue Cross Blue Shield of Florida, at al.		
6.	Michael A. Genord, M.D., John R. Sanborn, M.D., Paula A. Fishbaugh,		
	M.D., Andrea L. Schiller, M.D., Mark D. Dykowski, M.D., John E. Uckele,		
	M.D., and Betty S. Chu, M.D., on behalf of themselves and all others		
	similarly situated v. Blue Cross and Blue Shield of Michigan		

Page 1 of 7

UNITED STATES DISTRICT COURT FOR THE SOUTHERN DISTRICT OF FLORIDA MIAMI DIVISION

CASE NO. 03-21296-CIV-MORENO-KLEIN

RICK LOVE, M.D., et al.,

Plaintiffs,

v.

THE BLUE CROSS AND BLUE SHIELD ASSOCIATION, et al.,

Defendants.

EXHIBIT O (BCBSA)

AGREED TERMS OF POTENTIAL DISCOVERY FROM BCBSA

Notwithstanding the provisions of section 14.1 of the Settlement Agreement, the plaintiffs shall be permitted to take discovery of BCBSA as provided in this Exhibit O and nothing in this Exhibit shall be construed to prevent the plaintiffs from obtaining discovery from any defendant(s) that have not settled.

I. TERMS

- 1. Method of obtaining discovery from BCBSA. To take any permitted discovery (including documents, data, and deposition), the plaintiffs shall serve a subpoena on BCBSA in a manner consistent with the Federal Rules of Civil Procedure. The plaintiffs may seek discovery solely to the extent provided in this Exhibit O and any discovery requests shall be in substantially the same form as the corresponding provisions of this Exhibit O. BCBSA shall not seek to quash the subpoena, on the condition that the subpoena is properly served and the subpoena comports with and does not exceed the terms of the parties' agreement embodied in this Exhibit O.
- 2. Relevant period. Unless otherwise indicated, the period for discovery from BCBSA is May 22, 1999, to the end date for the class period certified by the Court.
- 3. Cost and expense. BCBSA shall bear its own costs and expenses, including attorneys' fees, associated with providing permitted discovery under this Agreement, provided that the plaintiffs shall reimburse BCBSA for costs and expenses excluding attorneys' fees exceeding \$175,000 invoiced by third party vendors for: (i) the search

for, review of, and production of electronically-stored mail or data; and (ii) copying, scanning, bates-numbering, electronically formatting, redacting, labeling, and related tasks associated with the search for, review of, and production of non-electronically stored documents (i.e., hard copies).

4. Scope of and timing of production from BCBSA.

- a. Subject matters. The subject matters on which BCBSA shall provide discovery from any source described below in this Exhibit O are limited to claims processing and provider contracting, including concerted activity among defendants in the Action that relates to claims processing and provider contracting.
- b. Nature of documents required to be produced. With respect to non-data discovery, BCBSA shall search for and produce responsive documents from the relevant period that are contained in the readily accessible central files and readily accessible paper files of the core employees most likely to have responsive documents in their files. In addition:
 - (1) With respect to documents relating to BCBSA committees or workgroups under paragraph III.A.1 below, BCBSA's obligation to search and produce documents shall be limited to searching and producing readily accessible paper documents from the individual(s) at BCBSA principally responsible for staffing or interacting with the particular committee or workgroup, such as the individual (if any) who maintains the central file for a particular BCBSA committee or workgroup.
 - (2) With respect to documents related to communications with certain third parties under paragraph III.A.4 below, BCBSA's obligation to search and produce documents shall be limited to searching and producing readily accessible paper documents from the individual(s) at BCBSA principally responsible for interfacing with the third parties identified below or, instead, most likely to possess materials regarding the interface between these third parties and the defendants.
- c. Additional documents from core employees. Upon a showing of special need or hardship and satisfying the standard set forth in subparagraph I.4.g below, the plaintiffs may request that BCBSA conduct a specific search for and produce the paper files from the offices of up to a maximum of 3 additional current officers or employees of BCBSA, in the aggregate, on subjects responsive to paragraphs III(A)(1), III(A)(2), or III(A)(4) below.
- d. Obligations regarding email. Except as provided in this subparagraph and section III.B below with respect to data discovery, BCBSA shall have no obligation to search for or produce claims data, any other forms of data, any electronic documents, or any electronically-stored emails. Upon a showing of special need

or hardship and satisfying the standard set forth in subparagraph I.4.g below, the plaintiffs may request that BCBSA conduct a specific search for and produce electronically-stored email of a limited number of current or former officers or employees of BCBSA, on subjects responsive to paragraphs III(A)(1), III(A)(2), or III(A)(4) below. If the plaintiffs seek email, BCBSA and the plaintiffs shall meet and confer regarding the scope of any search and production of emails, including for the purpose of negotiating the selection and definition of any search terms and the number and identity of any custodians, and the number and scope of search terms and custodians shall be narrowly tailored and shall not subject BCBSA to undue burden or hardship. BCBSA's obligations to search for and produce emails shall be limited to live emails on the exchange server(s) of BCBSA at the time of the search and the relevant period for any emails that BCBSA is obligated to search for and produce is May 22, 1999 to the Preliminary Approval Date. Email that relates solely to matters between BCBSA and a defendant that has settled need not be searched for or produced.

- e. BCBSA files only. BCBSA shall be obligated to search for and produce only discovery from BCBSA files on its premises. BCBSA shall have no obligation to search for or produce discovery in the possession, custody, or control of any of its licensees or their Affiliates.
- f. Privilege and similar protections. BCBSA shall not be obligated to provide information subject to the attorney-client privilege, work product doctrine, the joint defense or common interest privilege, or any other applicable privileges or doctrines that may shield information from discovery. If a log of documents withheld under this paragraph is required, it will be provided according to the terms and conditions for a privilege log as set forth in the June 14, 2004 Joint Case Management Order Setting Discovery Plan in this Action.
- g. Timing of data and document discovery from BCBSA. The plaintiffs shall not be permitted to seek any discovery unless and until: (i) merits discovery is proceeding in the Action, as stated by explicit order of the Court; and (ii) the plaintiffs have attempted to obtain from defendants that have not settled the discovery they seek from BCBSA, the plaintiffs have been unable to obtain such discovery from those defendants, and the plaintiffs have reasonably concluded that they cannot obtain the discovery they seek from defendants that have not settled. BCBSA may require that the plaintiffs meet and confer with BCBSA on any discovery requests the plaintiffs issue under this Exhibit O, to narrow such requests as BCBSA believes may be appropriate for the Action as it stands at the time of the requests and to verify that the plaintiffs have complied with the terms set forth in this Exhibit O.
- h. Timing of deposition of BCBSA. The plaintiffs shall not be permitted to call BCBSA for deposition unless and until: (i) merits discovery is proceeding in the Action, as stated by explicit order of the Court; and (ii) the plaintiffs have attempted to seek the information from defendants that have not settled or their

witnesses that they would seek to obtain by deposition from BCBSA, the plaintiffs have been unable to obtain such information from those defendants, and the plaintiffs have reasonably concluded that they cannot obtain the information they seek from defendants that have not settled. BCBSA may require that the plaintiffs meet and confer with BCBSA on any request by the plaintiffs to take BCBSA's deposition, to narrow the scope of the deposition as BCBSA believes may be appropriate for the Action as it stands at the time of the request and to verify that the plaintiffs have complied with the terms set forth in this Exhibit O.

- BCBSA shall maintain documents for the Action 5. Document preservation. consistent with the terms provided by the Court's April 27, 2006 Agreed Order for Preservation of Records in the Action for a period ending at the close of document discovery in the Action as set by the Court in its initial scheduling order relating to and explicitly permitting merits discovery, or as otherwise agreed to by the plaintiffs and BCBSA.
- 6. Confidentiality. The Court's December 9, 2004 Protective Order Governing Protected Material Other Than Confidential Health Information shall apply to any discovery from BCBSA.
- 7. Use of materials. The plaintiffs shall not use any discovery provided under this Exhibit O for any purpose except for purposes of litigating their complaint in this Action against defendants that have not settled.
- 8. Format of production. BCBSA will produce documents in the scanned and imaged format the parties agreed upon in connection with the October 2004 class certification production.
- 9. Jurisdiction of court to enforce or resolve disputes. Disputes over compliance with a subpoena issued pursuant to paragraph I.1, or enforcement of the terms, of this Exhibit O shall be determined by the Court.

II. DEFINITIONS

- 1. "Action" means Love, et al. v. BCBSA, et al. 03-21296-CIV-MORENO (previously captioned "Thomas et al. v. BCBSA").
- 2. "BCBSA" means the Blue Cross and Blue Shield Association.
- 3. "BCBSA committee or workgroup" means a board, committee, workgroup, task force, ad hoc group, or other analogous group staffed by BCBSA and consisting of representatives from one or more of the defendants.
- 4. "BlueCard" means the BlueCard® Program of BCBSA.
- 5. "Consortium" means Consortium Health Plans, Inc.

- 6. "Court" means the court adjudicating the Action.
- 7. "CPT" has the same meaning as in the Settlement Agreement.
- 8. "Data" means electronic information showing claim line information, including CPT codes, for BlueCard claims submitted by physicians to Blue Cross and Blue Shield health plans.
- 9. "Defendants" shall refer to any of the other named defendants (i.e., except BCBSA) in the Action as of the Preliminary Approval Date.
- 10. "Document" means materials consistent with Federal Rules of Civil Procedure 34 and 45, except as otherwise provided herein.
- 11. "McKesson" means McKesson Corporation.
- 12. "NASCO" means National Accounts Service Company LLC.
- 13. "Officers or employees of BCBSA" refers to officers or employees of BCBSA itself. Officers or employees of Plans serving functions with BCBSA, such as by leadership or membership on a BCBSA committee or workgroup, shall not be deemed officers or employees of BCBSA.
- 14. "Plaintiffs" means the named plaintiffs in the Action and any other person acting on their behalf including, but not limited to, their partners, fellow shareholders, employees, agents, and attorneys.
- 15. "Preliminary Approval Date" means the Preliminary Approval Date for the Settlement Agreement.
- 16. "Representative Plaintiffs" has the same meaning as in the Settlement Agreement.
- 17. "Settlement Agreement" means the settlement agreement submitted to the Court by the plaintiffs, BCBSA, and others of which this Exhibit O is a part. Any capitalized terms in this Exhibit shall have the same meaning as in the Settlement Agreement.
- 18. "Submission Formats" has the same meaning as in BlueCard.

III.DISCOVERY REQUESTS

Subject to the terms and definitions above, BCBSA shall provide responsive discovery as follows:

A. DOCUMENTS

1. Minutes, agendas, and meeting materials distributed to attendees of the following BCBSA committees and workgroups:

Page 6 of 7

- a. Board of Directors
- b. Member Plans
- c. Executive Committee of the BCBSA Board of Directors
- d. Plan Performance and Financial Standards Committee
- e. Inter-Plan Programs Committee
- Inter-Plan Operations Committee
- g. Emerging Issues Committee
- h. National Council of Physician Executives
- National Council on Medical Management
- BlueCard Oversight Group
- k. National Business Advisory Council
- National Account Executive Committee
- m. BlueCard Business and Operations Workgroup
- n. BlueCard Program Executives
- o. BlueCard Performance Workgroup
- p. Adjustment and Inquiry Design Workgroup
- q. Blue Strategy Group on Trizetto
- 2. Documents from BCBSA's Office of Clinical Affairs relating to its efforts to address providers' concerns regarding coding issues and outreach to medical associations and provider groups regarding provider contracting and claims editing, including without limitation the so-called "Plan Focus Group on Code Bundling Software."
- 3. The following materials:
 - a. The Blue Cross and Blue Shield License Agreements
 - b. The Blue Cross and Blue Shield Controlled Affiliate License Agreements
 - c. BCBSA's Membership Standards Applicable to Regular Members
 - d. BCBSA's Guidelines to Membership Standards Applicable to Regular Members
 - e. BCBSA's Brand Book
 - BCBSA's BlueCard Program Manual (including documents sufficient to show changes made during the relevant class period)
 - g. BCBSA's National Account Program Manual (including documents sufficient to show changes made during the relevant class period)
- 4. Communications or documents describing communications between BCBSA and the following entities:
 - a. NASCO
 - b. Consortium
 - c. McKesson

B. DATA DISCOVERY

Page 7 of 7

The plaintiffs may seek data discovery from BCBSA in the form of Submission Formats used in actual BlueCard claims showing claim line information, and BCBSA shall produce such data to the extent it possesses it.

C. DEPOSITIONS

- 1. The plaintiffs may take a deposition of BCBSA pursuant to Federal Rule of Civil Procedure 30(b)(6) on the subjects of discovery permitted from BCBSA pursuant to this Exhibit O.
- 2. The plaintiffs' deposition of BCBSA may not exceed eight hours, of which six hours shall be reserved for the plaintiffs, one hour for any non-settling defendants, and one hour for BCBSA. If the plaintiffs believe, after taking the deposition of BCBSA, that they need a second eight hour segment for the Rule 30(b)(6) deposition, BCBSA will meet and confer with the plaintiffs concerning the requested second segment. If BCBSA declines the plaintiffs' request, the plaintiffs may raise the matter with the Court and the Court will decide whether to permit a second segment. Any second segment must be consistent with all terms otherwise set forth in this Exhibit O.
- 3. BCBSA obligations under this section shall be satisfied by producing witnesses who are current officers or employees of BCBSA.

Page 1 of 1

Exhibit P

List of Blue Parties Utilizing Individual Compliance Dispute Program

Pursuant to §12.7, the Blue Parties listed below will use an individual plan compliance dispute program that follows the requirements set out in §§ 12.1 through 12.6, in lieu of participating in the Joint Compliance Dispute Program.

Plans Utilizing Individual Compliance Dispute Program
Empire Blue Cross Blue Shield

Horizon Blue Cross Blue Shield of New Jersey

Exhibit Q

List of Blue Parties Utilizing Individual Plan Compliance Reporting Program

Pursuant to §12.8(b), the Blue Parties listed below will use an individual plan compliance reporting program and will designate a local compliance reporting officer who will follow the requirements and procedures as set out for the joint compliance reporting program in §12.8(a).

Plans Utilizing Individual Plan Compliance Reporting Program
Horizon Blue Cross Blue Shield of New Jersey
Health Care Service Corporation
Empire Blue Cross Blue Shield

EXHIBIT 2

Page 2 of 15

UNITED STATES DISTRICT COURT FOR THE SOUTHERN DISTRICT OF FLORIDA Miami Division

RICK LOVE, M.D., et al.,))	
Plaintiffs))	
)	Case No. 03-21296-CIV- MORENO/SIMONTON
v.))	
BLUE CROSS AND BLUE SHIELD ASSOCIATION, et al.,))	
Defendants))	
))	

ORDER PRELIMINARILY APPROVING PROPOSED SETTLEMENT AMONG CERTAIN BLUE PARTIES AND PHYSICIANS, PHYSICIAN GROUPS, AND PHYSICIAN ORGANIZATIONS, SETTING FORM AND CONTENT OF NOTICE TO THE CLASS, AND SCHEDULING SETTLEMENT HEARING

The Court having reviewed and considered the Joint Motion for Preliminary Approval of Settlement ("Joint Motion"), filed on April 27, 2007, and having reviewed and considered the terms and conditions of the proposed settlement (the "Settlement") as set forth in the settlement agreement dated April 27, 2007 (the "Settlement Agreement"), 1/2 a copy of which has been submitted as Exhibit 1 to the Joint Motion, and on the basis of such submissions, together with any other submissions by the Parties in support of the Joint Motion, and all prior proceedings had in this litigation, good cause for this Order having been shown,

NOW, THEREFORE, IT IS HEREBY ORDERED:

1. The terms of the Settlement Agreement are hereby preliminarily approved, subject to further consideration at the Settlement Hearing provided for below. The Court concludes that

Page 3 of 15

the Settlement is sufficiently within the range of reasonableness to warrant the conditional certification of the settlement Class (defined below), the scheduling of the Settlement Hearing and the circulation of notice to putative Class Members (defined below), each as provided for in this Order.

Conditional Certification of the Class

2. For purposes of settlement only, pursuant to Federal Rules of Civil Procedure 23(a), 23(b)(2), and 23(b)(3), solely with respect to the Blue Cross and Blue Shield Association ("BCBSA") and Blue Cross and Blue Shield of Alabama, Premera, Premera Blue Cross, PremeraFirst, Inc., LifeWise Health Plan of Washington, LifeWise Health Plan of Oregon, Inc., LifeWise Health Plan of Arizona, Inc., Premera Blue Cross Blue Shield of Alaska, Corp., Medical Services Corporation of Eastern Washington, NorthStar Administrators, Inc., CareFirst, Inc., Group Hospitalization and Medical Services, Inc., Access America, Inc., The GHMSI Companies, Inc., CareFirst BlueChoice, Inc., Capital Care, Inc., CareFirst of Maryland, Inc., CFS Health Group, Inc., Blue Cross and Blue Shield of Florida, Inc., Health Options, Inc., Hawaii Medical Service Association, The Regence Group, Regence BlueShield of Idaho, Inc., Regence BlueCross BlueShield of Utah, Regence BlueCross BlueShield of Oregon, Regence BlueShield, Regence Life and Health Insurance Company, RegenceCare, Regence HMO Oregon, Regence Health Maintenance of Oregon, Inc., Healthwise, Asuris Northwest Health, Wellmark, Inc. d/b/a Wellmark Blue Cross and Blue Shield of Iowa, Wellmark Health Plan of Iowa, Inc., Wellmark Community Insurance, Inc., Wellmark of South Dakota, Inc. dba Wellmark Blue Cross and Blue Shield of South Dakota, Louisiana Health Service & Indemnity Company dba Blue Cross and Blue Shield of Louisiana, HMO Louisiana, Inc., Blue Cross and Blue Shield of Massachusetts, Inc.,

Capitalized terms used in this Order that are not otherwise defined herein have the meaning

Page 4 of 15

Blue Cross and Blue Shield of Massachusetts HMO Blue, Inc., Blue Cross Blue Shield of Michigan, BCBSM, Inc. dba BlueCross BlueShield of Minnesota, HMO Minnesota dba Blue Plus, Comprehensive Care Services, Inc., Blue Cross & Blue Shield of Mississippi, HMO of Mississippi, Inc., Blue Cross and Blue Shield of Montana, Inc., Horizon Healthcare Services, Inc. d/b/a Horizon Blue Cross Blue Shield of New Jersey, Horizon Healthcare Plan Holding Company, Inc., Horizon Healthcare Insurance Company of New York, Horizon Healthcare of New Jersey, Inc., Horizon Healthcare of New York, Inc., Enterprise Holding Company, Inc., AtlantiCare Administrators, Inc., Horizon Healthcare Administrators, Inc., Horizon AtlantiCare LLC, Horizon Healthcare of Delaware, Inc., NASCO of New Jersey, Inc., Empire HealthChoice Assurance, Inc., d/b/a Empire Blue Cross Blue Shield, Empire HealthChoice HMO, Inc., WellChoice Insurance of New Jersey, Inc., WellChoice Holdings of New York, Inc., WellPoint Holding Corp., Blue Cross and Blue Shield of North Carolina, Hospital Service Association of Northeastern Pennsylvania, HMO of Northeastern Pennsylvania, Triple-S, Inc.; Triple-S, Inc., of Puerto Rico, Triple-C, Inc., Blue Cross Blue Shield of Rhode Island, Blue Cross and Blue Shield of South Carolina,

BlueChoice HealthPlan of South Carolina, Inc. f/k/a Companion HealthCare Corporation, Planned Administrators, Inc., Thomas H. Cooper & Co., Inc., BlueCross BlueShield of Tennessee, Inc., Tennessee Health Care Network, Inc., Health Care Service Corporation, Group Health Maintenance Organization, Inc. d/b/a Bluelincs HMO, Group Health Service of Oklahoma, Inc. d/b/a Blue Cross Blue Shield of Oklahoma, HMO New Mexico, Inc., New Mexico Blue Cross and Blue Shield, Inc. d/b/a Blue Cross and Blue Shield of New Mexico, New Mexico Blue Cross and Blue Shield, Inc., Hallmark Services Corporation, BCI HMO, Inc., HCSC Insurance Services Corporation (collectively, the "Blue Parties"), and the other Released Parties, the action styled as

assigned to them in the Settlement Agreement.

Page 5 of 15

Love, et al. v. Blue Cross and Blue Shield Ass'n, et al., (formerly, Thomas, et al. v. Blue Cross and Blue Shield Ass'n, et al.), Case No. 03-21296-CIV-MORENO/SIMONTON (the "Action") is conditionally certified as a class action on behalf of the following persons (collectively, the "Class" and each member of the Class a "Class Member"):

> any and all Physicians, Physician Groups and Physician Organizations who provided Covered Services to any Plan Member or services to any individual enrolled in or covered by a Plan offered or administered by any Person named as a defendant in the Complaint or by any other primary licensee of the BCBSA or by any of their respective current or former subsidiaries or Affiliates, in each case from May 22, 1999 through the Preliminary Approval Date. The Class shall exclude: (i) all Persons who, in accordance with the terms of this Agreement, execute a timely request for exclusion (Opt-Out) from the Class; and (ii) the Blue Parties, their Affiliates and any of their officers, directors and employees.

Representative Plaintiffs are conditionally appointed as representatives of the Class. Class Counsel are appointed and designated as counsel for the Class. This conditional certification of the Class appointment of the Representative Plaintiffs as representatives of the Class and appointment and designation of Class Counsel as counsel for the Class are solely for purposes of effectuating the Settlement. If the Settlement Agreement is terminated or is not consummated for any reason, the foregoing conditional certification of the Class, appointment of the Representative Plaintiffs as representatives of the Class, and appointment and designation of Class Counsel as counsel for the Class shall be void and of no effect and the Parties to the Settlement Agreement and/or any Blue Party terminating its participation in the Settlement Agreement in accordance with its terms, shall be returned to the status each occupied before entry of this Order without prejudice to any legal argument that any of the Parties to the Settlement Agreement might have asserted but for the Settlement Agreement.

Page 6 of 15

Based on the Court's review of the Joint Motion and supporting materials, the Court conditionally finds that the proposed Class satisfies Rule 23(a), 23(b)(2), and 23(b)(3) of the Federal Rules of Civil Procedure.

Document 928-25

Settlement Hearing; Right to Appear and Object

- 3. A Settlement Hearing shall take place before the Court on [DATE], at [TIME] in the United States Courthouse, Courtroom IV, Tenth Floor, Federal Justice Building, 99 Northeast Fourth Street, Miami, Florida, 33132 to determine:
- (a) whether the Court should certify the Class and whether the Court should find the Representative Plaintiffs and Class Counsel as adequate representatives of the Class and counsel for the Class, respectively;
- (b) whether the Settlement, on the terms and conditions provided for in the Settlement Agreement, should be finally approved by the Court as fair, reasonable, and adequate;
- whether the Released Claims of the Class Members and Signatory Medical (c) Societies asserted in the Action should be dismissed on the merits and with prejudice as to the Blue Parties;
- whether the Court should permanently enjoin the assertion of any Claims (d) that arise from or relate to the subject matter of the Action against Released Parties by the Class Members or any other Persons;
- (e) whether the application for attorneys' fees, costs, and expenses to be submitted by Class Counsel in connection with the Settlement Hearing should be approved;
- (f) whether the application for payments to Representative Plaintiffs and Representative Plaintiffs in Other Actions to be submitted in connection with the Settlement Hearing should be approved;

Page 7 of 15

- whether each Tag-Along Action should be dismissed with prejudice as to (g) Released Claims against the Blue Parties, with the exception of any Tag-Along Action with respect to any named plaintiff who or which has timely submitted an Opt-Out request;
- (h) whether the Court should permanently enjoin the assertion of Claims by the Releasing Parties that arise from or relate to conduct by the Released Parties that occurred on or before the Effective Date of the Settlement Agreement, except for Claims for damages against Defendants in the Action that are not Parties, relating to providing Covered Services to Blue Plans' Plan Members in connection with the BlueCard Program and other similar national account delivery programs governed by BCBSA (including but not limited to NASCO-to-NASCO arrangements; and
 - such other matters as the Court may deem necessary or appropriate. (i)
- 4. The Court may finally approve the Settlement Agreement at or after the Settlement Hearing with any appropriate modifications agreed to by the Parties and without further notice to the Class Members.
- Any putative Class Member who or which has not timely and properly provided 5. notice of an election to Opt-Out of the Class and the Settlement Agreement in the manner set forth below, and any other interested Person, may appear at the Settlement Hearing in person or by counsel and be heard, to the extent allowed by the Court, either in support of or in opposition to the matters to be considered at the Settlement Hearing, provided, however, that no Person who or which has elected to Opt-Out of the Class and the Settlement Agreement shall be entitled to object; and provided further that no Person shall be heard, and no papers, briefs, or other submissions shall be considered by the Court in connection with the Court's consideration of those matters, unless on or before a date seventy five (75) days after the Notice Date, such Person:

Page 8 of 15

- files with the Court a written statement of objection setting forth such (a) Person's objections to the matters to be considered and the basis therefore, together with all other support, papers, or briefs that he, she, or it wishes the Court to consider. If any Person making an objection wishes to appear at the Settlement Hearing, they must also file a written notice of intention to appear; and
- (b) serve copies of all such materials either by hand or overnight delivery upon the following counsel:

Edith M. Kallas Joe R. Whatley, Jr. Joseph P. Guglielmo WHATLEY DRAKE & KALLAS LLC 1540 Broadway, 37th Floor New York, New York 10036

Archie Lamb, Jr. LAW OFFICES OF ARCHIE LAMB, LLC 2017 Second Avenue North, 2nd Floor Birmingham, Alabama 35203

Harley S. Tropin Kozyak Tropin & Throckmorton, P.A. 2525 Ponce de Leon Blvd., 9th Floor Miami, Florida 33134

And

Michael A. Pope MCDERMOTT WILL & EMERY LLP 227 W. Monroe Street Chicago, IL 60606

Chester T. Kamin JENNER & BLOCK LLP 330 North Wabash Chicago, IL 60611

All responses by the Parties to objections shall also be served by hand or overnight delivery on the objecting Person or his, her, or its attorney no later than fifteen (15) days before the Settlement Hearing. Objectors' papers may be supplemented ten (10) days prior to the Settlement Hearing

Page 9 of 15

following the filing of any responsive papers by the Parties and served in the manner and on counsel described above.

6. The Court may adjourn the Settlement Hearing, or any adjournment thereof, including the consideration of the application for attorneys' fees, costs and expenses, without any additional notice to putative Class Members other than an announcement in open court.

Form and Timing of Notice

- No earlier than thirty (30) days and no later than sixty (60) days after entry of this 7. Order, Class Counsel and the Blue Parties through the Notice Administrator shall cause copies of the Notice of Class Action and Proposed Settlement in the form attached as Exhibit E to the Settlement Agreement (the "Mailed Notice"), the form of which is hereby approved, and in the case of Class Members who are individuals, including the claim form substantially in the form attached as Exhibit A to the Settlement Agreement (the "Claim Form"), the form of which is hereby approved, to be mailed by first-class mail, postage pre-paid, to all putative Class Members who can be identified with reasonable diligence from (a) solely in the Blue Plans' discretion, the Blue Plans' books and records, at their last known address appearing in such books and records, and/or (b) records maintained by the American Medical Association (the "AMA") and/or a vendor associated therewith, to the extent that such records can be obtained for this purpose within the schedule for notice provided in this Order.
- 8. No earlier than thirty (30) days and no later than sixty (60) days after entry of this Order, Class Counsel and the Blue Parties through the Notice Administrator shall cause to be published three times in the legal notices sections of the national editions of The Wall Street Journal and USA Today, a summary notice substantially in the form attached as Exhibit G to the Settlement Agreement (the "Published Notice"), the form of which is hereby approved. To the

Page 10 of 15

extent reasonably feasible within the schedule contemplated by this Order, Class Counsel and the Blue Parties through the Notice Administrator shall cause the Published Notice to be published in a nationwide periodical addressing issues of concern to physicians, such as The Journal of the American Medical Association or The American Medical News. If publication in one or more said publications on the foregoing schedule is not practicable, then either Class Counsel or the Blue Parties may apply to the Court for alternative notice by publication. In addition, no later than thirty (30) days prior to the Opt-Out Deadline and continuing through the Final Order Date, each Blue Plan shall publish the Mailed Notice on its Public Website.

- 9. Beginning on the earliest date that notice is provided pursuant to paragraphs 6 and 7 of this Order and continuing through the deadline for putative Class Members to return a completed Claim Form (as that date is specified in the Settlement Agreement), Blue Parties and Class Counsel shall make available to putative Class Members printable versions of the Mailed Notice and the Claim Form on the public websites maintained by or on behalf of the Blue Plans and at least three of the law firms serving as Class Counsel in the Action.
- 10. Prior to the Settlement Hearing, Class Counsel and the Blue Parties shall serve and file a sworn statement attesting to compliance with paragraphs 6, 7, and 8 of this Order. Costs of providing the notice to the Class that is specified in this Order shall be paid as set forth in the Settlement Agreement.
- 11. The notice to be provided as set forth in paragraphs 6, 7, and 8 of this Order is hereby found to be the best means of notice to putative Class Members that is practicable under the circumstances and, when completed, shall constitute due and sufficient notice of the Settlement, the Settlement Agreement and the Settlement Hearing to all Persons affected by and/or entitled to

Page 11 of 15

participate in the Settlement, the Settlement Agreement or the Settlement Hearing, in full compliance with the requirements of due process and the Federal Rules of Civil Procedure.

12. All notices to any Party (including without limitation any designations made by Class Counsel pursuant to the Settlement Agreement) required under the Settlement Agreement shall be sent by first class U.S. Mail, by hand delivery, or by facsimile, to the recipients designated in the Settlement Agreement. Unless otherwise specifically stated, timeliness of all submissions and notices shall be measured by the date of the postmark. The Persons designated to receive notices under the Settlement Agreement are as follows (unless the addressee refuses or intentionally or willfully delays receipt, or unless notification of any change to such designated is given to each other Party to the Settlement Agreement pursuant to Section 5 of the Settlement Agreement).

Representative Plaintiffs and Signatory Medical Societies: Notice to be given to Class Counsel on behalf of Representative Plaintiffs and Signatory Medical Societies.

Class Counsel:

Edith M. Kallas Joe R. Whatley, Jr. Joseph P. Guglielmo WHATLEY DRAKE & KALLAS LLC 1540 Broadway, 37th Floor New York, New York 10036

Archie Lamb, Jr. LAW OFFICES OF ARCHIE LAMB, LLC 2017 Second Avenue North, 2nd Floor Birmingham, Alabama 35203

Harley S. Tropin Kozyak Tropin & Throckmorton, P.A. 2525 Ponce de Leon Blvd., 9th Floor Miami, Florida 33134

Page 12 of 15

Blue Parties:

Michael A. Pope MCDERMOTT WILL & EMERY LLP 227 W. Monroe Street Chicago, IL 60606

Chester T. Kamin JENNER & BLOCK LLP 330 North Wabash Chicago, IL 60611

Ability of Putative Class Members to Opt-Out of Settlement Class

- All putative Class Members who or which wish to Opt-Out of the Class and the 13. Settlement must do so by sending written notice of their election to Opt-Out to the Notice Administrator at the address and in the manner set forth in the notices to be provided pursuant to paragraphs 6, 7, and 8 of this Order. To be considered proper and timely, a complete, signed Opt-Out request must be postmarked by no later than 60 days after the Notice Date (the "Opt-Out Deadline"). Prior to the Settlement Hearing, Class Counsel or their designee shall submit to the Court a sworn statement setting forth the names and addresses of each putative Class Member who or which has properly and timely elected to Opt-Out of the Class and the Settlement.
- Any putative Class Member who or which does not properly and timely Opt-Out of 14. the Class shall be included in such Class and, if the Settlement is approved and becomes effective, shall be bound by all the terms and provisions of the Settlement Agreement, including but not limited to the releases, waivers, and covenants not to sue described therein, whether or not such Person shall have objected to the Settlement and whether or not such Person makes a claim upon, or participates in, the Settlement Fund or the other benefits to the Class to be provided under the Settlement Agreement.

Page 13 of 15

Other Provisions

- Released Parties in the Action, except with respect to discovery from the BCBSA as set forth in Exhibit O to the Settlement Agreement, and other than proceedings as may be necessary to carry out the terms and conditions of the Settlement, are hereby stayed and suspended until further order of the Court. Pending final determination of whether the Settlement and the Settlement Agreement should be finally approved and the Class permanently certified, all putative Class Members and Signatory Medical Societies are hereby barred and enjoined from commencing or prosecuting any action asserting any Released Claims, and any other action or proceedings brought by any putative Class Members or Signatory Medical Societies asserting any Released Claims are hereby stayed and suspended until further order of the Court.
- 16. All discovery and any other proceedings against or concerning the Blue Parties in Tag-Along Actions asserting any Released Claim, other than proceedings as may be necessary to carry out the terms and conditions of the Settlement, are hereby stayed and suspended until further order of the Court.
- 17. No discovery with regard to the Settlement or the Settlement Agreement shall be permitted other than as may be directed by the Court upon a proper showing by the Person seeking such discovery by motion properly noticed and served in accordance with this Court's Local Rule 7.1.
- 18. Pending entry of an Order setting forth the final determination of the Court with respect to the Parties' application for final approval of the Settlement, the Released Parties are hereby relieved from the date of this Order and going forward, of any and all obligations to preserve documents and evidence with respect to Released Claims, including any obligations to

Page 14 of 15

preserve Future Documents under the Agreed Order For Preservation of Records entered on April 27, 2006 in this Action, or any similar obligations to preserve future documents relating to Released Claims in any Tag-Along Action. This Order does not relieve any Party from their obligations under the Agreed Order for Preservation of Records as entered in and applicable to Solomon, et al. v. Blue Cross and Blue Shield Ass'n, et al., Case No. 03-22935-CIV-MORENO/SIMONTON.

- 19. The Blue Parties are authorized to communicate with putative Class Members and Signatory Medical Societies regarding the provisions of the Settlement Agreement (including, but limited to, the Blue Parties' right to withhold all Settlement benefits from putative Class Members who or which Opt-Out of the Class and the Settlement) so long as such communications are not inconsistent with the agreed upon communications concerning the Settlement Agreement. The Blue Parties shall refer to the Settlement Administrator or to Class Counsel any inquiries from Class Members about claims to be filed under the Settlement Agreement.
- Any putative Class Member may enter an individual appearance in this Action, at his, her, or its own expense. Any putative Class Member who or which does not enter an individual appearance will be represented by Class Counsel.
- 21. Neither the Settlement, nor the Settlement Agreement nor any provision therein, nor any negotiations, statements, or proceedings in connection therewith shall be construed as, or be deemed to be evidence of, an admission or concession on the part of any of the Representative Plaintiffs, the Signatory Medical Societies, Class Counsel, any putative Class Members, the Blue Parties, or any other Person of any liability or wrongdoing by them, or that the claims and defenses that have been, or could have been, asserted in the Action were or were not meritorious, and neither the Settlement, the Settlement Agreement, nor any such communications shall be offered

Page 15 of 15

or received in evidence in any action or proceeding, or be used in any way as an admission or concession or evidence of any liability or wrongdoing of any nature or that Representative Plaintiffs, the Signatory Medical Societies, any Class Member, the Blue Parties, or any other Person has or has not suffered any damage.

22. In the event that the Settlement Agreement is terminated by Class Counsel or by all .Blue Parties or is not consummated for any reason, the Settlement and all proceedings had in connection therewith shall be null and void, except to the extent expressly provided to the contrary in the Settlement Agreement, and without prejudice to the rights of the Parties to the Settlement Agreement before it was executed. In addition, in the event any individual Blue Party terminates its participation in the Settlement Agreement in accordance with its terms and provisions, then this Order and all orders entered in connection therewith, shall be rendered null and void as to the terminating Blue Party.

	DONE AND ORDERED after a hearing in open court at the United States District			
	Courthouse in Miami, Florida, on	, and signed this	day of	
•	, 2007.			

HON. FEDERICO A. MORENO UNITED STATES DISTRICT JUDGE